Mandibular Growth in Australopithecus robustus

by

Zachary Daniel Cofran

A dissertation submitted in partial fulfillment of the requirements for the degree of Doctor of Philosophy (Anthropology) in the University of Michigan 2012

Doctoral Committee

Professor, Milford H. Wolpoff, Chair Professor, Laura M. MacLatchy Assistant Professor, Eric A. Hetland Assistant Professor, Adam P. Van Arsdale, Wellesley College © Zachary Cofran All rights reserved 2012

Acknowledgements

There is absolutely no way I could have done any of this work, or survived any of the last five years, on my own. There are a number of people I wish to thank for their help and support along the way.

I have learned a tremendous amount about my field, the world, and the secrets of comedy from my advisor Milford Wolpoff, and I am very fortunate to've stood on his anthropological shoulders. I have also benefited tremendously from the sagacity and advice of Adam Van Arsdale, regarding this dissertation and especially fieldwork. These two and the rest of my committee, Laura MacLatchy and Eric Hetland, have been extremely patient and helpful in making this dissertation possible. And the past five years could not have gone as smoothly without the UM biological anthropology faculty, who have been exquisite mentors and friends.

I would not have survived a day of fossil research without the assistance of Stephany Potze, Laz Kgasi, and Tersia Perregil at the Ditsong National Museum. Not only did they help me in my time crunch, but they also helped me navigate the left-sided streets of Gauteng on my own. Good company and good advice from my cottage roommates Justin Adams and Anthony Kegley made a good research trip great.

I am very grateful to the Anthropology department and graduate students of Kent State University. Richard Meindl provided very generous access to the Libben collection, and I could not have navigated it without his and Mary Ann Raghanti's help. I am ever indebted to Caroline Tannert and Aidan Ruth who really made me feel at home in Kent.

ii

My friends at the University of Michigan, especially anthropologists, but especially my bioanthrobuddies, made this journey unforgettable. They were always there to help me get into, or out of, trouble – whichever was appropriate at the time. Dana Begun deserves special mention for helping me stay sane all the while. Last and most certainly not least, I owe all I am and all I have to the love and support of my family, the Cofrans and Stuckeys.

Table of Contents

Acknowledgements	ii
List of Figures	v
List of Tables	vi
List of Appendices	vii
Abstract	viii
Chapter 1 Introduction	1
Chapter 2 Background and Theoretical Contexts	8
Chapter 3 Materials and Methods	40
Chapter 4 Results of the ζ Test	65
Chapter 5 Implications of A. robustus mandibular growth	
Chapter 6 A Look Back and Ahead	
Bibliography	

List of Figures

 Figure 1 The Australopithecus robustus ontogenetic series from Swartkrans	\$ F 5 3,
Figure 4.3. ζ statistics for the overall size metric for all pairwise combinations of)
differently-aged humans	>
Figure 4.4 ζ statistic for the overall size metric comparing successive dental	
eruption stages73	3
Figure 4.5. The number of traits shared between individuals in the ζ test	_
)
Figure 4.6. The influence of the number of traits (K) on the value Statistics in	
Figure 4.7 The number of traits (k) shared between specimens spanning	,
different eruption stages	3
Figure 4.8. ζ statistics for species difference in overall size change between all	
eruption stages)
Figure 4.9. $\boldsymbol{\zeta}$ statistics for interspecific difference in relative size change between	
successive eruption stages85	5
Figure 4.10: The effects of preservation on the value of ζ	,
Figure 4.11. Distribution of ζ for each difference between eruption stages 90)
Figure 4.12. The distributions of all possible interspecific ζ statistics for each	
Individual trait	-
Figure 4.13. Comparison of the ζ statistic for individual traits and for the pairwise	h
Size file includes of (statistics measuring difference in overall size change	,
from the validation and interspecific tests	
Figure 5. 2. The relationship between the value of ζ and the number of traits	
used to calculate ζ	>
Figure 5.3. All (statistics for height at the mandibular symphysis and at the	
ramus)

List of Tables

Table 3.1. Provenience and catalog numbers of <i>Australopithecus robustus</i>	
specimens in this analysis.	42
Table 3.2. Sample sizes for each eruption stage	. 48
Table 3.3. List and description of measurements.	. 56
Table 3.4. Sample metadata (aside from the ζ statistic) stored for each	
resampled comparison	. 62
Table 4.1. Species' average pairwise size in each eruption stage	. 69
Table 4.2. ζ statistics for the overall size metric comparing the human sample	
with itself	. 71
Table 4.3. Relationship between resampled ζ and metadata	. 74
Table 4.4. Species difference in relative change for the pairwise size metric	
(geometric mean).	. 81
Table 4.5. Relationship between resampled ζ and metadata	. 86
Table 4.6. Summary of ζ statistics comparing humans and A. robustus for eac	;h
trait, across all age comparisons	. 96
Table 4.7. ζ statistics comparing humans and <i>A. robustus</i> for each trait betwee	en
successive eruption stages	. 99
Table 4.8. Position of the ramus root across eruption stages in each species	100

List of Appendices

Appendix I: Raw measurements plotted against eruption stage	171
Appendix II: "Robusticity index" plotted against eruption stage	186
Appendix III: R codes for analyses	190

Abstract

Mandibular growth in *Australopithecus robustus* by Zachary Daniel Cofran

Chair: Milford H. Wolpoff

This dissertation tests the hypothesis that humans' unique pattern of growth was present in our fossil relative, *Australopithecus robustus*. Growth and development encompass the mechanisms and processes that bring about morphological variation and adaptation. Growth is also an important life history variable influencing both an animal's energy requirements and how it is treated by predators and conspecifics. Humans' unique pattern of growth, featuring long juvenile and childhood phases of slow growth followed by an adolescent growth spurt, is critical to the acquisition of language and culture. Unfortunately, the fragmentary and cross-sectional nature of fossil samples makes it difficult to ascertain the evolution of this pattern.

A novel randomization method, the $\zeta \square \square \square \square \square \square$ test, is developed to statistically test the null hypothesis that patterns of ontogenetic variation cannot be distinguished in the mandibles of humans and *A. robustus*. This species is not a direct human ancestor, but is closely related with a recent common ancestor and its mandible comprises the largest ontogenetic series of an early hominid. The ζ test compares agerelated changes between pairs of specimens, and so maximizes information obtained from fossil samples compared with traditional methods. This method can be extended to test other hypotheses with datasets do not meet assumptions of traditional statistics.

Results of the ζ test indicate *A. robustus* mandibular growth differs from humans' during two periods. First, prior to the eruption of the first permanent teeth, *A. robustus*' corpus breadth increases much more than humans'. Second, the *A. robustus* mandible increases in most dimensions in the time between the eruption of the first and second permanent molars, whereas human mandibular size tends to increase more after the second molar is occluded. These results suggest that the human pattern of growth was not shared with *A. robustus*, and provide the first statistical evidence for differences in skeletal size growth between humans and an early Pleistocene hominid. These findings are discussed in terms of life history and evolutionary developmental biology, including the possibility of early weaning, the influence of the developing dentition, and bone functional adaptation in *A. robustus*.

Chapter 1

Introduction

Processes of growth and development are the proximate mechanisms responsible for humankind's unique biology and behaviors. The goal of this dissertation is to increase our understanding of growth and development in early hominid evolution, to better understand how and when the human condition evolved. The most direct evidence for this must come from the fossil record (Raff, 2007), and despite a large number of hominid fossils at various stages of ontogeny, there is still much to be learned about how humans' extinct ancestors and relatives grew up. Some of the gaps in our knowledge lie in the nature of fossil samples – subadults and adequate representations of ontogenetic series are relatively rare – while others lie in the methods with which ontogenetic studies have been carried out. In order to provide new evidence as to the evolution of human development, this dissertation presents a new method for analyzing patterns of growth in fossil samples, comparing the amount of size change between dental eruptions in mandibles of humans and our extinct relative Australopithecus robustus. This chapter introduces this species, and previews the content of this contribution to paleoanthropology.

Kissing Cousins.

The earliest age of the Pleistocene epoch, from 2.58-1.81 million years ago (mya; Gibbard et al., 2010), was a key time in hominid evolution. This period saw the rise of at least two contemporaneous and sympatric lineages: the genus *Homo* (Leakey et al., 1964), our ancestors, and a lineage of "robust" *Australopithecus* in both East (Leakey and Leakey, 1964; Leakey and Walker, 1976) and South Africa (Broom, 1938; Broom and Robinson, 1949). These lineages likely share a common ancestor in a 'gracile' australopith (or "australopithecine," referring to any species in the genus *Australopithecus*) such as *Australopithecus afarensis* or *africanus*, and can be distinguished largely on the basis of the development of the masticatory apparatus. Fossils attributed to *Homo* generally have more lightly built jaws, smaller muscle attachments, and smaller postcanine teeth than the robust australopiths: hence the term "robust" refers only to craniodental characteristics. Because of this gross similarity, robust australopiths are often grouped together in the genus *Paranthropus* (e.g. Strait and Grine, 2004), although the implied monophyly is uncertain on anatomical (Rak, 1983; Kimbel et al., 1988; McCollum, 1999; but see Suwa et al., 1997) and ecological or trophic grounds (Ungar et al., 2008; van der Merwe et al., 2008; Cerling et al., 2011).

This research examines similarities in the pattern growth (i.e. the amount of size change between dental eruptions) in members of each of these lineages, the extinct *Australopithecus robustus* and modern humans (*Homo sapiens*). The *A. robustus* mandibular sample is the largest ontogenetic series of an early Pleistocene hominid, and this species is very closely related to humans sharing a recent common ancestor. *A. robustus* fossils are known from a number of Pleistocene sites in the Sterkfontein Valley in South Africa. The type specimen, the partial skull TM 1517, comes from the B deposits of the Kromdraai site (Broom, 1938), but the bulk of this species' fossils are from the nearby site of Swartkrans (Broom, 1949; Brain, 1981, 2004). The Swartkrans

australopiths were initially distinguished as a separate species, *Paranthropus crassidens* (Broom, 1949; Grine, 1982), but most now consider these remains to be conspecific with the Kromdraai B remains (Brain, 1981; Fuller, 1996; Kaszycka, 2002; Cofran and Thackeray, 2010).

More recently, some remarkable findings from other Sterkfontein valley sites have been attributed to *A. robustus*. Two hominid teeth are known from the site of Gondolin (Menter et al., 1999). Of these, GDA-2 is a lower second molar (M₂) larger than any other South African australopith molar, but within the range of the East African "hyper-robust" *A. boisei*. In addition, DNH 7 is the most complete skull (cranium and mandible) attributed to *A. robustus*, coming from the site of Drimolen (Keyser, 2000; Keyser et al., 2000). Finally, COB 101 a partial (and crushed) cranium comes from the site of Cooper's Cave (Steininger et al., 2008). *A. robustus*-bearing deposits from all these sites, including Swartkrans and Kromdraai, are probably roughly contemporaneous (Herries et al., 2009; Pickering et al., 2011). However, for reasons discussed in Chapter 3, the present analysis focuses on the *A. robustus* mandibular series from Swartkrans only.



Figure 1. The *Australopithecus robustus* ontogenetic series from Swartkrans. From youngest to oldest, these include: (bottom row, left to right) SK 438, 64, 3978, 62, 61, 63; (top row, left to right) SK 25, 55b, SKX 4446, SK 843, 6, and SKW 5. SK 1587 is not pictured.

The *A. robustus* face is generally interpreted in terms of its adaptive capabilities for processing a variable diet, relying at least occasionally on hard-objects like nuts and/or requiring prolonged periods of chewing (Robinson, 1954; Rak, 1983; Wolpoff, 1999; Teaford and Ungar, 2000; Grine et al., 2012). From a functional standpoint, the species' jaws and face are adapted to creating high bite forces across the molars and premolars. Tall mandibular rami provide a lever advantage to the medial pterygoid muscles and to the masseter muscle laterally, for generating largely vertical chewing forces (Rak and Hylander, 2008). Tall rami also reduce gape (i.e. the ability to accommodate a large food bolus), which results in more evenly distributed forces throughout the length tooth row.

Consistent with its powerful chewing capabilities, mechanical properties of *A*. *robustus*' bony mandible itself allow it to withstand high stresses from mastication (Wolpoff, 1975; White, 1977). For instance, the ramus is strengthened against vertical bending forces from the temporalis muscle by a pronounced endocoronoid buttress. The mandibular symphysis is buttressed with superior and inferior 'tori,' keeping the symphysis from wish-boning during mastication. More distally, the corpus is resistant to transverse bending and torsion forces due to the distribution of cortical bone about a relatively broad cross-section (Daegling, 1989; Daegling and Grine, 1991; Grine and Daegling, 1993). The broad corpus in many cases takes on a nearly circular cross-section, a geometry that theoretically increases strength against torsion and mediolateral bending, although it is the distribution of cortical bone about this cross-section that actually determines the corpus' strength.

In light of *A. robustus*' masticatory capabilities, discerning the species' actual diet has been difficult (reviewed in Grine et al., 2012). Analysis of microscopic wear on the molar and premolar teeth of *A. robustus* suggest a diet including more hard-objects than *Australopithecus africanus* (a South African hominid predating and possibly ancestral to *A. robustus*), but there is overlap between the two species (Scott et al., 2005). In addition, both these species are indistinguishable from South African early *Homo* in their ratios of dietary carbon isotopes, a chemical signature that can discern only broad dietary categories (Lee-thorp et al., 2000; Lee-Thorp and Sponheimer, 2006; Ungar and Sponheimer, 2011). Finally, there is mounting evidence that *A. robustus* subsisted to some degree on termites, which it would have obtained using bone tools such as those found at Swartkrans and Drimolen (Backwell and D'Errico, 2001; Lesnik and Thackeray, 2007; Backwell and d' Errico, 2008; Lesnik, 2011 a-b). Suffice it to say, the derived facial anatomy of *A. robustus* betrays what was probably a rather broad diet in this early hominid (Pickering, 2006).

This dissertation tests the null hypothesis that patterns of mandibular growth in humans and *A. robustus* cannot be distinguished. This hypothesis is a basic but important place to investigate growth in this extinct hominid. Such a comparison allows inferences to be made about how *A. robustus*' developmental biology and life history. Subadult fossils make up a large proportion of this species' craniodental hypodigm (Brain 1981), and so it is surprising that these have not yet been used to analyze patterns of size change in *A. robustus* (for *adult* growth, see Lockwood et al., 2007).

Chapter 2 presents the theoretical and analytical reasons for analyzing mandibular growth in *A. robustus*. The research question about mandibular growth in *A. robustus* is

situated in two bodies of theory. First, from the view of evolutionary developmental biology (Raff, 2000), understanding how the *A. robustus* mandible attains its unique morphology helps elucidate the (adaptive) reasons for adult anatomy. Second, from a life history perspective, patterns of growth are part of an animal's adaption to survive in its ecological circumstances (Smith and Tompkins, 1995). Human growth reflects key aspects of humans' unique adaptation relying on language and culture (Bogin, 1999, 2009; Locke and Bogin, 2006), and so comparing patterns of growth in humans with *A. robustus* can help elucidate the origins of this adaptation. I conclude with a brief discussion of the null hypothesis, explaining the reasoning behind the null approach given the issues with using cross-sectional data.

Chapter 3 describes in more detail the human and fossil samples, and the methods employed to study them. The study will focus on *A. robustus* from Swartkrans, as well as a prehistoric human population from modern Ohio. Although there are numerous wellestablished ways to study growth cross-sectionally, none of these is sufficient to understand ontogenetic variation in the *A. robustus* mandibular sample. The remainder of Chapter 3 describes why the fossil sample necessitates a new approach, and then presents a novel method, which I call the ζ test, for assessing patterns of ontogenetic variation in fossil samples based on randomization/resampling statistics (Efron and Tibshirani, 1991; Sokal and Rohlf, 1995; Manly, 2007).

Chapter 4 presents the performance and results of the ζ test. The chapter begins by documenting the analytical behavior of this new procedure, first in terms of the metric chosen to describe overall mandibular size, and then in terms of the randomization strategy. Patterns of size change are then compared between humans and *A. robustus*,

first for the overall size metric and then for the individual metric traits examined. The chapter ends with a qualitative description of ontogenetic changes to the *A. robustus* symphysis.

Chapter 5 discusses the significance of these findings. The chapter begins with a discussion of the ζ test as an analytical procedure, reviewing why the new method was necessary and how it performed in light of its assumptions and the limitations of a cross-sectional sample. Next I treat the results in terms of evo-devo, examining the potential effects of early weaning, biomechanical adaptation, and dental development on *A*. *robustus* jaw growth. Finally, I discuss the implications for *A*. *robustus* life history.

Chapter 6 then summarizes and synthesizes the previous chapters, noting the significance of the present study and making suggestions for future work. Just as this dissertation is based on maximizing the information obtained from fossil samples, these suggestions focus on the development of methods rather than awaiting an expanded fossil record. Even though much of the *A. robustus* sample studied here has been known for as much as 60 years, it is my belief that there is still much to be learned from these and other familiar fossils.

Chapter 2

Background and Theoretical Contexts

This study compares mandibular growth in *Australopithecus robustus* with that of modern humans, and the purpose of this chapter is to present the theoretical underpinnings of the research question of whether their patterns can be distinguished. The chapter discusses the biological and theoretical contexts for the research question and hypothesis. First, from the perspective of evolutionary developmental biology ("evo-devo"), it is of interest to know when and how morphological differences between these species arise. Second, from the perspective of life history theory (Stearns, 1977, 2000; Smith and Tompkins, 1995; Robson and Wood, 2008), patterns of growth are part of an organism's adaptation for meeting the energetic demands of its lifestyle; human life history has several unique attributes including our pattern of growth (Bogin, 1999; Leigh, 2001). I conclude with a brief discussion of cross-sectional data and the strengths of this study.

2.1 Evolutionary Developmental Biology

This study is directly relevant to human evo-devo because it examines how and why anatomical differences between humans and *A. robustus* arise in ontogeny. This section therefore discusses the possible developmental bases of human and *A. robustus* adult mandibular morphology. Special attention is given to the issue of making inferences about growth from the fossil record, and why the *A. robustus* sample is well suited for

this. I then discuss specific hypotheses that have previously been proposed for *A*. *robustus* jaw development, and how this study can address them.

Evo-devo is a multifaceted field of biology that seeks to understand how growth and development are modified to effect evolution (Raff, 2000, 2007). Patterns of growth are important in evolution because no animal is able to thrive and reproduce at the size and shape it is at birth; adult form arises through processes of growth and development. Because a requirement of evolution by natural selection is for individuals to vary in their potential to survive and reproduce, and growth and development are essentially responsible for an animal's phenotype, ontogeny is an important mediator of evolution (Gould, 1977). As the preeminent biologist Leigh Van Valen stated, "A plausible argument could be made that evolution is the control of development by ecology" (Van Valen, 1973: 488). A developmental perspective therefore helps elucidate the evolutionary significance of similarities and differences between species. A comparison of the growth processes resulting in species' similarities and differences is an ideal focus for the evo-devo approach. At the molecular level, most fundamental developmental processes are shared in common (e.g. conserved) among distantly related organisms (Gilbert et al., 1996; Held, 2010). For instance, although mammalian teeth vary considerably in size and shape, the same processes underlie the formation of enamel and the delineation of cusps in all mammals (Mccollum and Sharpe, 2001), which is key to recognizing homologies (Hall, 2007).

But just how such homologous features (such as jaws) can be made to look different even though they develop through homologous processes is a central question in evo-devo. This question is often framed in terms of how species-specific morphologies

arise: species may also resemble one another at some point early in ontogeny postnatal but thereafter follow divergent ontogenies leading to differences between adults (Fig. 2.1A; Gould, 1977; Alberch et al., 1979; Cobb and O'Higgins, 2004). Some species differences may be established early in life (e.g. embryonically), but then follow basically the same trajectories of size/shape change postnatally (Fig. 2.1B).





The fossil record necessarily limits studies of ontogenetic evolution to the postnatal growth period, and so the importance of pre- versus postnatal growth is the dichotomized framework employed by many evo-devo examinations of hominid craniofacial variation (e.g. Ackermann and Krovitz, 2002; Zollikofer and Ponce de León, 2004). This framework reasonably assumes that evidence of parallel postnatal ontogenetic trajectories between different species is evidence that prenatal growth is the critical period for determining species morphology. And in fact, a good deal of these studies have indicated that morphological differences between species or populations are present either prenatally or early in postnatal morphology, rather than arising later during growth (Daegling, 1996; Ponce de Leon and Zollikofer, 2001; Ackermann and Krovitz, 2002; Mitteroecker et al., 2004; Zollikofer and Ponce de León, 2004; Rook and

O'Higgins, 2005; Bulygina et al., 2006; Mcnulty et al., 2006; Fukase and Suwa, 2008). Analyses of prenatal skeletal growth are rare, but do suggest that a great deal of morphological differentiation between species occurs during this period (Zumpano and Richtsmeier, 2003; Coquerelle et al., 2010a; Rafiq et al., 2012).

A major limitation to most previous studies of early hominid craniofacial growth (Richtsmeier and Walker, 1993; Ackermann and Krovitz, 2002; McNulty et al., 2006) is that the nature of the fossil record limits many analyses to inferences of growth between a single developmental stage and adulthood. These sole-subadult samples are usually older than infants (e.g. Taung, KNM-WT 15000). This limited sampling almost certainly overlooks major changes in growth patterns that are evident from longitudinal studies, such as are known to happen during human growth in stature and body mass (Tanner, 1951; Hochberg, 2012).

The present study, utilizing the relatively complete ontogenetic series of *A*. *robustus* mandibles, has the advantage of sampling multiple points in ontogeny and thereby provides a more nuanced glimpse into just how similar or different postnatal growth patterns truly are between humans and an early hominid. Also, some age groups (see Chapter 3) are represented by more than one *A*. *robustus* mandible, which sheds some light on variation in patterns of growth in this species; this is important since studies of fossil ontogeny suffer from limited sample sizes, underestimating variation within and between age groups (Cobb and O'Higgins, 2004).

Adult mandibles of humans and *A. robustus* are quite different in size and shape, and so an ontogenetic comparison allows one to examine just how these species differences arise. Although these species' mandibular corpora overlap in height, the *A*.

robustus corpus can be twice as broad as humans' (Daegling 1989). The *A. robustus* ramus is also absolutely and relatively taller than humans' (Broom and Robinson, 1952; Bromage, 1989; Rak, et al. 2007; Rak and Hylander, 2008). To be sure, each species' mandible is adapted to generating and withstanding quite different masticatory forces. Nevertheless, all hominids share early canine development and eruption contrasting with living apes (Mann, 1975; Simpson et al., 1990; Smith, 1994). Small anterior tooth size, coupled with anteriorly-placed zygomatic bones, give *A. robustus* a relatively flat ("orthognathic") face, similar to humans but in contrast to earlier australopithecines (Simpson et al. 1990; McCollum 1999).

The similarities between these two hominids – relatively small anterior teeth and a non-projecting face – may arise through similar developmental mechanisms and processes. Facial growth and development occur as the hard and soft-tissues of the face enlarge and are displaced away from the brain case (Moss and Young, 1960; Enlow and Hans, 1996). In order to compensate for this displacement and maintain functional coherence of facial components, bone deposition occurs at facial sutures, and the outer surfaces of bone remodel by deposition and resorption throughout ontogeny. Because humans and *A. robustus* have small canines and incisors, their faces do not need to undergo as much anterior growth as apes or gracile australopithecines (Simpson et al. 1990). As a result, the incisive suture, which allows the anterior face to grow forward, fuses relatively early in humans and *A. robustus* (Braga, 1998).

Patterns of periosteal facial bone remodeling are also similar between *A. robustus* and modern humans, shared to the exclusion of 'gracile' *Australopithecus* and early *Homo* (Bromage, 1989; McCollum, 2008). For example, the anterior surface of the

mandible is largely depository throughout ontogeny in chimpanzees (*Pan troglodytes*; Johnson et al. 1976) and non-robust hominids (Bromage, 1989), reflecting these species' prognathism and predominantly forward direction of facial growth. The anterior alveolar surface is fully resorptive in humans (Enlow and Harris, 1964), and in *A. robustus* this surface is depository medially but resorptive laterally (Bromage, 1989). As with the incisive suture, this locus of resorption reflects the fairly orthognathic face in humans and *A. robustus*. Thus, sutural and periosteal growth processes of the anterior face seem to track the size and development of the anterior teeth (Simpson et al., 1990). These similarities in form and process lead to the hypothesis that anteroposterior length dimensions of the jaw including the symphyseal region should follow similar patterns of size change in each species, at least until the eruption of the permanent incisors.

Whether such "morphogenetic interpretations" of hominid facial remodeling fields can be translated to global ontogenetic changes (Bromage 1989: 759), or rather whether these fields reflect merely "local surface sculpting" (McCollum, 2008: 12), has not yet been tested. O'Higgins and Jones (1998) could not reject the hypothesis that the position of growth fields of bony remodeling across ontogeny is mirrored by facial shape change during growth in the monkey *Cercocebus torquatus*. Further complicating our understanding of craniofacial growth, the relationship between growth fields and sizeshape change has been called into question because of great intraspecific and intrasexual variation at any given age (Wealthall, 2002; Mccollum, 2008). In addition, McCollum (2008: 12-13) argued that the remodeling fields on the maxilla of *A. robustus* reflect local topographic change, and are not responsible for creating the orthognathic facial profile superficially similar to modern humans. While the present study cannot directly examine

this question, if the results indicate divergent ontogenies in light of the similar distribution of growth fields in *A. robustus* and humans, this would undermine a causal relationship between superficial growth fields and global shape change.

The gross differences between *A. robustus* and human jaws also inform predictions about growth patterns. During the ontogeny of humans and apes, the height of the posterior face increases more than the front, resulting in the apparent rotation of the mandible relative to the cranial base (Björk, 1955; Björk and Skieller, 1972; Bromage, 1989; McCollum and Ward, 1997; Wang et al., 2009). As a result, the angle between the (horizontal) corpus and (vertical) ramus usually decreases across ontogeny. Because adult *A. robustus* have very tall rami (Broom and Robinson 1952; Rak and Hylander, 2008), Bromage (1989) reasoned that *A. robustus* likely experienced greater mandibular rotation during postnatal ontogeny than do humans (Fig 2.2). This hypothesis is difficult to test given the lack of rami in the fossil sample, but may be addressed roughly using other measures of facial height. Similarly, the thick corpus of *A. robustus* could be a characteristic that is present in the youngest infants, or may it arise later during postnatal growth.



Figure 2.2. Schematic of facial rotation in *A. robustus* (Fig 3. from Mccollum 1997). The mandibular ramus (not shown, but in the area of the left-most arrow) increases in height relatively more than the mandibular symphysis anteriorly (in the area of the smaller middle arrow).

The present study therefore contributes important information to evo-devo debates about how and when differences or similarities in species morphology arise. While the most proximate bases of development, such as bone and tooth morphogenesis, are known to be conserved across taxa, the question is open as to whether more global aspects of facial growth are similarly conserved. The null hypothesis of conserved developmental patterns in human and A. robustus makes testable predictions. A first prediction is that no species differences in size change should be found for any trait. Second, if differences are found, they should occur across all traits (or at least most of them), and be of similar magnitude. In this scenario, ontogenetic differences are heterochronic, e.g. related to shifts in the rate/timing of development (Gould, 1977, 2000), rather than absolutely deviations in how and where bony growth occurs. Contrarily, the rejection of the null hypothesis would support the **alternate hypothesis of divergent** ontogenetic trajectories. This alternate hypothesis is rather open and so makes few specific predictions, except that that the species differences described above should be reflected in only a subset of traits.

2.3 Life History Theory

The analysis of mandibular growth in *A. robustus* will also address the evolution of human life history. This section begins by reviewing the role of growth in life history. I then discuss the more and less unique aspects of modern human life history. Next, I review the fossil evidence for the evolution of human life history and growth. Finally, I describe why the present study address and/or overcomes many of the shortcomings of

previous studies, and conclude with predictions about the outcome of this analysis based on life history theory.

Growth is more than merely the means by which adult form arises, it is a key aspect of an organism's life history (Smith and Tompkins, 1995). Life history theory stipulates that the timing of major events in an organism's life is part of an optimal strategy to survive and reproduce. This assumes an organism will only have limited energy during its lifetime, and this must be partitioned among requirements of growth, maintenance and reproduction (Smith and Tompkins, 1995). Given the necessity of these tradeoffs, it should be possible to predict patterns of covariation between these variables. Because human life history is remarkable in a number of ways, situating growth in this context can thus help us understand the evolutionary origins of the human adaptive strategy.

Bodily growth is important for life history because it is a process that requires a great deal of energy from outside the organism. This is notably problematic because a developing animal is dependent on available resources, and must also allocate energy to fight infection, forage, develop the physical (e.g. neural) and practical bases for social life (among primates at least), travel, and avoid being preyed upon (Smith and Tompkins, 1995). The energetic requirements of growth are thus shared with other biological systems, and an individual's optimal adaptive strategy at any given time may be to divvy resources between these systems differently than at other times. The changing challenges and advantages to survival that arise during ontogeny theoretically necessitate that a species' ontogeny be well suited to its ecological and social circumstances. As such, an

animal's energy expenditure toward growth, specifically rates and/or duration thereof, must be in accord with these changing ecological and adaptive demands.

Human Growth and Life History

The human adaptive strategy, relying on symbolic language and culture, distinguishes us from our primate relatives, and this uniqueness is marked in several aspects of our biology. Most notably, our brains are much larger than expected for our body size, and account for upwards of 20% of an adult human's basal metabolic rate compared with less than 10% in other Primates (Leonard and Robertson, 1992, 1996; Aiello and Wheeler, 1995). This expensive organ both allows and necessitates our sociolinguistic strategy, and is almost certainly related to our pattern of growth: while primates in general grow slowly compared with other mammals of comparable size (Charnov and Berrigan, 1993; Kaplan et al., 2000; Walker et al., 2006 b), humans grow most slowly of all, having both the absolute longest ontogeny of any primate, as well as a longer ontogeny than expected for a primate of our size (Bogin and Smith, 1996; Leigh, 2001). This protraction of maturation is not equally distributed across the subadult period. Rather, human maturational stages and growth rates appear to reflect the energetic requirements of developing the brain, or using it to develop social-linguistic skills critical to being a successful adult (Allman and Hasenstaub, 1999).

The neonatal human brain averages 382 cm², only some 30% adult size at birth compared with 40% adult size in our closest relatives, chimpanzees (DeSilva and Lesnik, 2006, 2008). This smaller proportion nevertheless equates to an *absolutely* larger size than a neonatal chimpanzee's. To reach an average adult size of over 1200 cm³, human

infants retain rapid (fetal) brain growth rates into the first year postnatal life (Dobbing and Sands, 1979), which then sharply decelerate soon thereafter (Coqueugniot et al., 2004; Leigh, 2006; Coqueugniot and Hublin, 2012). Growing such a large brain can require up to 60% of an infant's energetic expenditure (Leonard et al., 2003), and to offset some of this cost human infants are absolutely and relatively fatter than most other species (Leonard et al. 2003; DeSilva, 2011; Hochberg 2012). In light of this high cost, humans tend to wean offspring at a relatively young age: 2.5 years on average (between 2-4 years globally) among human natural-fertility populations compared with at least four years among living apes (Kaplan et al., 2000; Kennedy, 2005). Bodily growth rates of human infants, while relatively high, are rapidly decelerating during this time (Bogin, 1999; Hochberg, 2012).

Infancy ends with the completion of weaning and is followed by the childhood stage (Bogin, 1999, 2009). One thing that makes this period unique to humans is that the weaned child has only its deciduous teeth, whereas in most mammals completed weaning roughly coincides with the full eruption of the first permanent molar tooth (Smith and Tompkins, 1995; Bogin, 1999). Most other animals are able to forage for themselves at this stage of maturation, but a growing child is poorly equipped to handle adult foods and is highly dependent upon others for acquiring and preparing its food. The child's costly brain, however, is still growing, albeit not as rapidly as during infancy (Leigh, 2006). Growth of the body slows significantly and stabilizes to a relatively low level during this period (Hochberg, 2012). Because of these features, the childhood period has been argued to be uniquely human (Bogin, 1999).

Childhood ends with the eruption of the first permanent teeth (first molars and incisors; Bogin, 1999), usually between the ages of five and seven years in humans (Liversidge, 2003). Bodily growth rates, decelerating through infancy and childhood, reach their lowest levels during this time, the juvenile stage. The human brain has essentially finished *size* growth during by this time (Leigh 2006). Nevertheless, this is still an important period for neural and cognitive development in humans (Campbell, 2006, 2011), coinciding with the behavioral period known as 'middle childhood' to developmental psychologists (Monge and Mann, 2010). Around the time the brain stops growing, the *zona reticularis* within the adrenal gland matures, beginning a steady increase in circulating levels of the neurotransmitter dehydroepiandrosterone (DHEA) and DHEA sulfate (DHEAS; Campbell 2006). This shift in hormone levels is referred to as adrenarche.

Campbell (2006, 2011) notes that DHEA/S are involved in the development of the cerebral cortex from this time until early adulthood. This is thus an especially important time for a child's social development (reviewed in Monge and Mann, 2010). Juveniles begin to acquire reasoning skills, and an understanding of self and the difference between right and wrong, and they begin forming relationships with people other than their parents. In short, this is the time when children begin learning social norms and become integrated into society. Human middle childhood (juvenility) has two other unique correlates with life history. First, the period lasts about 1.5 times longer in humans compared with chimpanzees, providing youngsters with extra time to develop important social, linguistic and foraging behaviors (Kaplan et al., 2000). In addition, this time coincides with a "quiescent period" (or hiatus) in dental development exclusively seen in

humans (Eveleth and Tanner, 1988) and fossil hominids (Mann 1975; Monge and Mann, 2010).

Middle childhood (juvenility) ends with the onset of puberty, the activation of the hypothalamic-pituitary-gonadal axis, resulting in high levels of sex hormones (Hochberg 2012). This marks the final stage of ontogeny, adolescence, which spans the 5-10 years between puberty and adulthood. Adolescent growth contrasts with the generally slow bodily growth during the childhood and juvenile stages. Rather, humans experience a brief growth spurt during which linear skeletal growth rates are at their highest since infancy (Eveleth and Tanner 1988; Bogin 1999; Walker et al. 2006b), shortly after which growth in stature is complete. While many primates experience a spurt in body *mass* around puberty (Leigh 1996), none experiences the rapid increase in *skeletal dimensions* seen in humans (Watts and Gavan, 1982; Bogin, 1999; Hamada and Udono, 2002).

Adolescence is not only important for somatic growth, but it also marks sexual and neural development and a final time when youths can learn critical social behaviors (Bogin and Smith, 1996; Bogin, 1999, 2003, 2009; Locke and Bogin, 2006). Many though not all brain centers, such as the fornix, finish maturing during this time (Hochberg 2012). Adolescents tend to take greater interest in adult activities (Kaplan et al. 2000) and adopt more social uses of language (Locke and Bogin, 2006). Bogin (2003) has further argued that sex differences in the timing, intensity and duration of the adolescent growth spurt promote each sex's "sociosexual" development, facilitating a smooth transition to adult social status. The adolescent growth spurt occurs earlier in girls than boys, making the former appear relatively more mature and so helping them enter into adult social, sexual and economic spheres. Boys, on the other hand, usually begin the

spurt later and it lasts longer compared with girls. This may serve to make adolescent boys' social behavioral mistakes better tolerated by adults who perceive them as boys when they are trying to act more like men.

Relevant to this study, the most remarkable aspect of human ontogeny and somatic growth compared with other primates is our exceptionally long childhoodjuvenile period (Leigh and Park, 1998), with concomitant delay in the onset of reproduction: age at first birth averages around 20 years in human many societies, compared with just under 15 years in chimpanzees (Bogin, 1999; Kaplan, et al. 2000). (Some human societies, however, reproduce at earlier ages comparable to chimpanzees, e.g. Filipino Aeta: Migliano et al., 2007; Venezuelan Pumé: Kramer and Greaves, 2010) A long, drawn-out pre-reproductive period is risky from an evolutionary perspective as the delay increases the chance of dying before reproduction, and it may potentially reduce the duration of the reproductive career (Crespi, 2011). Despite the risk of this strategy, the lifetime fitness (i.e. reproductive) consequences can be compensated by our potential to have multiple dependent offspring of different ages (Gurven and Walker, 2006), to reduce the interbirth interval by early weaning, and to have greatly reduced (especially subadult) mortality (Kaplan et al. 2000). What is further unique about this strategy is that the energetic burden of caring for developing youths is not confined to the parents, but older siblings and other family members usually play a crucial role (Hawkes et al., 1998; Kaplan et al., 2000; Bogin, 2003).

In conclusion, human bodily growth is inextricably connected with the energetic and cognitive requirements of developing a large brain capable of high cognitive (i.e. social) function. In the early subadult period (i.e. infancy and childhood) energy is

preferentially devoted to the brain over the body. Bodily growth rates remain low during the subsequent juvenile period, partly in order to provide ample time to begin practicing complex social and foraging behaviors, and partly to offset the energetic costs for caretakers. This 'middle childhood' is essential to humans' sociocultural adaptation. The final stage of growth, adolescence, includes a brief 'spurt' of growth to attain adult body size and is important for youths' incorporation into adult society. Despite the importance of the human pattern of bodily growth to our adaptive strategy, it is still unclear just how early in hominid evolution this pattern arose, or whether all aspects of this pattern appeared at once or piecemeal over the course of evolution.

Early Hominid Life History

In the absence of complete fossil hominid populations, researchers have used a number of lines of evidence to make inferences about the evolution of hominid life history and growth. Lovejoy (1981, 2009) hypothesized that the defining feature of hominids, bipedalism, accounted for our origins and evolutionary success because it increased offspring survivorship. Lovejoy noted that living humans have the potential for higher lifetime fertility than the great apes for two main reasons. First, great apes have a much higher juvenile mortality rate than modern humans. Second, great apes have fairly long intervals between successful births compared with modern human hunter-gatherers. While these are the central tendencies for these species, it should be noted that there is overlap between the ape and human interbirth intervals (Kaplan et al. 2000; Robson and Wood 2008). The ape pattern results in their relatively low reproductive rates and slow population growth, which can create substantial adaptive hurdles in seasonally fluctuating

ecosystems like those of the late Miocene (Lovejoy 1981). Basal hominids, Lovejoy argued, overcame this 'demographic dilemma' with a novel strategy in which a male provisioned a monogamously pair-bonded female and their dependents (i.e. a nuclear family) with resources. Familial provisioning allowed hominids to wean infants early and reduce their interbirth interval (Kaplan et al. 2000; Kennedy 2005; Humphrey 2010). This system implies that early hominids would have cared for multiple offspring at a time.

Under the cooperative breeding and provisioning implied by Lovejoy's (1981) model, selection would favor slow growth rates during childhood to offset the energetic demands of dependent offspring on (provisioning) parents (Gurven and Walker, 2006), especially at hominids' relatively large body sizes (McHenry and Coffing, 2000). Great ape growth rates, on the other hand, are relatively higher than humans' throughout a shorter ontogeny (Walker et al. 2006a). Thus, unique aspects of *modern* human life history are hypothesized to be in place at the onset of hominid origins. Consistent with this hypothesis, DeSilva (2011) showed that humans give birth to relatively large neonates, up to 5-6% of mother's mass compared to only around 3% in chimpanzees. Estimating neonatal and adult female body mass for extinct hominids, DeSilva further found that relatively large (i.e. human-like) neonates characterized hominids as early as Australopithecus afarensis (3-4 million years ago). Thus, one aspect of the human life history strategy (birthing large infants) seems to have been present very early in human evolution, raising the possibility that other behavioral and ontogenetic aspects of the strategy may have been in place as well.

Contrary to this prediction, however, many researchers have argued that early hominids would *not* have experienced a human-like pattern (e.g. duration or rates) of

growth. First, early hominids' diminutive adult brain sizes, less than 600 cm³ compared with over 1200 cm³ in living humans (Holloway, 2008; Coqueugniot and Hublin, 2011), hint that early hominids would not have had a long growth period like humans. The human adaptive strategy relying on language and culture necessitate large brains for the storage and processing of such complex information (but see Brown et al., 2004, for a recent, small-brained tool-maker). This cognitive skill set is hypothesized to be related to the prolongation of human growth, as it will take a long time to learn the complex skills necessary to be an at least competent member of society (Mann 1975; Allman and Hasenstaub, 1999; Bogin 1999; Kaplan et al., 2000; Bogin, 2003; Bogin and Locke, 2006). Australopithecines' smaller brains suggest a lack of human-like behavioral capacities, meaning a prolonged subadult period would be both unnecessary for survival and evolutionarily risky. Alternatively, slow and prolonged juvenile growth could have been required for australopithecines to learn complex behaviors *because* of their small brains (i.e. learning may take longer; Wolpoff 1999).

A fast, non-human-like ontogeny was further argued for australopithecines based on the observed correlation between brain size and various life history parameters at supra-generic taxonomic levels (e.g. BH Smith 1989). Sacher (1978) found that variation in adult brain size appeared to be explained in part by variance in lifespan across mammals. Building on Sacher's work, BH Smith (1989, 1991; RJ Smith et al., 1995) further explored the correlations between brain size and other life history variables in primates. Brain size was found to correlate fairly highly with age at M₁ eruption (r = 0.99among anthropoid species; Smith 1991), which roughly marks the end of weaning in mammals except humans. Smith (1989, 1991) used the relationships between brain size

and other life history variables to predict that early hominids' small brains imply that they would have erupted their M₁s and weaned at younger ages than humans, with the implication that overall growth was like non-human apes. RJ Smith et al. (1995) came to similar conclusions, however they urged caution when using such correlations to predict hominid life history because of error inherent in statistical models and the inescapable uncertainty of how early hominids would truly fit linear models based on a wide range of living species.

In addition to the correlation between age at M₁ eruption and adult brain size, teeth have been used extensively to make inferences about the evolution of hominid growth and life history. This is due in part to the fact that teeth and jaws are some of the most durable and so well-represented elements in the fossil record, as well as the fact the fact that teeth form and erupt in predictable patterns making it possible to estimate individuals' relative stage in ontogeny, if not chronological age. An assumption of these studies, usually implicit but sometimes explicit (Bromage and Dean, 1985), is that anywhere dental development is more like modern humans or apes, bodily or skeletal growth is also more like that group (Conroy & Vannier, 1991a-b; Smith, 1994).

Human dental development is distinguished from that of the other apes by 1) first permanent molar formation and eruption at roughly the same time as the permanent incisors, as opposed to long before the incisors; 2) a canine that forms and erupts prior to the second molars, as opposed to coincident with or after the third molar (M₃ or "wisdom tooth"); and 3) delayed molar formation, especially of the M₃, such that the formation of adjacent molar crowns do not overlap with one another (Mann, 1975; Anemone et al., 1996; Kuykendall, 1996).

Evidence for dichotomized 'ape-like' and 'human-like' patterns have been sought in relatively complete dentitions of early hominids. Mann (1975) examined the sequence of dental development in *A. robustus* from Swartkrans, and found he could not distinguish the fossil pattern from that of humans. The *A. robustus* sample showed a human-like pattern of spacing between molar crown formation and eruption (viz. M₃). Mann therefore concluded that *A. robustus*, and presumably early other hominids, probably had prolonged maturation like humans, but this was soon called into question.

First, patterns of tooth formation in (non-robust) early hominids were argued to match chimpanzee standards better than humans (although none of the studies were directly comparable to Mann's analysis of the Swartkrans fossils). If a fossil juvenile preserves more than one developing tooth, one can examine each tooth's development relative to the other teeth, and these patterns can then be compared with known patterns and sequences of crown formation in living apes and humans. From here, Smith (1994b: 308) reasoned that, "developmental scores of an individual's teeth are...more likely to agree when scores are assigned from standards based on the individual's correct species or group." For fossils, this implies that better conformation to one species' dental standard equates with overall (e.g. somatic) developmental similarity to that species.

Smith (1986, 1993, 1994a-b) found that teeth of gracile australopithecines and early *Homo* tended to provide more similar age estimates when using chimpanzee compared with human standards, while *A. robustus* and *boisei* actually conformed better to the human standards. The principle deviation of the gracile australopiths from the human pattern was that permanent incisor and canine formation was delayed relative to first molar development and eruption (Conroy and Vannier 1991a-b), e.g. point 1 two
paragraphs above. However, as Simpson and colleagues (1990) point out, this probably due to the larger anterior teeth of these species compared with humans and robust australopithecines, rather than to each species' rates of bodily growth. Moreover, several patterns said to be ape-like, and found in gracile australopiths (i.e. delayed eruption of the permanent incisors relative to first molars, or canines relative to second molars), do not actually completely distinguish apes from humans: while these may be rare in the Euro-Americans on which human standards are developed, they are not uncommon in other populations (Garn and Koski, 1957; Dahlberg and Menegaz-Bock, 1958; Mann, 1988; Wolpoff et al., 1988; Liversidge, 2003). Key to this discussion – Mann's (1975) important observation of a human-like relationship between successive molar formations was never falsified.

Early hominids were also argued to show ape-like maturation on the basis of the study of dental histology. Perikymata and striae of Retzius are lines found on the surfaces of and within tooth crowns, respectively, reflecting the deposition of enamel during the formation of the crown (Guatelli-steinberg, 2009; Tafforeau et al., 2012); perikymata are merely the manifestations of striae of Retzius that reach the crown surface. If the periodicity with which these lines are formed can be known (it ranges from 6-12 days in humans and apes), it is theoretically possible to count these lines and estimate the actual time it took a tooth crown to form (Bromage and Dean, 1985). These crown formation times can then be inferred to reflect overall maturation period. As such, several studies have estimated shorter crown formation times in early hominids compared with humans, and therefore inferred that prolonged maturation had not evolved by the early Pleistocene

(Bromage and Dean, 1985; Beynon and Dean, 1988; Dean et al., 2001; Lacruz et al., 2006, 2008).

A major problem with this technique is that the link between (estimated) tooth formation and growth rates is yet to be firmly established. For instance, the earliest studies argued that incisor crown formation times were shorter in hominids compared with humans, and that they therefore experienced a rapid, ape-like pattern of growth. However, as Mann and colleagues (1990) pointed out, not only do nearly all hominid incisor perikymata counts fall within the impressive range of modern humans, but do ape incisor crowns do not form more rapidly than humans, which was key to the argument that rapid crown formation equated with rapid ontogeny. Similarly, histology-based estimates of molar crown formation times overlap between humans and apes, and in some cases ape crowns have an even longer estimated formation time than humans' (Macho and Wood; Macho, 2001; Smith et al., 2007). This is important given the staggered molar crown formation identified by Mann (1975). Thus, even if dental histology can be used to accurately estimate how long a crown took to form or to estimate an individual's age at death (e.g. Antoine et al., 2009), the relationship between crown formation time and overall maturation rate is probably not tenable.

In summary, studies of dental development in living and extinct hominids have been very informative, but insofar as they tell us anything about life history (e.g. Smith, 1991), dental data alone are not sufficient to make inferences about overall patterns of bodily growth. As Simpson and colleagues (1991: 119) state, these data tell "nothing about the pattern at which [a hominid's] component structures will grow. These are more likely to vary directly with the directional selective milieu that is responsible for their

emergent adaptations." In other words, even if early hominids formed their teeth relatively rapidly, this alone cannot be used to distinguish ape or human patterns of somatic growth.

Skeletal Growth in Early Hominids

Rather than extrapolating growth inferences from teeth or brains to the rest of the body, it may be more useful to examine actual patterns of bony size change across ontogeny. To date, however, there are only a handful of studies of growth in early hominids (i.e. *Australopithecus* and early *Homo*) that have actually utilized *skeletal* size data in conjunction with dental indicators of age. While these studies have been informative in their own right, they nevertheless have various limitations. In this section, these studies are summarized and critiqued, ending with a summary of how the present study builds off of and improves upon these previous works.

Some evidence about ontogenetic size change in *Australopithecus* comes from pelvic remains fossils of *A. africanus*. Berge (1998) compared of iliac growth in African apes (including humans) and *A. africanus*. Significant from a life history perspective, she noted that the adolescent female *A. africanus* ilium (represented by MLD 7) is essentially the same size as what she then considered a young adult female (Sts 14); because this pattern of age-related size variation was common in her ape but not human samples, Berge reasonably concluded *A. africanus* did not follow a human-like pattern of size (or shape) growth. However, Robinson (1972: 76) noted that the distance from the acetabulum to the ischial tuberosity was only 8.0 mm compared with Sts 14's 16.1 mm, raising joint possibility of both a large amount of remaining growth and/or individual

variation in ischial length. Berge and Gommery (1999) later described the Sts 14 pelvic bones as showing several markers of immaturity (viz. unfused epiphyses). The revised assessment of skeletal maturity for Sts 14 both lowers the age disparity between it and MLD 7, and leaves open the question of just how much growth was left in either of these subadults.

Tardieu's (1998) analysis of African ape and early hominid distal femoral growth also suggested early hominids differed from living humans. Tardieu found that distal femora of East African australopithecines (*A. afarensis* and *A. boisei*) do not display the 'squared' shape characteristic of human adults, and which arises through adolescent growth. Rather, these australopiths' distal femora more closely resembled human children. Fossil femora attributed to the genus *Homo*, on the other hand, displayed a more adult human-like morphology. Tardieu thus concluded that australopithecines likely did not experience a human-like adolescent spurt in skeletal growth. While this inference seems reasonable given her fossil sample, it can only be fully tested with an ontogenetic series of femora that the fossil record currently does not afford.

Many studies other of skeletal growth and development of hominids in the early Pleistocene have necessarily focused on a single specimen, KNM-WT 15000. This is one of the only nearly complete skeletons of *Homo erectus*, belonging to an adolescent male from around 1.5 million years ago (Brown et al., 1985; Gathogo and Brown, 2006), slightly postdating *A. robustus*. Well-preserved adult and adolescent partial skeletons from the 1.8 million year old site of Dmanisi are known (Lordkipanidze et al., 2007), but these have yet to be used to assess postcranial growth in this sample (but see Van Arsdale et al., 2012 for treatment of skeletal maturation).

BH Smith (1993) compared age estimates for the KNM-WT 15000 based on published skeletal and dental maturation standards for modern humans and chimpanzees. Estimating this individual's age using ape skeletal and dental standards each provided fairly consistent results, around 8 years at death. Human standards, however, gave very different results: a human showing the same stage of dental development as WT 15000 would tend to be 10-11 years old, while a human of the same skeletal development would be closer to 13 years. Smith took the concordance of the estimates based on the chimpanzee model, and discordance of estimates based on a human model, to indicate that overall growth in *H. erectus* would have been chimpanzee-like.

While interesting, Smith's (1993) study could not fully statistically take into account variation *within* each species, as the skeletal and developmental standards she used were derived from single population averages that do not incorporate natural intraspecific variation, which is known to be extensive in humans (Eveleth and Tanner, 1988; Liversidge, 2003; Walker et al., 2006 a). The "fit" of WT 15000 to either ape or human standards is difficult to interpret without knowing just how much overlap there is between chimpanzee and human patterns of skeletal versus dental maturation (i.e. Wolpoff et al., 1988; Zihlman et al., 2004; Monge et al., 2007). Moreover, the human skeletal and dental standards used were developed on different samples, and so it was unclear just how modern humans vary in the relationship between dental and skeletal maturation.

To this end, Clegg and Aiello (1999) and SL Smith (2004) examined the disparity between skeletal and dental maturation in samples of recent humans whose actual chronological ages were known. Both of these studies found that the 2 year disparity

between skeletal and dental age estimates seen in WT 15000 were not uncommon in modern humans, although it was commoner for dental to be advanced over skeletal development, the opposite of the fossil. Clegg and Aiello's (1999) study has since been criticized for utilizing broken archaeological specimens (Dean and Smith, 2009), but SL Smith's (2004) findings remain unchallenged. Her results corroborate other studies showing that skeletal and dental maturation are not especially tightly linked, and so estimates of an individual's age using both systems will often provide different results (Lewis and Garn, 1960; Cardoso, 2007).

It is further problematic to make the assumption that early hominid fossils will conform to either a chimpanzee or a human model of development only. While either is a reasonable and testable null hypothesis, it is not clear which model, if either, actually best fits early hominids. As a result, some studies run their analysis twice - once assuming a human-like pattern of growth and development and again assuming a chimpanzee model. For example, Antón and Leigh (2003) examined growth rates of facial dimensions in small samples of *Homo erectus*, specifically looking for evidence of a human-like acceleration at adolescence. They plotted the facial height measurements against age, once assigning ages based on a human model of dental development, and then again with a chimpanzee model. They found negative/ambiguous evidence for an adolescent acceleration if H. erectus erupted their teeth at the same ages as humans, but a chimpanzee-model of dental development, rather ironically, suggested a marked acceleration in facial growth at adolescence. This is because their dependent variable (facial height) stayed the same while the independent variable (age) was shortened in a chimpanzee model relative to a human one; humans experience developmental

milestones on average around 1.8 times later than (captive) chimpanzees (Smith 1993). If *H. erectus* was ape-like in dental development then the species may have been human-like in skeletal growth, whereas a human-like pattern of dental development would imply a chimp-like (more linear) pattern of skeletal growth.

In a statistically rigorous analysis of mandibular growth in an early Pleistocene hominid, Van Arsdale (2006) examined patterns of age- and sex-related variation in the early *Homo erectus* mandibles from the 1.8 million year old site of Dmanisi. This sample contains a large adult (D2600, one of the largest Pleistocene *Homo* mandibles), and both an early and a late adolescent. Using resampling methods similar to those employed in the present study, Van Arsdale found that age- and sex-related variation in ten out of 31 traits in the Dmanisi mandibles exceeded levels of variation in human and chimpanzee samples. This finding is consistent with other studies of adolescent growth in the face of *H. erectus*, concluding that a good deal of size change occurs between adolescence and adulthood in the species, possibly indicating a human-like growth spurt (Richtsmeier and Walker 1993; Antón and Leigh 2003).

While the ape-human maturation rate contrast is useful from a comparative standpoint, it also sets up a false dichotomy between the biology of different groups (i.e. Simpson et al. 1991). Traditionally in anthropology, humans have been said to have a "slow" life history (Walker et al., 2006a) compared to apes and others with a more fast-paced life cycle. While this is true for many of the populations studied from developed nations, not all human populations follow the purportedly slow human pattern (Walker et al., 2006; Migliano et al., 2007 Kraemer and Greaves, 2010). Regardless, the resulting expectation for fossil hominids, then, has often been either an ape-like life history, or one

intermediate between (some) humans and chimpanzees (Smith, 1993; Bogin and Smith, 1996; Bogin 1999; Dean et al. 2001; Graves et al., 2010). But as shown for *H. erectus* (Van Arsdale 2006), mandibular growth between adolescence and adulthood exceeded the variation seen in human and chimpanzee samples, and so could not be said to be like either species. It is therefore timely to reappraise hominid skeletal vis-à-vis dental development.

This dissertation will therefore contribute important information about the antiquity of human life history. From this perspective, human growth is most remarkable in that there are a number of growth stages each with its own characteristic rate of size change. As in section 2.1 (evo-devo), the **null hypothesis of identical patterns of growth (and therefore life history) between humans and** *A. robustus* predicts that each species should undergo the same amount of *overall* size change between eruption stages. Rejection of this null hypothesis would support the **alternate hypothesis that patterns of overall size change differ between these species**. Again, as in section 2.1, this alternate hypothesis is rather open and makes few specific predictions. As human growth has been argued to be especially unique in featuring childhood and adolescent periods of relatively slow and rapid growth, respectively, a reasonable alternate hypothesis predicts that *A. robustus* should evince greater size change between eruption stages prior to puberty (i.e. M_2 eruption; Chapter 3), and less size change thereafter.

2.3 A Small, Cross-Sectional Sample

Although the Swartkrans *A. robustus* mandibles are the most complete subadult age series of any part of the skeleton in the fossil record of early Pleistocene hominids,

the sample still presents special analytical challenges. Fossil assemblages are necessarily cross-sectional, sampling different individuals of varying ages (i.e. a cross-section of a population), in contrast with longitudinal data that follow individuals across multiple time points. Analytically, cross-sectional data charting change in size with age assume that the younger individuals could have 'grown into' the older (Fig. 2.3). With cross-sectional data, the many sources of individual variation – such as sex, health and idiosyncrasy – could obscure individual patterns of growth, most notably dampening the effects of changes in growth rate (Leigh, 1996; German, 2004). Comparisons can also imply unrealistic patterns of growth for a population (e.g. the dashed blue line in Fig 2.1). Adding to difficulties, fossils are differently preserved, so not all individuals can be compared for all the same (measurable) features.



Figure 2.3. Sampling possibilities created by cross-sectional data. Data are humans only. The solid black line is the lowess regression fitted to the data, while the dashed blue line is one of many hypothetical growth trajectories implied by cross-sectional sampling.

Nevertheless, cross sectional data are preferable for this study. It is logistically

very challenging to amass large longitudinal datasets, and so the sample sizes afforded to

cross-sectional data make them likelier to indicate the central tendencies of a population (Dahlberg and Menegaz-Bock, 1958). This is important, as the individual (longitudinal) variation in growth alluded to above precludes the specification of any single human growth model; there are myriad ways and reasons fossil variation *could* differ from any humans (e.g. Lee and Wolpoff, 2005), just as there are myriad ways humans might differ from one another. Related, the relatively small fossil sample size raises the issue of how well it may represent the species' normal ontogeny. The null approach adopted here is ideal because it simply asks whether the fossil pattern of variation is likely to be found within a larger, similarly cross-sectional referent.

These aspects of fossil preservation and sampling led to the development of a new method to compare patterns of growth in cross-sectional samples with missing data (Chapter 3). The null hypothesis predicts that between successive dental eruptions, human and fossil jaws will show the same proportional size difference. Statistically, this predicts that the size difference between any differently aged pair of *A. robustus* mandibles is likely to be seen in a randomly-selected pair of humans at comparable stages of dental development as the fossils. Because specimens cannot always be compared for the same sets of traits, pairwise comparisons are critical to maximizing the information from the fossil sample (Lee and Wolpoff, 2005).

In sum, this research adopts the null approach to comparing ontogenetic variation in humans and *A. robustus*. Cross-sectional data, such as are only available for extinct species, have the downside of potentially obfuscating individual patterns of growth, but they are ideal for statistically comparing patterns of variation between samples. Limiting the comparison to humans only yields a straightforward and interpretable result, which is

important in an ontogenetic study already making multiple comparisons between several different age groups. The human comparison directly addresses the question of whether one can identify growth correlates of *human* life history early in the fossil record. If this null hypothesis cannot be rejected then comparison with other species tells nothing about the evolution of human life history.

Conclusion

The fossil record is critical for reconstructing and understanding the evolution of human growth and development. The fragmentary and cross-sectional nature of fossil samples is best handled with a null hypothesis testing approach, asking whether the fossil pattern of ontogenetic variation has a high or low chance of being sampled from a large human reference population. The randomization-based comparative method to be presented herein is designed specifically to test this hypothesis and maximize the information gleaned from the fossil sample. Conceptually, this hypothesis has important implications for human evo-devo and life history.

The *A. robustus* sample contains individuals across nearly the entire subadult growth period, and so is better suited than other early hominid samples to address the evo-devo question of whether species differences and similarities arise largely during pre- versus postnatal growth, and whether species follow identical patterns of postnatal shape change. While this question has been examined for hominids previously (e.g. Ackermann and Krovitz, 2002), the current study uses a more complete fossil sample and a novel method test these alternatives more rigorously than earlier studies could.

This study will also facilitate a better understanding the evolution of human life history. Our unique pattern of growth is a key part of our life history, characterized by a prolonged period of slow growth prior to puberty followed by an adolescent acceleration. This pattern is essential for allowing youngsters the opportunity to learn necessary social skills while minimizing energy requirements. While the antiquity of the human pattern has been investigated extensively by numerous researchers, these studies were hindered by logical and methodological issues, either looking in the wrong skeletal places or utilizing insufficient samples. If the null hypothesis cannot be rejected, this may imply that the human pattern of growth was established by the early Pleistocene (although the failure to reject would not necessarily imply the converse).

The present study has many advantages over previous analyses of growth in extinct hominids. Many earlier studies did not to take into account variation within species. Some of these works fail to capture an adequate example of past population variation because they limit their analysis to one or two fossils. Many studies also failed to account for intraspecific variation by relying on dichotomous models or standards of dental development. Related, some studies did not statistically test whether hominid growth patterns could be distinguished from referent species (Cobb and O'Higgins 2004), but instead simply plotted measurements against age and then made inferences based on visual inspection of these plots (e.g. Ponce de León and Zollikofer 2000; Thompson and Nelson, 2000; Antón and Leigh 2003).

The present analysis is designed to avoid many of the obfuscating factors faced by previous studies of hominid ontogeny. By adopting a statistical hypothesis-testing framework and comparing *A. robustus* only with a large sample of modern humans, this

study is ideally poised to address questions about the evolution of hominid development and life history. The null hypothesis predicts that each species follows the same pattern of size change. For evo-devo, an alternative hypothesis is that characteristic morphologies might arise through differences in the growth of only a subset of traits. For life history, rejection of the null hypothesis would represent the first *skeletal* evidence that an early hominid may not have had the same pattern of growth as humans, a claim that at present is based almost entirely on inferences from dental development. This study will therefore make an important contribution to paleoanthropology regardless of the analysis' outcomes.

Chapter 3

Materials and Methods

The evolution of human development and life history is addressed with the hypothesis that ontogenetic variation among *A. robustus* and recent humans cannot be distinguished. This is tested by comparing patterns of size change in ontogenetic series of mandibles of each species, using a new test designed especially for small, fragmentary samples. This chapter describes the background of each of the species samples, as well as the rationale for their selection. Mandibles in each species are further subdivided into five equivalent age groups based on stages of dental eruption. I then describe the metric variables that will be compared in each sample, explaining why these data are most appropriate for the analysis. Finally, I describe the logic and logistics of the new randomization procedure used to compare patterns of age-related change in each species.

Samples

Australopithecus robustus. The fossil sample is comprised of 13 individuals spanning from infancy to late adolescence with the last permanent molar in the early stages of eruption (Table 3.1). These fossils are from the early Pleistocene site of Swartkrans in South Africa (Broom, 1949; Broom and Robinson, 1952; Brain, 1981, 2004; Sutton et al., 2009; Pickering et al., 2012). The hominid fossils studied come from Member 1 with an additional specimen from Member 2. Cave sites like Swartkrans have a great deal of uncertainty about their depositional contexts and absolute age. These caves lack volcanic sediments that can be used to estimate absolute ages with accurate and reliable radiometric techniques like argon-argon dating (Ludwig and Renne, 2000; Herries et al., 2009). Moreover, caves can be exposed at any time, raising the possibility that Swartkrans' early Pleistocene deposits may be contaminated by small infillings long post-dating the bulk of the deposit (Broom and Robinson, 1949; Brain, 1981, 2004).

Herries and colleagues (2009) recently reviewed attempts to date Swartkrans and other South African cave sites. Comparison of Swartkrans' non-hominid fauna with that from radiometrically-dated East African sites suggests the majority of Member 1 was deposited between 2.0-1.6 mya, whereas Member 2 may be closer to 1.4-1.5 mya. However, the presence of both relatively primitive and advanced taxa in Members 1-3 could indicate each deposit accumulated contemporaneously between 2-1 mya. Absolute dating techniques such as electron spin resonance and uranium-lead (U-Pb) have been employed on tooth enamel from Swartkrans fossils (Balter et al., 2008). These methods suggest that Members 1-2 date to around 1.8 and 1.5 mya, respectively, but the error ranges of these estimates are great and overlap for each member. More recently, Pickering and colleagues (2011) obtained new U-Pb estimates for the flowstones both under- and overlying Member 1, and were able to bracket the accumulation of Member 1 to between 2.33-1.64 mya.

In sum, it is quite possible that the *A. robustus* sample examined here is the result of a relatively short period of deposition, but this cannot be said for certain. Both faunal correlations and absolute dating techniques provide consistent estimates that Member 1

may be slightly older than Member 2, but the composition of each deposit is similar enough to be consistent with idea that the members being contemporaneous (de Ruiter, 2003; Brain, 2004; Herries et al., 2009). All but one of the fossils examined come from Member 1, which may be constrained to between 1.8-1.6 mya (Pickering et al., 2011). SKX 4446, the only fossil examined from Member 2, has been shown to be very similar to other *A. robustus* mandibles from the site (Grine, 2005), suggesting that even if time averaging were an issue here, it is probably not a major contributor to variability within the sample.

Swartkrans Deposit	A. robustus mandibles
Member 1	SK 6, 25, 55b, 61, 62, 63, 64, 438, 843, 1587, 3978, SKW 5
Member 2	SKX 4446

Table 3.1. Provenience and catalog numbers of *Australopithecus robustus* specimens in this analysis.

A. robustus mandibular remains are also known from the nearby site of Drimolen (Keyser et al., 2000), which was probably deposited around the same time as Swartkrans. Mandibular fossils from Drimolen will not be considered in this analysis, however, as these fossils are not yet available for study. This may not be a problem for present analysis, as Drimolen contains few subadult individuals that address ontogenetic variation in *Australopithecus robustus* (Moggi-Cecchi et al., 2010). DNH 44 from Drimolen is a right mandibular corpus and partial ramus of an individual whose dm₂ is as incompletely erupted as SK 438 from Swartkrans. Comparison of SK 438 fossil with the published pictures of DNH 44 suggests that these infants share the same size and proportions. DNH 46 is a partial right mandibular corpus with an occluded rdm₂ and RM₁. The M₁ cusps of DNH 46 are worn and flattened, compared with the freshly erupted M₁s of the Swartkrans mandible SK 63 (which is virtually complete). DNH 46 is therefore probably a little older than SK 63, and again comparisons of the latter with published

photographs of the former (Moggi-Cecchi et al., 2010) give the impression of similar overall size and shape. The only other subadult jaw from Drimolen is DNH 12, a small fragment of posterolateral right mandibular corpus of an adolescent with the RM₃ nearing alveolar eruption. No measurements can be taken on this mandible save for perhaps an estimate of the width of the extramolar sulcus. Thus, only two fossils (DNH 44 and 46) from Drimolen are potentially useful to the study of mandibular ontogeny in *A. robustus*.

It may be appropriate to omit the Drimolen fossils in any event, as they might represent a population distinct from Swartkrans. Dental remains from Drimolen are, for the most part, on the lower end of the Swartkrans range of variation (Moggi-Cecchi et al. 2010), which could reflect systematic differences in body size between hominids at the two sites. In line with this, when Keyser (2000) described the complete female skull (DNH 7) from the site, he noted it was "noticeably smaller" than SK 48 and most other Swartkrans craniodental fossils that were erstwhile considered female; Lockwood and colleagues (2007) have since suggested SK 48 and others may actually have been small adult males. Thus, it is possible although not certain that Drimolen *A. robustus* were generally smaller-bodied than those from Swartkrans, implying differences in growth and age-related variation between the two sites; this would be an interesting hypothesis to test in the future.

A. robustus specimens from Kromdraai are also omitted from this study. The type specimen of *Paranthropus* (*Australopithecus*) *robustus* (Broom, 1938) is not included because it belonged to a young adult (as judged by an M₃ in occlusion), while the present analysis focuses on subadult growth prior to occlusion of M₃. The other subadult *A*.

robustus from the site, TM 1536, is also omitted because it does not preserve measurable anatomy to be compared with the other fossils.

Modern humans. The human sample includes 122 subadults from the site of Libben in Ohio (Lovejoy et al., 1977). Libben is radiocarbon dated to between 800–1100 CE, and was probably continuously occupied during this time. The site is located on the Portage River near Lake Erie, and would have been at the edge of a swamp during its occupation. The site spans over 30,000 square feet and has yielded remains of over 1,300 individuals. Libben was an egalitarian society, and there is no indication of preferential or differential burial of certain individuals or groups (Meindl et al., 2008).

Fauna from the site point to a large animal protein component to the diet, based largely on trap-and-weir hunting of small animals and fish, and there is an odd dearth of vegetables aside from wild rice (Lovejoy et al., 1977; Meindl et al., 2008). Lovejoy and colleagues (1977: 291) report there may have been "marginal corn agriculture at some time," but this appears not to have been a major component (Meindl et al., 2008). Archaeology and paleodemography of the site indicate that adults must have endured extremely high work loads to support what was a healthy and robust population (Howell, 1982). Important for this study, growth of the long bones at Libben appears to have been comparable to other humans both prehistoric (Mensforth, 1985) and modern (Lovejoy et al., 1990).

Libben was selected as a comparative sample for three main reasons. First, subsistence was based on fishing, hunting and gathering (Lovejoy et al., 1977), so individuals' growth would not have been affected by the nutritional and adaptive changes associated with recent agriculture and domestication (Wood et al., 1992; Okazaki, 2004;

Sardi et al., 2006). The high incidence of dental crowding and malocclusion between upper and lower teeth in recent compared with prehistoric human populations has been attributed a more easily processed diet in the former (e.g. Begg, 1954; Beecher and Corruccini, 1981; von Cramon-Taubadel, 2011). Reduced masticatory demands for recent humans fail to stimulate adequate bony jaw growth, and reduce the rate and degree of interproximal wear between adjacent teeth. This is not an issue within the Libben sample, which shows higher than modern interproximal wear and normal occlusion consistent with their non-modern diet.

The second reason Libben was selected is because it is one of the largest skeletal populations. Of 1,327 skeletons from the site, 122 subadults preserved measurable mandibles. With at least 10 individuals in each eruption stage, the sample used here is large enough to provide a reliable estimation of variation within stages (Table 3.2). Finally, Libben alone was selected as a modern referent in order to minimize any effects of population variation in growth (Eveleth and Tanner, 1988; Walker et al., 2006a. For example, Holmes and Ruff (2010) were able to detect expected differences in mandibular growth between two prehistoric North American human populations (n=42 and n=60) with contrasting diets. Intraspecific variation could be unpredictable and random, could artificially inflate sample variance, and could thereby increase the likelihood of failing to reject the null hypothesis. Libben was therefore selected because it provides a sufficiently large sample and represents a single, continuous occupation of hunter-gatherers (Lovejoy et al., 1977).

Dental Eruption Stages

One reason the mandible makes an ideal element for the study of growth is that the presence of developing and erupting teeth (and the bony alveolus containing them) reflect an individual's age and maturity. Calendar ages can be difficult to estimate because of variation both within and between populations in the ages at which children's teeth form and erupt (Dalhberg and Menegaz-Bock, 1958; Wolpoff, 1979; Hägg and Taranger, 1981; Eveleth and Tanner, 1988; Tompkins, 1996; Liversidge, 2003). Although it is possible to assign chronological ages to each mandible studied, there is not a clear answer to which (if any) population standard is best to use.

In contrast to the great variation in the ages at which humans erupt various teeth (Wolpoff, 1979; Liversidge, 2003), the sequence in which teeth emerge into the mouth is fairly consistent. For instance, the first permanent molars and incisors always erupt before the second molars and incisors. Important for life history, the eruption of certain teeth roughly coincides with other developmental milestones (Hägg and Taranger, 1981; Kraemer et al., 1982; Smith, 1991, 1993; Bogin, 1999; Zihlman et al., 2007; Hochberg, 2012), allowing the growth period to be divided into developmental stages. For instance, Humphrey (2010) notes that the full occlusion of the deciduous dentition roughly coincides with the end of the nursing period in many human populations. The first permanent molar erupts around the age that weaning is complete in most mammals, to facilitate a weanling's adoption of an adult diet (Smith 1991); humans are a glaring exception to this. Some studies have also found the eruption of the second molar corresponds (though fairly roughly) with the onset of behavioral, skeletal and hormonal correlates of puberty in humans (Hägg and Taranger, 1982) and chimpanzees (Kraemer et al., 1982, Zihlman et al., 2007).

This analysis therefore places individuals into age groups based on stages of dental eruption (e.g. Krogman, 1931a-c), as these are directly comparable across populations and hominid taxa. Eruption is a continuous process wherein a tooth emerges out of its bony crypt (alveolar eruption) and then past the gums (gingival eruption), and comes into contact with its maxillary counterpart (occlusal eruption), although the process continues even after teeth are fully occluded (Wolpoff, 1979). Skeletal remains can only document occlusal and alveolar eruption, and so teeth are treated here as either unerupted (still within a crypt), erupted through the alveolus but not in occlusion, or fully occlusally erupted. With these eruption criteria, eruption stages are defined so as to maximize the number of age groups and minimize the potential overlap between groups. Samples are divided into as many stages as possible such that specimens in one stage can be assumed likely to be less mature, chronologically or developmentally, than those in the next group.

The analysis concerns only growth prior to adulthood as defined by full occlusion of the last molar (M₃). Lockwood and colleagues (2007) have suggested that *A. robustus* males continued growing through adulthood while females did not, resulting in higher levels of sexual dimorphism than in humans. Adults are thus excluded from this analysis to minimize the possible effects age- and sex-related variation.

Eruption Stages

Individuals are assigned to the following groups defined by dental eruption:

Eruption stage	A. robustus	Humans
1: Deciduous m ₂ erupting/occluded	3	36
2: M ₁ /I ₁ erupting but not in	2	10

occlusion		
3: M ₁ occluded but M ₂ unerupted/unoccluded	1	32
4: M ₂ occluded but M ₃ unerupted	6	33
5: M ₃ erupting but not occluded	1	11
total	13	122

Table 3.2. Sample sizes for each eruption stage.

Eruption Stage 1: The youngest specimens in the fossil sample are either about to complete (SK 438) or have just completed (SK 64 and 3978) the eruption and occlusion of their deciduous dentition, with only light attrition of preserved teeth. The dm_2 of SK 438 is not quite to full occlusion, and there is no visible wear on its cusp tips, whereas these tips are only slightly worn and rounded in SK 64 and 3978. These three fossils show identical development of unerupted permanent teeth where they can be compared (P₄ and M₁; Conroy and Vannier 1991a), and none shows signs of exfoliation of the di₁ or of I₁/M₁ eruption. SK 438 is thus probably slightly younger than the other two fossils which are of comparable age.

A limited human sample whose dm_2 was in a similar stage of eruption as SK 438 (n = 5) precluded division of this subsample into two groups; this would be an interesting comparison for future work with a larger sample. Eruption stage 1 is defined, then, as a deciduous dentition, either fully erupted or with dm_2 nearing full occlusion, and prior to the exfoliation of the di₁ or eruption of the I₁/M₁.

<u>Eruption Stage 2</u>: The next youngest *A. robustus* are SK 61 and 62, which have completely erupted deciduous dentitions and are in the earliest eruption stages of M_1 and I_1 , respectively. Each of these jaws is noticeably larger and more robust (i.e. thicker corpora) than both SK 64 and 3978, and their teeth much more worn. **Eruption stage 2 is** thus defined by an I_1 and/or M_1 in the process of erupting, but not yet in full occlusion.

Eruption Stage 3: A single A. *robustus*, SK 63, is in this category as its M_1 is freshly occluded and its incisors in the process of erupting. Eruption stage 3 is defined by full occlusion of I_{1-2} and/or M_1 , but prior to the exfoliation of the deciduous canines or molars or eruption of any other permanent teeth.

<u>Eruption Stage 4</u>: The next oldest group of fossils includes six individuals with various stages of premolar and M_2 eruption and occlusion, and of different levels of preservation. It is problematic to create distinct eruption stages based on $P_{3.4}/M_2$, because of inter- and intraspecific variation in their eruption sequence. Modern humans tend to erupt the premolars before M_2 , although there is variation in this sequence (Garn and Koski, 1957; Smith, 1994; Liversidge, 2003). Australopithecines and great apes, on the other hand, tend to erupt M_2 before the premolars (Schultz, 1960; Smith, 1994). While this latter pattern characterizes most of these six *A. robustus*, the fossil sample suggests $P_{3.4}/M_2$ sequence polymorphism was present in this species, since the M_2 of SK 55b is freshly occluded but the dm₂ are retained. Given this polymorphism, and the temporal proximity of the eruption of these teeth, it is impossible to create distinct dental categories based on the eruption of these teeth. Thus, **eruption stage 4 is defined by an M₁ in occlusion, premolar(s) and/or M₂ erupting or occluded, and M₃ showing no sign of beginning eruption.**

Eruption Stage 5: This stage includes a single fossil, SKW 5, whose adult teeth are all occluded save for the M₃, which had probably just broken through the gum line but are unworn, indicating they had not come into full occlusion (cf. Grine and Daegling, 1993). Thus, **the fifth and final eruption stage is defined by a completely occluded permanent dentition up to the M₂, with M₃ in the process of erupting without having reached occlusion.**

In sum, the *A. robustus* and human samples are each divided into five comparable subgroups based on dental eruption. Whereas there is a large amount of intra- and interpopulation variation in the chronological ages at which these stages are attained, dental eruption is more closely related to the growth process and life history. Thus, the stages are sequential, but do not necessarily the same in time span. These stages, and not chronological ages, are directly comparable between samples, and so provide the basis for comparing ontogenetic variation in the mandibles of each species.

Variables for Analysis

There are a number of ways to measure size and shape, and to assess growth in these measures statistically. However, the nature of fossil samples places severe limitations on analytical options. The present study examines 29 linear measurements that describe mandibular size and shape (<u>Table 3.3</u>). The 29 traits can be broadly divided into five categories: [1] corpus (including symphysis) breadth, [2] corpus (including symphysis) height, [3] anteroposterior mandibular length, [4] anterior mandibular breadth, and [5] ramus height and anteroposterior length. Many of these measurements will

covary and be slightly redundant – for instance corpus height at P_4 compared with the height between P_{3-4} – but these both are necessary to maximize the number of comparisons that can be made between fragmentary fossil individuals; redundancy is minimized by the unfortunate fact that the one thing that fossil samples have a lot of is missing data. Where possible, only an individual's left side was measured, although data from the right side were used if the data from left side were absent. Measurements were taken to the nearest 0.10 mm with digital sliding calipers, made of plastic so as not to damage precious fossils and delicate subadult bones.

To assess measurement error, each of the 29 measurements was taken three times on each fossil, and on a subset (n=48) of the humans encompassing all eruption stages. Most measurements differed by 0.0-0.2 mm between replications, and the difference between the maximum and minimum triplicates rarely (<1% of the time) exceeded 1.0 mm. For the fossils and this subset of individuals, the average value of the replications was used in the analysis.

In addition to these 29 traits, a pairwise size metric, the geometric mean of the measurements shared by pairs of specimens, is also examined (Jungers et al., 1995). This is calculated as the n^{th} root of the product of n traits. This metric serves a two main purposes. First, it is of interest to summarize the size of an individual with a single number, as size is an important life history variable. Second, and related, not all specimens preserve the same traits, so defining size in terms of traits shared by *pairs* of specimens maximizes the ontogenetic information that can be taken from the sample.

While the analysis of linear measurements has been eclipsed in recent years by descriptions of shape based on configurations of three-dimensional landmark coordinates

(geometric morphometrics, or GM), linear measurements are more appropriate for the present study. GM has been used to compare mandibular growth between groups (e.g. Franchi et al., 2001, 2007; Coquerelle et al. 2010b, 2011), but will not be used in this analysis for two main reasons. First, GM requires that shapes (defined by 2- or 3-dimensional landmark coordinates) be superimposed on one another before they can be compared in a multivariate "shape space," but the choice of superimposition technique is fairly arbitrary and affects the results of an analysis (Lele and Richtsmeier, 2001). The principle superimposition method is generalized Procrustes analysis (Rohlf, 1990), which centers a set of shapes on their centroids (the point in space closest to all other landmarks), after removing the effects of size and shape. This procedure thereby distributes shape variation equally across landmarks.

This superimposition is biologically unrealistic for modeling facial growth, especially in an ontogenetic series – the mandible does not simply enlarge equally in all directions throughout the growth process (Enlow and Hans, 1996). The mandible initially has space only for the developing deciduous dentition, and so space for the additional and replacement teeth is added by differential growth in different anatomical regions. Histological and X-ray based implant studies of mandibular growth in humans (Enlow and Harris, 1964; Johnson et al., 1972; Enlow and Hans, 1996) have shown that the growing mandibular corpus accommodates new teeth distal to the dm₂/P₄ by adding bone at the corpus-ramus junction. During growth the corpus itself "drifts" anteriorly with the oral tissues surrounding it, and so bone is deposited at its posterior extent to maintain functional contact with the ramus. Thus, an ontogenetic series of mandibles will

necessarily contain missing data, as subadults will lack data taken at the positions of teeth that have not yet erupted – because these positions literally do not exist yet.

Second and more importantly, GM and multivariate statistics are not employed here because they are not appropriate for small samples with missing data. As mentioned above, the traditional superimposition method analyzes shapes in a unique multivariate "shape space" defined by complete configurations of landmarks (Rohlf, 1990). Comparing these superimposed shapes in samples with missing data is impossible because different landmark configurations would occupy different shape spaces. The shape-space constraint characterizes traditional multivariate statistics beyond landmarkbased GM.

The necessity to have complete specimens severely limits the scope of traditional methods, causing many researchers to restrict their samples to include only the few, more complete fossils. This strategy does not take advantage of the potentially useful information found in the more fragmentary specimens. Moreover, it often limits questions of growth to examining only the starting and end-points of growth, blind to patterns of growth between these time points. In addition, the variance of traits in small fossil samples typically violates the assumptions of parametric and multivariate statistical tests commonly used in studies of growth (Ackermann, 2005; Ahern et al., 2005).

In the present study, the variables of interest are thus linear measurements that can be taken on very fragmentary fossils. Growth is measured here as the proportional size change between eruption stages, that is the ratio of the older divided by the younger individual. This strategy is similar to that used in Euclidean Distance Matrix Analysis (EDMA; Richtsmeier and Lele, 1993; Lele and Richtsmeier, 2001; Richtsmeier et al.,

2002). A ratio as opposed to an absolute size difference between stages is used, because the *A. robustus* mandible attains an absolutely larger adult size than any human, and growth rates at a given time must be somehow proportional to size at the time (Huxley, 1932; Tanner, 1951). For instance, if the absolutely smaller human mandible undergoes an adolescent growth spurt during which it achieves a large proportion of its overall size, but the absolute size increase is the same in the already-larger *A. robustus*, the significance of this human spurt would be lost if growth were measured as an absolute and not relative change.

Comparisons across age groups raise special issues of homology. Mandibles with deciduous teeth contain crypts with developing crowns and roots, whereas in older individuals these teeth may be erupted and the crypts gone. In this study, tooth positions occupied by deciduous and permanent counterparts are considered homologous (e.g. corpus height at dm_2 is comparable to height at P_4 ; Table 3.3). The growing mandibular corpus elongates to accommodate additional teeth by deposition posteriorly at the lingual tuberosity, which causes tooth positions anterior thereto to migrate anteriorly (Enlow and Hans, 1996). Thus, even though teeth in these positions may be different at different eruption stages, the actual location of these teeth is the same. Similarly, measurements are taken on younger specimens at the positions of unerupted permanent molars where these can be determined.

In summary, data for this study include 29 linear measurements describing mandibular size and shape, plus a size metric that quantifies the size of individuals based on only those traits shared between pairs of specimens. These variables were selected because they allow fragmentary individuals to be examined and compared whereas

multivariate GM cannot handle so much missing data as in the present ontogenetic samples. Growth in these variables is measured as the ratio between older and younger individuals as determined by dental eruption.

Analysis of growth in <u>A. robustus</u>

The pattern of ontogenetic variation displayed by the fossils must be situated within a statistical framework – just how do mandibular size and shape change in ontogeny, and just how likely or unlikely is it to find the *A. robustus* pattern of variation within a similarly-sized and -preserved sample of modern humans? This section describes difficulties inherent in analyzing growth in fossil samples, and presents a new test using resampling (viz. exact randomization; Sokal and Rohlf, 1995; Manly, 2007) to compare the small ontogenetic series of *A. robustus* mandibles with the larger human sample.

The nature of fossil samples requires analytical methods beyond the traditional (parametric) growth curve – multivariate or univariate (but see Martín-González et al. 2012 for a recent resampling-based treatment). If fossils samples were complete and large enough, one way to make comparisons would be to create a growth curve by plotting trait values against age. From there, an exponential mathematical model approximate age-related changes could be fit to these data (Laird, 1967; Jolicoeur et al., 1991), and the null hypothesis that two groups follow the same growth trajectory or curve could then be tested by comparing each group's residuals from the modeled line (e.g. Holmes and Ruff 2011). When examining growth of many variables at once, multivariate regression or angles between principal components can be used to test for differences between groups (e.g. Cobb and O'Higgins 2004).

Trait ID	Mandibular Measurement		
Corpus breadth: taken perpendicular to the base of the corpus			
X223	Breadth at P3		
X225	Breadth at septum between P3-4		
X227	Breadth at P4		
X229	Breadth at septum between P4-M1		
X231	Breadth at M1		
X233	Breadth at septum between M1-M2		
X235	Breadth at M2		
X709	Breadth at symphysis, parallel to its long axis		
Corpus height: taken perpendicular to the base of the corpus*			
X301	Height at mental foramen		
X313	Height at septum between P3-4		
X315	Height at P4		
X317	Height at septum between P4-M1		
X319	Height at M1		
X321	Height at septum between M1-2		
X323	Height at M2		
X331	Height-mental foramen to base		
X333*	Distance from mental foramen to the nearest point on alveolar margin		
X355	Height from alveolar margin perpendicular to basal ramus-corpus junction		
X711	Height at symphysis		
	Anteroposterior mandibular lengths		
X401	Distal P4 to infradentale		
X403	Posterior condyle to distal-lingual I2		
X404	Distance from the center of the mental foramen to the posterior edge of the lingual tuberosity		
X415	Distance from the anterior margin of the mandibular foramen to distal- lingual P4		
	Anterior mandibular breadths		
X501	Bimental breadth		
X502	Bi-canine breadth (external)		
X510	Bi-canine breadth (internal)		
X511	Bi-P ₃ breadth, internal		
Ramus height and length			
X603	Posterior ramus margin on the alveolar plane to septum between P_{3-4}		
X607	Perpendicular distance from the mandibular foramen to the inferior ramus margin		
Table 2.2 I	ist and description of measurements. Measurements involving the normanent D		

Table 3.3. List and description of measurements. Measurements involving the permanent P_{3-4} correspond to the deciduous teeth they replace, dm₁₋₂, respectively, in juveniles.

But such a strategy cannot be done with the current samples. As described above, multivariate methods cannot handle missing data characteristic of fossil samples. And because not all individuals preserve all traits, not all traits can be studied throughout the entire growth period in the same way. For instance, a few corpus dimensions (e.g. height at P₄) can be measured throughout the entire *A. robustus* ontogenetic series, but many other variables are unevenly and sporadically distributed across eruption stages. This means that no single growth curve (uni- or multivariate) will describe growth in all variables. If some traits are present in only two stages this would only imply linear growth, and it would be inappropriate to model this with mathematical growth curve with as many parameters as the models used for traits sampling more of the growth period. Given these limitations of fossil samples, novel techniques for the study of growth are presented here.

Resampling or randomization methods are computationally intensive statistical techniques that provide flexible and rather assumption-free means of testing hypotheses about growth in fossil samples (Manly, 2007; Mattfeldt, 2011). Resampling works by randomly drawing individuals from a large comparative sample to create a subsample that matches the fossil sample in size and composition (i.e. preservation), and then computing a test statistic that describes the randomized sample. This random sampling is then repeated a large number of times (e.g. >10,000) to generate an empirical distribution of test statistics, providing a basis for understanding the likelihood of observing the fossil pattern in the comparative sample. The present study resamples an excessive number of times to obtain all possible combinations of specimens, and discards redundant observations, making this an exact randomization test (Sokal and Rohlf, 1995). This has

an advantage over parametric statistics that make assumptions about sample variation (Potvin and Roff, 1993) and rely on theoretical distributions of test statistics. Moreover, randomization *alone* allows inclusion of fossil fragments that are generally viewed as useless for more traditional statistical analyses, making the most of what little biological variation is preserved in fossil datasets.

Randomization/resampling thus makes it possible to test a null hypothesis that two patterns of variation are identical. To be sure, a failure to reject the null hypothesis may not indicate identity in growth, but rather could reflect the limitations of the fossil sample itself. Rejection of the hypothesis would also need to be treated with caution because of the potential for Type I statistical error (rejection of a hypothesis when it is in fact true). Thus, the validity of the randomization method described below was tested by running the procedure on the human sample alone, the logic being that a sample compared with itself should show no difference on average. The results of the validation are presented in Chapter 4.

The Randomization Procedure

This section outlines the randomization procedure used to make statistical inferences about differences in patterns of growth between *A. robustus* and humans. All analyses are written in the statistical computing package *R* (R Development Core Team, 2011), and can be found in Appendix III.

Recall that 'growth' is defined here as the proportional size change between eruption stages, and that patterns of mandibular growth in *A. robustus* can only be inferred from 13 individuals. Some pairs of these fossils will preserve more traits than

other pairs, some may share no traits in common, and different pairs will preserve different sets of traits. To maximize the amount of growth-related information that can be gleaned from this sample, the analysis makes pairwise comparisons of proportional size change between *every possible combination* of fossil pairs. This set of all possible proportional changes in *A. robustus* is then sought in the set of all possible pairwise changes that can be resampled in the cross-sectional ontogenetic sample of modern humans from Libben. This strategy situates the fossil pattern of ontogenetic variation within the context of human variation for hypothesis testing.

Proportional changes in size are examined for a pairwise size metric, in addition to the 29 individual traits. This metric allows an individual's different measurements to be summarized with a single number. As this analysis relies on pairwise comparisons, and not all individuals preserve all the same traits, this size metric is calculated as the geometric mean (Darroch and Mosimann, 1985) of all the measurements that a set of individuals shares in common (cf. Lockwood 1999; Gordon et al., 2008). This means that an individual may be described by many size metrics depending on the pairwise comparison, and that 'size' is not necessarily homologous between comparisons. However, as detailed in the algorithm description below, the size metric need only be homologous within pairwise comparisons.

The resampling algorithm is performed for each trait including the pairwise size metric. This algorithm is schematized in Figure 3.1, and works as follows:

 Randomly select two *A. robustus* mandibles in different eruption stages, and then randomly select two human mandibles so that the pair matches the fossils' stages. Record the following metadata (Table 3.4): identities of which fossils and which

humans are the older or younger of the pair, and the individuals' eruption stages and the number of traits they preserve in common (k).

2. Calculate a test statistic ζ, describing the species difference in the amount of relative change in size between the eruption stages (*nota bene*: ζ is not the within-species proportional changes themselves). Recall that when calculating ζ for the pairwise size metric, this must be based only on the set of traits all four resampled individuals share. If k>0, the ratio is calculated first by taking the ratio of the older to younger individual in each species. Then take the difference, ζ, between the *A. robustus* ratio minus the human ratio:

$$\zeta = [(\mathbf{R}_{\text{OLDER}} / \mathbf{R}_{\text{YOUNGER}}) - (\mathbf{H}_{\text{OLDER}} / \mathbf{H}_{\text{YOUNGER}})].$$

As the absolute difference between two ratios, $\zeta=0$ means no difference in proportional size change, while positive values indicate that *A. robustus* underwent greater relative size change in this comparison, and negative values the opposite.

Because the analysis is based on cross-sectional data, it is possible that in a randomly selected pair, the older individual may be smaller than the younger individual (Fig. 2.3, Appendix I), whereas growth implies an increase in size. To prevent this biologically unrealistic situation, if a ratio of older to younger is less than 1 (older is smaller), the ratio is set to 1, indicating no size change between eruption stages for the species in that comparison. The human sample is large and there can great overlap in size between individuals in different eruption stages, whereas there tends to be less overlap between *A. robustus* in different stages (Appendix I). This means that the human sample is more likely to give the impression of negative growth than the fossils, and this fix therefore obviates the overestimation of ζ in such cases.

3. Repeat a large number of times (e.g. 300,000) to ensure that every possible pairwise comparison is made. For each trait, if 0 is within 95% of the randomized ζ statistics, the null hypothesis cannot be rejected for the trait. That is, if 0 is outside the randomized 95% confidence interval, there are statistically sufficient grounds to reject the null hypothesis in favor of an alternate hypothesis of different patterns of size-change between species.





Resampled data (including metadata) are stored in a matrix with variables as columns and resampled iterations as rows (described in Table 3.4). Iterations that returned NA values (no traits shared between resampled individuals) are discarded. Because the identities of resampled individuals are recorded in each iteration, redundant comparisons consisting of the same two fossils and two humans are removed, making the test an exact randomization procedure. Although it is computationally inefficient to store sample metadata (identities of resampled individuals), this is an important step for ensuring that the program works as intended. Moreover, anyone who is interested in checking whether and how the procedure worked may do so by reviewing the metadata, as the results matrices (even retaining redundant comparisons) are available upon request.

Metadatum	Description
tr	Trait used to calculate size change difference (ζ), individual or the pairwise size metric. The trait is listed according to its number preceded by X (e.g. X223).

k	Number of traits shared by resampled individuals, used to calculate pairwise size metric
a1-2	Eruption stage of the younger (a1) and older (a2) individual of a resampled pair.
da	Difference in eruption stages between resampled pair, e.g. da=1 when the pair includes stages 1 and 2, da=3 when the pair includes stages 2-5, etc.
f1-2	Identities of each fossil, f1 corresponding to the younger and f2 to the older of the two.
h1-2	Identities of each human, h1 corresponding to the younger and h2 to the older of the two.

Table 3.4. Sample metadata (aside from the ζ statistic) stored for each resampled comparison

Summary and Conclusions

Fossil samples present two major challenges to statistical analysis and inference, small sizes and copious missing data. Rather than ask whether *A. robustus* conforms to a specific growth model of growth or pattern of shape change, the present study asks the more basic question of whether two patterns of variation can be distinguished. A novel randomization procedure was presented to quantify and statistically test this question: is the proportional size difference between any pair of *A. robustus* mandibles likely to be seen in a randomly-selected pair of humans at comparable stages of dental eruption as the fossils?

The pattern of *A. robustus* ontogenetic variation is sought in an ontogenetic series of human mandibles from the archaeological population from Libben. This comparative sample was chosen because it is large (n=122 mandibles with at least 10 individuals in each of five dental eruption stages), homogeneous and preagricultural. Comparison was limited to a single human population because this study is concerned with the evolution of human growth and life history. This narrow focus moreover prevents any potential artificial inflation of variance in the human sample due to natural and unpredictable interpopulation variation.
A. robustus and humans were compared for patterns of size change between five dental eruption stages, spanning from having only an erupted deciduous dentition to just before the last permanent tooth (M₃) comes into occlusion. Eruption stages are directly comparable between species, and correlated with the growth process and other aspects of life history. The ontogenetic series ceases before adulthood, marked by the full occlusion of M₃, because of the great range of variation in the age at which M₃ comes into occlusion in humans, and because of the potentially confounding effects of high sexual dimorphism in adult *A. robustus*.

Fossil samples, small and missing lots of data, cannot be analyzed using current multivariate methods including landmark-based geometric morphometrics. The present study therefore examined 29 linear measurements and a pairwise size metric based on these variables. These data can allow all measurable fragments – not just the most complete specimens – to be included in a randomization analysis. The ζ test presented above resamples *pairs* of individuals, which uniquely allows it to analyze ontogenetic variation in small, fragmentary fossil samples. The null hypothesis predicts that resampled ζ statistics, which measure the difference in relative size change between pairs of *A. robustus* and humans, will equal zero (no difference) on average.

Failure to reject the null hypothesis could occur for two main reasons: first, the fossil sample may appear to fit a human pattern in light of an actual difference in growth, due to the small fossil sample size. Alternatively, failure to reject the hypothesis may indicate true similarity. Rejection of the hypothesis, however, would strongly indicate differences in patterns of growth between species. Either outcome would provide new

information for paleoanthropology, about the limitations of present fossil samples and methods, and about the paleobiology of *A. robustus*.

Chapter 4

Results of the ζ Test

"Postnatally, the mandible undergoes more variation in shape and greater increase in size than any other facial bone." (Scheuer and Black, 2000: 144)

This chapter presents the results of the ζ test, testing whether patterns of relative change in size can be distinguished in cross-sectional series of human and *A. robustus* mandibles. As this is a novel procedure, the first section (4.1) investigates the behavior of the ζ test. First I examine whether the pairwise size metric (the geometric mean of the set of measurements a pair of specimens shares) is an appropriate measure of size and can be used compare individuals. Next, I assess the behavior of the ζ test, by using it to compare the human sample with itself. This examines the likelihood of rejecting a null hypothesis when it is true. I also examine the potential effects of specimen preservation and the difference in eruption stages being compared on the outcome of the analysis.

Section 4.2 then presents the results of the ζ test comparing the amount of relative change in overall size, measured with the pairwise size metric, between humans and *A*. *robustus*. Relative change in size is first treated across all possible comparisons, and then focusing on comparisons of individuals in successive eruption stages (i.e. across ontogeny). I also investigate the relationship between the value of the ζ statistic and resampled metadata.

Section 4.3 presents the results of the ζ test for differences in size change of individual traits, presented in five anatomical categories (Table 3.3). As in sections 4.1-2,

 ζ statistics for each trait are first treated across all comparisons, followed by comparisons between successive eruption stages. The section ends with a qualitative description of the ontogenetic changes to the anterior corpus (and symphysis) of *A. robustus*, as the complex topography of the region could not be reliably measured with the present techniques.

<u>4.1 Validation of the ζ Test</u>

The ζ test is novel in that it analyzes patterns of change (i.e. age-related) by randomly sampling differently (dentally) aged pairs of individuals, which is necessary with fossil samples in which not all individuals preserve all the same traits. The ζ statistic measures whether the proportional size change between dental eruptions is the same in both *A. robustus* and humans, and is expressed as the difference between the ratios: $\zeta =$ [($\mathbf{R}_{OLDER} / \mathbf{R}_{YOUNGER}$) – ($\mathbf{H}_{OLDER} / \mathbf{H}_{YOUNGER}$)], where R and H are an *A. robustus* and human mandible, respectively. Even though this analysis is not the first to use the geometric mean of several measurements to compare fragmentary samples (i.e. Lockwood, 1999; Gordon, et al. 2008), this has not been done in an ontogenetic context. The purpose of this section is therefore to establish the validity of both the pairwise size metric and the randomization methods employed by the ζ test, to show that it can be used to compare humans with *A. robustus*.

Measuring size

With the present samples, focusing on any single trait for the study of growth will require some fossils to be left out, and there are too fossils to prefer approach. A pairwise size metric, based on whatever trait or traits a pair (or pairs) of specimens preserve in common will maximize the number of comparisons that can be made between age groups. As such, size may be measured differently between different pairs of individuals. This could be problematic because it is not clear how such a size metric behaves analytically. A simple visual comparison human and *A. robustus* mandibles shows that *A. robustus* adults are larger than humans but this gross size difference is less marked in the youngest mandibles. The pairwise size metric ought to be able to detect this pattern.

To examine whether patterns of size difference can be detected using the pairwise size metric, the ζ test described in Chapter 3 is modified to extract all individual pairwise sizes that can be calculated within eruption stages. First, a fossil and human in the same stage are selected at random. If the pair shares at least two traits in common, each individual's size is calculated as the geometric mean of these measurements; pairs with fewer than two shared traits are discarded. There are only 360 possible pairwise fossil-human comparisons (e.g. Table 3.2), but the procedure resamples 5,000 pairs to ensure full coverage, and redundant observations are discarded.

Of the 360 possible pairwise comparisons, 300 shared at least two traits in common from which to measure size (Fig. 4.1). Of the 300 pairs, the *A. robustus* is larger than the human in all but 10 (3.33%), and all of these occur within eruption stage 1. These pairwise sizes can be plotted against eruption stage, similar to a growth curve (Fig 4.2). Overall size overlaps considerably in eruption stage 1, less so in stages 2-3, and finally *A. robustus* sizes are completely outside the upper limits of the human size ranges in eruption stages 4-5. By eruption stage 3 the average *A. robustus* size (calculated solely

from comparing SK 63 with humans) is larger than most humans, even those in stages 4-5 (Table 4.1).



Human size

Figure 4.1. Comparison of *A. robustus* and human pairwise size metrics (geometric mean of measurements shared in pairwise comparisons). Each point represents a pairwise comparison, with the *A. robustus* size on the y-axis and the human size on the x axis. Colors correspond to eruption stages: Green = stage 1, Purple = stage 2, Black = stage 3, Blue = stage 4, Yellow = stage 5. The dashed black line indicates identity in size (y=x).

The trajectory of the human sizes across eruption stages (green box and whiskers

in Fig. 4.2) is comparable to published growth curves for body mass and stature in humans (Eveleth and Tanner, 1988). There is a general increase in size up to eruption stage 3 (occlusion of the M_1), and a relatively large increase in size by eruption stage 5 (alveolar eruption of M_3) similar to the adolescent growth spurt. An important difference, however, is that average mandible size is relatively stagnant between stages 3-4, whereas stature and mass growth increase relatively slowly during this period in published growth curves. Note also that in this cross-sectional sample there is a very large range of variation in stage 4, and the median human size is slightly lower than in the preceding period.



Eruption stage

Figure 4.2. *A. robustus* (blue) and human (green) pairwise sizes across ontogeny. Individuals' size may be represented more than once within any stage. Boxes include the 50% quartiles, solid dark lines within boxes are medians, and whiskers extend to the maximum and minimum values.

	Stage 1	Stage 2	Stage 3	Stage 4	Stage 5
A. robustus	16.6 (1.36)	20.8 (1.39)	23.1 (1.35)	28.8 (1.86)	31.0 (1.65)
Humans	14.8 (2.14)	16.5 (1.95)	19.2 (2.03)	18.7 (2.54)	22.0 (2.31)

Table 4.1. Species' average pairwise size in each eruption stage: Mean (σ). Pairwise size is measured as the geometric mean of whatever traits a resampled *A. robustus*-human pair shares in common.

This is probably due only partly to sample size, stage 4 being the largest

subsample for humans and A. robustus. High variation is probably also due to both sexual

and ontogenetic variation. Eruption stage 4 is defined by the full occlusion of M₂, which

occurs in a second wave of dental eruptions that roughly coincides with the onset of puberty, prior to the adolescent growth spurt in height and weight (Hägg and Taranger, 1981). There is a great deal of variation in the timing, duration and rate of adolescent growth. The female spurt tends to be earlier, shorter and less intense (lower growth rates) than males', and so sex differences in size and shape become apparent usually around this time, even for the mandible (Coquerelle et al., 2011). Thus, I believe that size shows the most variation in eruption stage 4 because it includes both boys and girls at slightly different stages of skeletal or mandibular growth, although sex is unknown (not estimated) for most Libben subadults.

The distribution of human mandible sizes across eruption stages is thus similar to expectations from known human bodily growth patterns. Similarly, the distribution of *A*. *robustus* mandible sizes is consistent with visual comparisons of the actual fossils. It is therefore reasonable to measure an individual's size as the geometric mean of the traits it shares with another, which is at the heart of the ζ test.

Randomization methods

In any statistical test, one runs the risk of rejecting a true hypothesis (Type I error) and failing to reject a false hypothesis (Type II; Sokal and Rohlf, 1995). The risk of type II error cannot be assessed in the present case, as this is the very question being examined in the first place (whether *A. robustus* and humans can be distinguished). However, the risk of Type I error can be assessed by comparing the human sample with itself: ζ statistics calculated from within a single sample ought to equal 0 on average.

To assess the risk of Type I error, the randomization procedure described in Chapter 3 is modified to compare humans with humans for the overall size metric (geometric mean). As in the original procedure, two *A. robustus* in different eruption stages are randomly sampled. Next, four humans are resampled with replacement (as opposed to only two), two for each of the stages sampled. The four humans are stripped down to the traits shared between all four and the two fossils, to maintain similarity with fossil preservation. The ζ statistic is then computed based solely on the humans – is there any difference in relative size change between two randomly sampled pairs of humans at different eruption stages? Ideally, the test statistics should cluster around 0, as individuals within a single population should provide the same signal of size change. Humans were resampled 100,000 times with replacement to ensure all possible comparisons are obtained, and redundant comparisons were removed.

Period between	n unique	Median			
eruption stages	resampled	k	ζ (mean)	ζ 95% CI	P ζ _{≤0} (1-tailed)
All Possible Combinations	58,709	5	-0.0005	-0.398,0.394	0.530
1-2	6,383	5	-0.004	-0.264,0.264	0.526
2-3	2,007	6	-0.0005	-0.262,0.262	0.534
3-4	5,984	6	0.002	-0.307,0.298	0.535
4-5	6,751	8	0.002	-0.289,0.291	0.543

Table 4.2. ζ statistics for the overall size metric comparing the human sample with itself.

Note that in this validation analysis, even though *A*. robustus is not involved in the calculation of the ζ statistic, two fossils are nonetheless resampled to determine the number of traits used to calculated size (geometric mean). This is important because the human sample is generally better preserved than the fossils, meaning that randomly selected humans will usually have more traits in common than with fragmentary fossils. If the outcome of the test has anything to do with the number of traits shared, this step

will ensure fossil-like preservation, and is therefore more true to the testing conditions when using fossils.



Figure 4.3. ζ statistics for the overall size metric for all pairwise combinations of differently-aged humans. ζ =0 means no difference between resampled pairs.

Results indicate that the ζ test is unlikely to reject the null hypothesis (no difference between groups) when it is true. Of 100,000 iterations, the procedure returned 58,709 unique comparisons. The mean ζ =-0.0005, and the median is 0 (Table 4.2, Fig. 4.3). Moreover, although this distribution is symmetrical about a mean of 0, it is not normal, but rather appears exponential, heavily weighted toward the mean with long tails. This indicates that within a single sample, the ζ test is unlikely to indicate divergent patterns of growth. However, the great range of ζ statistics (and long distribution tails) reflects the ontogenetic variation inherent in cross-sectional samples.



Eruption Stage

Figure 4.4 ζ statistic for the overall size metric comparing successive dental eruption stages. The red dashed line indicates ζ =0. Boxes include the 50% quartiles, dots are outliers, and solid black lines in each box are medians.

The same pattern emerges when comparing ζ statistics across ontogeny, i.e. between subsequent eruption stages (Table 4.2). For each pairwise comparison between successive stages, the mean ζ is essentially 0, although there is a large range of values as is clear from Figure 4.4. This demonstrates that within a cross-sectional sample such as the one employed here, normal individual variation can create the appearance of many patterns or amounts of growth between dental eruptions. This highlights the importance of comparing the small fossil sample with a larger reference sample, rather than making inferences based on the fossils themselves. For instance, the variation in ζ statistics describing size change between eruption stages across ontogeny (Fig 4.4) shows that it is possible (though highly unlikely) for a small subsample of humans from Libben to appear to follow a different pattern of growth than the rest of the sample (cf. Fig. 3.1).

Table 4.3 shows the rank correlations (Kendall's tau) between the ζ statistic and metadata, da (the difference between a pair's eruption stages) and k (the number of traits used to calculate the ζ statistic), in each resampled comparison. ζ statistics calculated from the human sample are not correlated with either da or k. There is a small but statistically significant negative correlation between the difference between individuals' eruption stage and the number of traits they share. Each of these relationships and their influence on the ζ statistic are discussed in more detail below.

Variable	ζ	da	k
ζ	-	0.949	0.658
da	-0.0002	-	2.2E-16
k	-0.001	-0.146	-

Table 4.3. Relationship between resampled ζ and metadata. Kendall's tau rank correlations are below the diagonal and the statistical significance is above. **Bold** cells are statistically significant.

The Effects of Preservation, k

Part of what makes the ζ test for overall size unique is that by calculating 'size' as the geometric mean of the traits shared between resampled specimens, it is able to make sample-based inferences even though a large amount of data is missing (e.g. Gordon et al. 2008). But because ζ is calculated from different sets of measurements depending on the pairwise comparison, this means overall size is not necessarily homologous across all comparisons. It is therefore possible that the number (and types) of traits used to calculate ζ in a given comparison will have an influence on the value and distribution of ζ statistics.

Figure 4.5 shows the distribution of how many traits were used to calculate all ζ statistics. The distribution shows a clear bias toward a small number of traits being preserved in common between specimens. Across the nearly 59,000 comparisons, a

median of five traits are shared and used to calculate size for a given comparison. Although there were 29 traits examined in this analysis, the most that any resampled set of individuals (six: four humans plus two fossils) shared was 20 traits, and this was in only seven (0.01%) of all possible comparisons. While most sizes were calculated off of more than just one trait, nevertheless there are 10,887 comparisons (18.5%) for which k=1. Although the k=1 subset is sizeable, it does not exert too great an influence the results: when excluding this subset, the median number of traits increases from five to only six traits; the mean ζ value shifts from -0.0004 to -0.0003; and the median ζ is still 0.



Number of traits (k)

Figure 4.5. The number of traits shared between individuals in the ζ test validation. These k traits were used to calculate the 58,709 ζ statistics. The red dashed line is the sample median.

No matter how many traits are shared, the resampled ζ statistics cluster around ζ =0 (Fig. 4.6), hence the low rank correlation between ζ and k in Table 4.3. But while there is no relationship between k and ζ , Figure 4.6 suggests a relationship between k and the *variance* in ζ . This is because chance preservation of fossils means that two specimens are more likely not to preserve features than to share them, so that there is a



negative correlation between k and the number of individuals sharing k traits (tau=-0.937, $P_{\zeta_{\leq 0}}=0.137E^{-13}$).

Number of traits (k)

Figure 4.6. The influence of the number of traits (k) on the value ζ statistics in the test validation. The red dashed line indicates ζ =0. Boxes include the 50% quartiles, dots are outliers, and solid black lines are medians.

Because the variation in ζ decreases as k increases (Fig. 4.6), this suggests that

the *absolute value* of ζ depends on the number of traits used. In fact, the rank correlation

between k and the absolute value of ζ is tau=-0.091 (P $\zeta_{\leq 0}$ =2.2E⁻¹⁶). This significant,

negative correlation contrasts with the lack of correlation between (raw) ζ and k, but it

should be noted that the correlation is nevertheless still fairly low. The central tendency is

for ζ to be close or equal to zero, only the length of the distribution tails become

absolutely larger as fewer traits are sampled. Thus, the influence of trait number on the

magnitude of measured difference in size change seems to be very small in the present case.

In summary, preservation is capricious in skeletal samples, and so the potential influence of the number of traits (k) used to measure differences in size change was investigated. Most comparisons are based on a relatively small number of traits, and there is a weak negative correlation between the *absolute* magnitude of ζ and the number of traits used to calculate this value. On the whole, however, the ζ statistic averages 0 across different age intervals, regardless of the number of traits shared between individuals. Effects of preservation appear to have little influence on the analytical outcomes, at least when a single population is compared with itself. However, when comparing two populations that do differ in patterns of growth, it is theoretically possible for preservation to influence the results of the ζ test. For instance, if species are similar in the growth of some traits but not others, preservation bias may favor, say, those traits that grow similarly in each group. In this case, ζ values close or equal to 0 would be overrepresented. This is one of the main issues that led to the present study, and the storage of resampled metadata allows the influence of preservation to be assessed in such a situation.

The Effects of Difference in Eruption Stage, da



Figure 4.7. The number of traits (k) shared between specimens spanning different eruption stages. For instance da=1 includes comparisons based on eruption stages 1-2, 2-3, 3-4, and 4-5. Boxes include the 50% guartiles, dots are outliers, and solid black lines are medians.

The difference between eruption stages (da) used to calculate ζ could also

potentially influence the value of ζ . There is basically no correlation between ζ and da, as shown in Table 4.3. However, there is a modest rank correlation (tau=0.112, P_{$\zeta \le 0$}=2.2E⁻¹⁶) between the *absolute value* of ζ and da, as with k above. This modest relationship may be explained, at least in part, by the weak but negative and significant rank correlation between da and k (Fig 4.7). Recall that even though the average ζ =0 for all possible values of k, the *variance* in ζ decreases as k increases. The outliers to this relationship are the comparisons for which da=4, which measures size change between eruption stages 1 and 5 (i.e. across the whole of ontogeny). The relatively complete fossils SK 3978 and SKW 5 are in eruption stages 1 and 5, respectively, and the pair preserves 21 of the possible 29 traits in common. Thus, the difference between eruption stages, da, itself probably exerts little or no influence on the value of ζ . The modest negative correlation between da and the absolute value of ζ is most likely due to the (weak) relationship between ζ and k and between k and da.

Summary of the ζ test validation

The ζ test, examining potential species differences in patterns of overall mandibular size change, was validated by first demonstrating that the pairwise size metric adequately reflects both inter- and intraspecific variation in mandibular size (Figs. 4.1-2), suggesting it is an appropriate size measure for the analysis. The ζ test was then used to compare the human sample with itself, to test whether the method performs according to null predictions: ζ statistics should indicate no difference between randomly sampled pairs of individuals from a single species - or better, a temporally constrained population such as we have with Libben (Lovejoy et al., 1977). As predicted, ζ statistics in the validation cluster around 0 (no difference), although some specific comparisons would suggest substantial age-related differences. For the most part, the value of ζ is hardly influenced by either preservation (the number of traits, k, shared in common between resampled individuals), or the eruption stages (viz. their difference, da) of the individuals used in a given comparison. There is a tendency for older individuals to preserve better and therefore have data for more traits. As a result, the most notable relationships detected were between k and da, as well as between the *absolute value* of ζ and both these metadata, but these relationships were weak.

The ζ test was designed to analyze differences between groups in overall size change between various (dental) ages, in the presence of large amounts of missing data. The validation of the test presented here suggests that this procedure can be extended to the compare *A. robustus* and human samples.

4.2: Species differences in overall size change

This section presents the results of the ζ test of the null hypothesis of indistinguishable patterns of age-related variation in humans and *A. robustus*. The first part compares patterns of overall mandibular size change for the pairwise size metric. As in section 4.1, the results of the ζ test across all possible comparisons are first presented, followed by results from comparisons of individuals in successive eruption stages (i.e. across ontogeny). The section ends by examining the influence of resampled metadata (number of traits shared, difference between a pair's eruption stages) on the value of the ζ statistic.

<u>Results of the ζ test comparing change in overall size</u>

All Possible Comparisons

A total of 29,148 unique human-*A*. *robustus* comparisons can be made with the present sample sizes and preservation (Table 4.4, Fig. 4.8). Mean ζ =0.310 (median = 0.299) for these comparisons, indicating that the amount of relative size change between eruption stages is greater in *A*. *robustus* than it is in humans on average. This difference is not statistically significant at the traditional p≤0.05 level, meaning the hypothesis of indistinguishable size change is not rejected. Nevertheless, the likelihood of seeing such a

large size difference in a human sample is still quite low ($P_{\zeta \le 0}=0.124$), and it will be

shown below that species differences in size change are not the equal between all

Period between	n unique	Median		Median		Probability
eruption stages	resampled	k	Mean ζ	ζ	ζ 95% Cl	ζ≤0 (1-tailed)
All possible						
comparisons	29,148	5	0.310	0.299	-0.178,0.908	0.124
1.2	2 120	6	0 1/17	0 155	0.081.0.305	0.005
1-2	2,120	0	0.147	0.133	-0.001,0.303	0.095
2-3	625	9	-0.051	-0.035	-0.261,0.050	0.635
3-4	4,091	7	0.317	0.314	0.020,0.574	0.019
4-5	1,985	8	-0.057	-0.021	-0.380,0.141	0.768
1_3	2 4 3 2	7	0 127	0 138	-0 146 0 342	0 162
1-0	2,402	1	0.127	0.150		0.102
1-4	13,139	4	0.457	0.459	-0.090,1.11	0.042
1-5	1,156	9	0.452	0.461	0.153,0.712	0.002
2-4	3,031	6	0.202	0.226	-0.240,0.520	0.151
2-5	220	13	0.159	0.166	-0.019,0.312	0.045
3-5	349	13	0.267	0.269	0.047,0.451	0.017

eruption stages.

Table 4.4. Species difference in relative change for the pairwise size metric (geometric mean). **Bold** values indicate that less under 5% the exact randomized distribution is less than or equal to $0 \ (P\zeta_{\leq 0})$.

Results corroborate visual comparison of infant and adult *A. robustus* mandibles, which shows that this species clearly underwent greater *absolute* size change than humans during postnatal ontogeny (Table 4.1, Fig. 4.2). The greater absolute and relative size change undergone by the *A. robustus* mandible is exemplified in ζ statistics calculated from individuals in eruption stage 1 to stages 4 and 5 (i.e. periods 1-4 and 1-5 respectively; Table 4.4). These comparisons have the highest mean ζ values, both above 0.450 versus the overall average of 0.310 across all possible comparisons. The ζ distributions for these two periods (both separately and combined) are significantly greater than 0 at p<0.05. Period 1-5 has the lowest probability of ζ being less than or equal to 0 (i.e. greater relative size change in humans), and is one of only three dental periods in which 0 is completely outside the 95% confidence interval of the ζ distribution. In contrast, there is negligible difference between species in relative size change between eruption stages 4-5 (discussed below).

This and the great concordance between the mean ζ for periods 1-4 and 1-5 indicate that the bulk of mandibular growth (as implied by the present sample) in *A*. *robustus* takes place by the time the second molar has come into occlusion, a dental milestone that roughly corresponds with the onset of behavioral and hormonal cues of adolescence (Hochberg, 2012).



Figure 4.8. ζ statistics for species difference in overall size change between all eruption stages. The red dashed line indicates ζ =0 (no difference between species), and the three bars in the rug indicate the 95% confidence interval (-0.178,0.907) and median (0.299) of the (exact) randomized distribution.

Eruption stages 1 and 4 have the largest sample sizes for both species (Table 3.2),

and as such these allow the largest number of unique ζ statistics to be calculated (45% of

all 22,060 comparisons). Not surprisingly, this period contains both the largest and the smallest values of D, and shows the largest variance in D. For instance, the upper 95% confidence limit for this period is 1.11, whereas all other periods' upper limit is less than 1 (Table 4.4). A good deal (31%) of the ζ statistics for period 1-4 were calculated from only k=1 trait, which somewhat misses the point of the ζ test measuring size as a summary of several traits. If the resampled ζ s comparing period 1-4 for which k=1 are omitted, the mean ζ drops only mildly (from 0.457 to 0.436), but the standard deviation decreases noticeably from 0.278 to 0.174. Importantly, 0 is no longer within 95% of the exact randomized distribution for which k>1 (n=9,118). In summary, even though there is great variation in ζ statistics for period 1-4, the period's better-preserved individuals indicate that *A. robustus* undergoes much larger amount of change in mandibular size than humans by eruption stage 4 (full occlusion of M₂).

In contrast with the large period 1-4 distribution, there are only 1,156 ζ statistics for period 1-5. Nevertheless, this period contains two relatively well-preserved fossils, SK 3978 in stage 1 and SKW 5 in stage 5, who preserve 21 traits in common. Period 1-5 thus significantly represents both the youngest and oldest age groups, and two of the most complete fossils in the analysis. Because of this pair, the median k=9 traits for this period which is high in comparison with the rest of the resampled statistics (overall median k=5). The n=388 comparisons including this fossil pair have a fairly high mean ζ =0.368±0.122. Because more traits tend to be used to calculate ζ in period 1-5, which samples the beginning and end points of this age series, this period provides a good estimate of the difference in overall size change that occurs in the mandible of each species. In summary, the distribution of ζ statistics from all possible pairwise comparisons of individuals reveals a general tendency for *A. robustus* to undergo greater size change between eruption stages than modern humans, reflected in a clear bias toward positive values of ζ . In all but two periods between eruption stages (2-3 and 4-5) the mean ζ is greater than 0, with just over 12% of all ζ statistics indicating greater relative size change in humans compared with *A. robustus*. The pervasive tendency for greater *relative* size increase in *A. robustus* gets compounded across ontogeny, so that by the end of the subadult period, the *A. robustus* mandible has undergone significantly greater size increase than a human on average. Period 1-5 contains some of the best-preserved fossils in this analysis, and so overall size is generally based on more traits than for other dental eruption periods. Period 1-4 contains a relatively large number of fossils, and although these are comprised nearly entirely of corpora, they reveal individual variation within an eruption stage. Taken together, these data indicate that the *A. robustus* mandible increases in size substantially more than humans do prior to alveolar eruption of M₃.

Successive Eruption Stages

Of all ζ statistics, 8,821 were between successive eruption stages. This set of stage-successive statistics provides the most fine scale view of potential ontogenetic variation between species. This is significant from a life history standpoint because these dental eruption landmarks roughly coincide with other ontogenetic milestones.

There are two periods in which *A. robustus* and human mandibles show very different amounts of relative size change. First, between eruption stages 1-2, the mean ζ =0.147 (P $_{\zeta \leq 0}$ =0.095; Table 4.4). While this difference is not significant at the traditional

($P_{\zeta \le 0}=0.05$) level, it is nevertheless very unlikely for the amount of relative size change seen in *A. robustus* between eruption stages 1-2 to be sampled from a single human population. A more notable and significant difference between species occurs between eruption stages 3-4, where mean $\zeta=0.317$ ($P_{\zeta \le 0}=0.019$). In the other two successive periods (stages 2-3 and 4-5) the mean and median ζ are negative but not significantly different from 0 ($P_{\zeta \le 0}=0.635$ and 0.768, respectively).



Figure 4.9. ζ statistics for interspecific difference in relative size change between successive eruption stages. The red dashed line indicates ζ =0 (no difference between species). The blue boxes include all 8,821 unique resampled statistics between successive dental eruption stages. Overlain on these in green is the subset of this group (n=6,377) for which k≥5 traits (the sample median). Note that the k≥5 distributions overlap with the full-sample distribution, but with reduced variation (i.e. less deviation from ζ =0). Boxes include the 50% quartiles, dots are outliers, and thick blue/green lines are medians.

Figure 4.9 shows the distribution of ζ statistics between successive eruption stages.

This depicts the degree to which each species deviates from a common pattern of size

change across ontogeny. Reflecting *relative* change in size, the ζ test is able to compare

humans with A. robustus in light of the latter's increasingly larger sizes. This figure

therefore illustrates how disparate patterns of size change result in *A. robustus* achieving a much larger adult size than humans: both species' stage 1 mandibles overlap in overall size, but *A. robustus* generally undergoes a relatively greater change in size by eruption stage 2. The now absolutely larger *A. robustus* mandible then undergoes a relatively greater amount of size change between stages 3-4, accentuating the species difference in overall mandibular size. Even though ζ statistics for periods 2-3 and 4-5 tend to be negative on average, this is not enough for human mandibular size to 'catch up' with *A. robustus* overall size. The ζ test therefore indicates that compared with modern humans, *A. robustus* experienced two 'spurts' of greater change in overall size of the mandible.

Influences on ζ

Table 4.5 shows the Kendall's tau rank correlations between resampled ζ statistics and metadata (da and k) and their significance. All of these variables are significantly correlated with one another. This is quite unlike the correlation matrix for the ζ test validation (Table 4.3), in which correlations were generally low and insignificant. These relationships are examined in more detail below.

Variable	ζ	da	k
ζ	-	2.2E-16	2.2E-16
da	0.407	-	2.2E-16
k	-0.167	-0.196	-

Table 4.5. Relationship between resampled ζ and metadata. Kendall's tau rank correlations are below the diagonal and the significance is above. All correlations are significant.

The Effects of Preservation, k

The median number of traits shared between individuals across all comparisons is 5. As in the test validation, a good deal of these statistics (21%) are based on only k=1

trait. ζ statistics in the k=1 subset exceed the range of observed ζ statistics for all other values of k (Fig. 4.10). Excluding these, the median number of traits shared rises only to k=6, and the average ζ statistic decreases from 0.301 to 0.293.

There is a general, but by no means universal, tendency for ζ to decrease with increasing numbers of traits used to calculate size. A least squares regression of ζ against k yields the line $\zeta = -0.016*k + 0.406$, and although this low slope is significantly different from 0 (F=1,792 on 1 and 29,146 df, p=2.2E-16), only a small amount of the variation in ζ is explained by k (r²=0.058).



Figure 4.10: The effects of preservation on the value of ζ . The red dotted line indicates ζ =0, and the blue dashed line is the least squares regression line of ζ against k. Boxes include the 50% quartiles, dots are outliers, and thick blue/green lines are medians.

A clear exception to the tendency of decreasing D with increasing numbers of

traits occurs for the n=79 comparisons where k=18-20. This group has a mean ζ of 0.294,

close to the mean for all possible comparisons. Reference to the results matrix (not included here due to its large size, but available upon request) indicates that the fossil comparisons in this set include only SK 3978 (eruption stage 1), SK 62 (eruption stage 2) and SKW 5 (eruption stage 5). The n=44 comparisons involving SK 3978 and SKW 5 have a higher mean ζ (0.374) than the other comparisons in the subset (period 1-2 mean ζ =0.182; period 2-5 mean ζ =0.136). Even though ζ statistics for periods 1-5 and 2-5 have similar variation (σ_{1-5} =0.093, σ_{1-4} =0.082), the period 2-5 statistics are shifted to lower values of ζ compared to period 1-5. Thus, this subset of resampled statistics for which k=18-20 shows a similar pattern to the overall results (Table 4.4). The difference in mean ζ for periods 1-5 and 2-5 highlights both the greater amount of overall size change seen in *A. robustus* between eruption stages 1-2 (Fig. 4.10) and overall between stages 1-5.

Also deviant from the weak relationship between ζ and k are comparisons based on k=16-17 traits. This subset (n=302) has a mean ζ =0.040 and median ζ =0. The relatively low average ζ in this subsample is driven by the n=192 comparisons including SKX 4446 in eruption stage 4 and SKW 5 in stage 5: for the subset k=16-17, all but three of the 180 ζ statistics less than or equal to 0 are based on this pair, and the mean ζ for this pair alone is -0.066. The remaining ζ statistics in the subset k=16-17 are based on periods 1-2, 1-5 and 2-5, and have a mean ζ =0.226, echoing the findings in the previous paragraph. This subset thus underscores the key findings described above: the comparisons of SKX 4446 and SKW 5 indicate relative size increase between eruption stages 4-5 is hardly different between humans and *A. robustus*, with humans actually showing slightly more size change. Nearly all other comparisons in this subset reflect relatively greater size change in *A. robustus* during infancy and across all ontogeny. That these comparisons are based off a large number of traits bolsters the rest of the results, relying on larger resampled distributions but based on fewer traits.

The Effects of Difference in Eruption Stages, da

While the difference between resampled individuals' eruption stages was shown to have negligible effects on D in the test validation, there is a significant, positive rank correlation between D and da in comparisons of humans with *A. robustus* (Table 4.6). Inspection of Fig. 4.11 indicates this is not a strong or causal relationship. Rather, it is driven by the fact that comparisons between individuals relatively close in eruption stage age (da=1-2) generally have a lower D than those spanning a greater part of ontogeny (da=3-4). This is consistent with findings described above, namely that *A. robustus* tends to undergo greater size change between eruption stages, and this tendency compounds so that the *A. robustus* mandible achieves a much larger adult size than humans.



Figure 4.11. Distribution of ζ for each difference between eruption stages. The red dotted line indicates ζ =0. Boxes include the 50% quartiles, dots are outliers, and thick black lines are medians.

Summary of Species Difference in Overall Size Change

Results of the ζ test indicate that *A. robustus* undergoes more size change between eruption stages on average compared with humans. Most but not all periods between stages show positive D values. This is most notable in comparisons of the amount of size change between the beginning and end of the ontogenetic sequence (periods 1-4 and 1-5). The effects of the tendency for greater size change in *A. robustus* get compounded such that by the end of ontogeny, *A. robustus* has undergone a much greater change in overall size compared with humans. This size difference is accomplished by relatively greater size change in *A. robustus* during periods 1-2 (not significant) and 3-4 (significant), and negligible difference between periods 2-3 and 4-5. There are significant rank correlations between ζ and sample metadata da and k. The relationship between D and da is due to the fairly well preserved fossils in eruption stages 1-2 and 5. This relationship cannot fully explain the negative correlation between ζ and k, and is not easily explained due to the high output of the ζ test. Nevertheless, the inverse relationship between preservation (k) and the value of ζ explains only a small amount of the variation in ζ . Moreover, in spite of the relationship, there is a tendency for ζ to be greater than zero across all possible values of k, save for the subset in which k=16-17, which is biased toward period 4-5 when there is little size difference between *A*. *robustus* fossils (SKX 4446 and SKW 5).

4.3: Species Differences in Individual Traits

In addition to examining changes in overall size using the pairwise size metric, the ζ test was run for all 29 mandibular measurements on their own (Table 3.2), comparing every possible set of human and *A. robustus* pairs. This section summarizes these results, presented in five anatomical categories: [1] corpus (including symphysis) height, [2] corpus (including symphysis) breadth, [3] overall mandibular length, [4] overall mandibular breadth, and [5] ramus height and length. These findings are summarized in Tables 4.6-7, and in Figure 4.12. Results are also interpreted in the context of the raw data for each trait, which are presented graphically in the Appendix I.

The results matrix of all ζ statistics for each trait contains 117,200 rows (unique resampled comparisons), and nine columns for the output variables/metadata (Table 3.3). Table 4.6 summarizes these results. It is clear from the table that comparisons are unequally distributed across traits and age groups. For instance, corpus heights and

breadths are fairly well represented, whereas other types of measurements are not as well sampled. This disparity in trait representation is due to a relative overrepresentation of mandibular corpora, and the fact that younger individuals are necessarily missing many variables because they had not yet developed by the time the individual died.



Figure 4.12. The distributions of all possible interspecific ζ statistics for each individual trait. Trait ID numbers and descriptions can be found in Tables 3.2 and 4.6-7. Boxes are color-coded according to anatomical category: Blue=corpus breadths, Green=corpus heights, Red=anteroposterior mandible lengths, Yellow=overall mandibular breadths, White=ramus. Traits X709 and X711 (far right) are measures of symphysis breadth and height, respectively. Boxes include the 50% quartiles, whiskers span the entire range of values, and thick black lines are medians.

Corpus breadths (traits X223-235 and -709)

There are eight measures of corpus breadth, including at the mandibular symphysis and at seven positions along the length of the corpus from the P₃ to the M₂. As could be predicted from visual comparison of fossil and human mandibles, average ζ statistics for each trait in this group tend to be positive (i.e. greater relative jaw breadth growth in *A. robustus*), although not 'significantly' so, as 0 is within the 95% confidence interval for each trait. Nevertheless, the general probability of D being less than or equal to 0 is fairly low (p≤0.181) for all breadth measures except at the position of M₂. Fewer than 9% of the resampled ζ statistics for breadth at P₃, P₄ or M₁ are less than 0. When considering relative size change across all eruption stage comparisons, the ζ test fails to reject the null hypothesis of indistinguishable patterns of corpus breadth change; but this is certainly not the case when considering patterns of change only between successive eruption stages.

In *A. robustus*, the mandibular corpus widens at every tooth position throughout ontogeny, but usually with little or no increase between eruption stages 4-5 (Table 4.7 and Appendix I). As with the pairwise size metric, positive ζ values tend to cluster within periods 1-2 and 3-4. In period 1-2, not a single one of the n=930 ζ statistics for breadth at P₃ is less than 0.042: in other words, in this cross-sectional sample of humans, one will never sample as great an increase in breadth at P₃ as is seen in *A. robustus* in period 1-2. Also during this period, breadths at more anterior tooth positions have higher average ζ values, with very low probabilities of resampling $\zeta \leq 0$ (i.e. *A. robustus* showing significantly greater relative size change).

The sole measurement deviating from this pattern during period 1-2 is corpus breadth at the alveolar septum between P₄ (or dm₂) and M₁. This is due largely to comparisons involving SK 3978 (eruption stage 1), which has a larger breadth at this position than the other fossils in stage 1 (SK 64 and 438), and thereby giving the impression of smaller size increases. (Interestingly, SK 3978 shows no sign of P₄ crypt formation or germ calcification; Conroy and Vannier, 1991b) As a result, ζ statistics involving SK 3978 average -0.010 (σ =0.091), much lower than the distributions of ζ statistics involving SK 64 (mean= 0.051, σ =0.092) and SK 438 (mean=0.123, σ =0.092). Omitting ζ statistics based on SK 3978, the mean ζ increases to 0.092 (P_{ζ =0}=0.182).

Corpus breadths in period 3-4 also produce fairly high, positive ζ statistics. As in period 1-2, the species difference in size change is larger for more anterior tooth positions than for around the M₁ more posteriorly. Mean ζ for the anterior breadths in period 3-4

are high and more similar to one another than they are in period 1-2. Posteriorly, the species difference in corpus breadth increase at M_1 and between M_{1-2} is not nearly as substantial as it is anteriorly. As the corpus had already become much broader in *A*. *robustus* than humans by eruption stage 2, the (significantly) high ζ statistics for corpus breadths in period 3-4 means that the *A*. *robustus* corpus becomes even more relatively broad compared with humans.

 ζ statistics for corpus breadths in periods 2-3 and 4-5 are quite different from the other periods. Nearly all traits have *negative* mean ζ values, indicating that humans undergo greater relative increase in corpus breadth in these periods. These ζ distributions, however, all center fairly close to $\zeta = 0$. During period 2-3, the ζ statistics for the anterior three corpus breadths average just under 0, but posterior to these ζ for breadth at M₁ averages just above 0. Similarly, during period 4-5, ζ statistics for corpus breadth anterior to M₁ are always negative, but ζ statistics are positive at M₁ (nearly significantly) and M₂. Thus, the ζ test indicates that during periods 2-3 and 4-5, *A. robustus* and humans experience similar change in corpus breadth anterior to the M₁ but *A. robustus* undergoes (equivocally) greater size change posteriorly.

Trait ID	Trait	c	Mean	Minimum,	95% CI	P ζ≤0
			ζ	Maximum		
	All traits	117,200	0.256	-1.13, 1.83	-0.288,0.933	0.193
X223	Breadth at P ₃	4,509	0.362	-0.402,0.927	-0.081,0.927	0.081
X225	Breadth at P ₃₋₄	4,959	0.311	-0.488,0.793	-0.096,0.793	0.114
X227	Breadth at P ₄	11,282	0.277	-0.531,0.561	-0.069,0.561	0.082
X229	Breadth at P ₄ -M ₁	9,761	0.242	-0.370,0.520	-0.085,0.520	0.125
X231	Breadth at M ₁	8,541	0.228	-0.043,0.648	-0.073,0.601	0.067
X233	Breadth at M ₁ -M ₂	896	0.105	-0.382,0.203	-0.144,0.203	0.123
X235	Breadth at M ₂	677	0.102	-0.233,0.332	-0.123,0.331	0.440
X709	Breadth at symphysis	1,297	0.280	-0.354,0.990	-0.150,0.802	0.181
X301	Height at mental foramen	6,506	0.125	-0.607,0.775	-0.276,0.560	0.303
X313	Height at P ₃₋₄	7,690	0.201	-0.494,0.982	-0.259,0.761	0.252
X315	Height at P ₄	11,904	0.352	-0.681,1.33	-0.235,1.06	0.161
X317	Height at P ₄ -M ₁	10,525	0.269	-0.744,1.06	-0.291,0.780	0.205
X319	Height at M ₁	5,398	0.158	-0.760,0.892	-0.343,0.636	0.284
X321	Height at M ₁₋₂	2,724	0.534	-0.676,1.28	-0.404,1.28	0.331
X323	Height at M ₂	736	-0.159	-0.739,0.025	-0.497,0.025	0.861
X331	Height-mental foramen to base	10,019	0.156	-1.12,1.09	-0.532,0.882	0.330
X333	Height-mental foramen to alveolar margin	5,652	0.202	-1.13, 1.056	-0.494,0.945	0.332
X355	Posteromedial corpus height	4,556	0.295	-0.525,1.00	-0.250,0.888	0.170
X711	Height at symphysis	282	0.133	-0.290,0.428	-0.181,0.376	0.199
X401	Length from P_4 - M_1 to Infradentale	104	0.022	-0.134,0.055	-0.113,0.055	0.173
X403	Length from the condyle to I ₂ -C	14	0.211	0.070,0.347	0.084,0.332	0
X404	Length from the mental foramen to lingual	2,481	0.282	-0.251,0.767	-0.102,0.671	0.100
	tuberosity					
X415	Length from mandibular foramen to P_{3-4}	5,036	0.401	-1.00,1.83	-0.378,1.44	0.152

X501	Bimental breadth	327	0.244	-0.245,0.585	0.178,0.525	0.187
X502	Bi canine breadth (external)	84	0.112	-0.261,0.348	-0.185,0.238	0.119
X510	Bi canine breadth (internal)	124	0.259	-0.243,0.508	-0.115,0.507	0.065
X511	Bi-P3 breadth, internal	172	0.084	-0.357,0.236	-0.248,0.236	0.291
X603	Ramus posterior margin to P ₃₋₄	258	0.399	-0.196,1.06	-0.151, 1.00	0.127
X607	Ramus height from mylohyoid foramen to	686	0.396	-0.439,0.899	-0.124,0.899	0.071
	base					
	• • • • • • •			-		

Table 4.6. Summary of ζ statistics comparing humans and A. *robustus* for each trait, across all age comparisons.

		Peri	od 1-2	Perio	od 2-3	Per	iod 3-4	Peri	od 4-5
Trait ID	Trait	mean	P ζ≤0 (n)	mean	P ζ≤0 (n)	mean	P ζ≤0 (n)	mean	P ζ≤0 (n)
	All traits	0.135	0.199 (11,053)	-0.056	0.579 (3,950)	0.301	0.091 (15,712)	-0.045	0.719 (11,561)
X223	Breadth at P ₃	0.304	0 (030)	-0.021	0.368 (365)	0.261	0.011 (281)	-0.053	1 (191)
X225	Breadth at P ₃₋₄	0.247	0.007 (907)	-0.04	0.725 (411)	0.258	0.012 (396)	-0.052	1 (213)
X227	Breadth at P ₄	0.112	0.073 (1,675)	-0.022	0.353 (476)	0.231	0.023 (863)	-0.048	1 (489)
X229	Breadth at P ₄ -M ₁	0.062	0.275 (1,438)					-0.025	1 (767)
X231	Breadth at M_1	0.147	0.030 (436)	0.043	0.190 (210)	0.029	0.270 (1,342)	0.120	0.053 (756)
X233	Breadth at M ₁ -M ₂					0.105	0.123 (896)		
X235	Breadth at M ₂							0.101	0.440 (677)
X709	Breadth at symphysis	0.315	0.021 (291)					-0.052	1 (146)
X301	Height at mental foramen	0.061	0.324 (552)	-0.090	0.828 (303)	0.183	0.112 (850)	-0.081	0.858 (660)
X313	Height at P ₃₋₄	0.048	0.392 (564)	-0.070	0.651 (281)	0.248	0.080 (1,126)	-0.083	0.846 (729)
X315	Height at P ₄	0.095	0.317 (1,006)	-0.045	0.588 (362)	0.319	0.020 (1,372)	-0.131	0.962 (679)

X317	Height at P_4 - M_1	-0.070	0.629 (642)	-0.0008	0.411 (387)	0.333	0.026 (1,823)	-0.127	0.946 (905)
X319	Height at M_1			-0.051	0.519 (376)	0.251	0.105	-0.095	0.733 (1,004)
X321	Height at M ₁₋₂					0.971	0 (1,491)	-0.111	0.834 (1,081)
X323	Height at M ₂							-0.159	0.861 (736)
X331	Height-mental foramen to base	0.120	0.317 (961)	-0.182	0.837 (484)	0.274	0.119 (1,330)	0.028	0.489 (767)
X333	Height-mental foramen to alveolar margin	0.222	0.230 (513)					-0.141	1 (638)
X355	Posteromedial corpus height	0.181	0.101 (328)	-0.128	1 (134)	0.346	0.055 (488)	0.063	0.356 (407)
X711	Height at symphysis	0.127	0.171 (117)						
X401	Length from P_4 - M_1 to Infradentale	0.019	0.209 (43)						
X403	Length from the condyle to I ₂ -C								
X404	Length from the mental foramen to lingual tuberosity	0.126	0.162 (284)	-0.042	0.553 (132)	0.270	0.005 (219)	0.002	0.333 (198)
X415	Length from mandibular foramen to P_{3-4}					0.264	0.068 (1,225)	0.045	0.330 (512)
X501	Bimental breadth	0.361	0 (163)						
X502	Bi canine breadth (external)	0.121	0 (31)						
X510	Bi canine breadth (internal)	0.188	0.137 (51)						
X511	Bi-P3 breadth, internal	0.181	0 (73)						
X603	Ramus posterior margin to P ₃₋₄	0.251	0 (48)	-0.094	0.897 (29)				
X607	Ramus height from mylohyoid			0.276	0.083				
------	-----------------------------	--	-----------	-------	-------	--			
	foramen to base		0.270 (21		(216)				

Table 4.7. ζ statistics comparing humans and *A. robustus* for each trait between successive eruption stages. Empty cells mean no comparisons could be resampled for successive eruption stages. **Bold** values indicate that fewer than 5% the exact randomized distribution is less than or equal to 0 ($P\zeta_{\leq 0}$).

It is not clear how the position of the root of the ramus contributes to species differences in corpus breadth growth (Table 4.8). In dental stage 1, the root flanks the dm_2 in both species. By stage 2 the root migrates posteriorly in humans to flank the M_1 , and then the distal M_1 by stage 3. However, in *A. robustus* the root contributes to corpus breadth at the dm_2 up to stage 3. Importantly, ζ statistics are lowest (i.e. not significantly positive) around dm_2 in period 1-2, and the higher and more significant ζ s are by the dm_1 and symphysis. In addition, mean ζ is modestly (but significantly) positive for breadth at the (unerupted) M_1 at this time. The significant differences between species in breadth increase at M_1 and dm_1 during this time suggest that M_1 and P_3 crown development is at least partly responsible for the difference, and irrespective of the position of the ramus root.

Eruption	Auchustus	Humong		
stage	A. robustus	numans		
1	Mesial dm ₂	dm_2-M_1		
2	mid dm ₂	M ₁		
3	Distal dm ₂	Distal M ₁		
4	mid/distal M ₁	M ₁₋₂		
5	mid M ₁	M ₁₋₂		

 Table 4.8. Position of the ramus root across eruption stages in each species.

However, the different position of the ramus root may contribute to species differences in corpus breadth growth later in ontogeny. The only positive mean ζ for corpus breadth in period 2-3 occurs at the position of M₁. During this period, the root of the ramus migrates from the mid-M₁ to flank the more distal portion of the tooth in humans, whereas it continues to flank the dm₂ in *A. robustus*. It is tempting to conclude the greater breadth increase in *A. robustus* here is due to humans' loss of the ramus root

laterally, but it should be noted that mean ζ is neither very high nor significantly positive (mean $\zeta=0.043$, $P_{\zeta\leq0}=0.190$).

Summary of corpus breadth. Whereas the *A. robustus* mandible can be characterized by generally increasing breadth through subadult ontogeny, the human corpus tends to broaden only modestly, if at all, in this cross-sectional sample. The species differ most in the amount of size change between eruption stages 1-2 and 3-4. In addition, size change at the more anterior tooth positions (P_{3-4}) generally show greater species difference than posterior positions (M_{1-2}). As ζ measures differences in *relative* size change, the relatively greater breadth increase of *A. robustus* during period 3-4 is all the more remarkable since it compounds a species difference in breadth established earlier during ontogeny. As a result, the modest disparity in corpus breadth between species early on quickly gets amplified during ontogeny. Even though species differences in relative size change are small in toward the end of subadulthood (period 4-5), the posterior corpus at M_1 of *A. robustus* undergoes slightly more size increase than humans, due to the more anterior presence of the ramus on the corpus of in the former.

Corpus heights (traits 301-355 and 711)

There are 11 measurements of corpus height including at the symphysis and at 10 other positions along the length of the corpus. Corpus heights are very well represented in this fragmentary sample, with anywhere from 2,700-11,904 unique ζ statistics for all but two height measures. Plots of raw data hint that amounts of relative height increase are similar between species, although *A. robustus* heights are often shifted to higher values

(Appendix I). The ζ test (for all possible comparisons) indicates that although *A. robustus* tends to experience greater relative increase in height at most positions, it is not uncommon for humans to show at least as much change as *A. robustus* for all measures of corpus height (Table 4.6, Fig. 4.15).

Across all possible comparisons of size change in corpus height (Table 4.6), height at M₂ sticks out as having a highly negative mean ζ of -0.159. This trait is represented only in eruption stages 4-5, a time when humans undergo relatively greater size change in not only height at M₂ but also overall size and most other mandibular measurements. This result must be treated with caution since there is only one *A*. *robustus* in stage 5.

During eruption period 3-4, all height measures have positive mean ζ values. There is a very low chance (0-10%) of sampling *A. robustus*-like corpus height increases in the human sample, and many species differences are statistically significant. Most notably, height between the first and second permanent molars experiences a much greater size increase than humans on average (mean ζ =0.971), and of n=1,491 unique comparisons for this trait and time period, the lowest value of ζ =0.162. That so many measures show such non-trivial differences between species may reflect a global size increase during this period, but it could also relate to the development of molar tooth roots within the corpus.

Similar to results for corpus breadth, ζ statistics for each height measurement tend to be positive in periods 1-2 and 3-4, but negative in periods 2-3 and 4-5. In period 2-3, mean ζ is only slightly negative for most height measurements, but markedly more negative for height from the mental foramen to the corpus base and height at the

posteromedial margin of the corpus (i.e. lingual tuberosity). For this latter height, humans always undergo greater size change than *A. robustus*, though this is based a small sample (n=134) compared to other traits.

In period 4-5, humans on average undergo a greater amount of relative change in nearly all aspects of corpus height, and for many traits there is only a small chance of sampling human-like size change from the smaller *A. robustus* sample. The most pronounced difference between species in height change during this period occurs around the P₄ and mesial M₁. Here, mean ζ ranges from -0.08 to -0.13, and the difference is significant for height at P₄ (P ζ_{so} =0.038). Interestingly, each species is indistinguishable in the amount of height change at the position of the mental foramen, but mean ζ =0.141 for the height from the mental foramen to the alveolar margin. Humans in eruption stages 4-5 overlap entirely in this measure of alveolar height, whereas the sole *A. robustus* in stage 5 has a shorter height here than *A. robustus* in stage 4. As a result, ζ statistics for this period 4-5 can never be greater than 0 (Chapter 3.3). Thus, the significant difference between species in alveolar height growth in period 4-5 is likely an artifact of sampling and may not be indicative of an actual biological difference.

Summary of corpus height. The ζ test indicates that the amount and pattern of change in corpus height across ontogeny are fairly similar between humans and *A. robustus*, but there are still interesting differences. Again, because the ζ statistic is a relative measure, the fact that the *A. robustus* corpus is generally at or above humans' height range for most measures means that similar amounts of relative size change between species (i.e. ζ is not significantly different from 0) nevertheless allow *A. robustus* to attain a greater

height than humans by stage 5 for some measures. In addition, the predominantly positive (and significant) ζ statistics in period 3-4 followed by predominantly negative ζ statistics in period 4-5, indicates that each species differs in the timing of a mandibular height 'spurt,' *A. robustus* accomplishing a greater amount of height growth prior the full occlusion of M₂ and humans thereafter.

Overall lengths (traits 401-415)

There are four measurements of overall mandibular length. Two are poorly represented in the sample, and so their results should be viewed with caution. The first of these, the length from the alveolar septum anterior to M_1 to the point between the central incisors, fails to distinguish species. This may support the prediction made in Chapter 2, that the relatively small canines and incisors of humans and *A. robustus* might be reflected in similar patterns of jaw growth in this region (e.g. Simpson et al., 1990). However, it also agrees with John Hunter's over 200 year-old observation, that the distance between the symphysis and the "sixth tooth" (dm₂/P₄) does not change through ontogeny (Meikle, 2002). Moreover this is based on only 43 comparisons (Table 4.7)

The second of the less well-preserved length measurements, the length from the posterior mandibular condyle to the septum anterior to C₁, is only preserved in two specimens in the fossil sample, SK 63 in eruption stage 3 and SKW 5 in stage 5. This measurement reflects the length and depth of the oral cavity, and as such could be one of the most informative traits, were it not for the fact it is poorly represented in the fossil sample. The ζ test shows that humans never increase this measurement as much as *A*. *robustus*, but this result remains open to question since it is based on only 14

comparisons. Thus, the ζ test performed on these poorly represented traits suggests that species are indistinguishable in terms of size change of anterior mandibular length, but that the size of the oral cavity in *A. robustus* increases in size more than humans during ontogeny.

The other two mandibular length measurements are based on the more posterior corpus, and so are very well sampled. Across all possible comparisons, ζ statistics for average from 0.28-0.40, although these are not significant below the level of $P_{\zeta \le 0}=0.10$ (Table 4.7). The only one of these two measurements to be preserved across all successive dental periods is the length from the mental foramen to the posterior edge of the lingual tuberosity. The species difference in this trait across ontogeny is similar to other results – *A. robustus* undergoes greater size change in periods 1-2 and 3-4, but species are nearly indistinguishable in the amount of size change in periods 2-3 and 4-5. However, only in period 3-4 could the ζ test statistically distinguish species' amount of size change. The other well-sampled length measurement, from the mandibular foramen to the septum anterior to M₁, also undergoes greater size change in *A. robustus* ($P_{\zeta \le 0}=0.068$).

Summary of overall lengths. The ζ test indicates that the *A. robustus* mandible increases in antero-posterior length more greatly than humans across ontogeny. Species could not be distinguished, however, in the length of the anterior corpus (anterior M₁ to infradentale), which also measures mediolateral (unilateral) mandibular breadth. These results are not unexpected given these species' tooth sizes. The larger molars and premolars of *A. robustus* necessitate a longer jaw. Anterior tooth size is less disparate in

humans and *A. robustus* and so it is not unexpected for them to show similar patterns of relative size change in the length of the anterior jaw (Simpson et al., 1990). The only successive dental period in which species could be distinguished in terms of length growth is period 3-4: during this period, both humans and *A. robustus* must create space for the developing M₃ crown. Thus, at least some aspects of anteroposterior mandible growth probably reflect species differences in tooth size.

Overall (bilateral) breadths (traits 501-511)

Bilateral breadth measurements require relatively complete mandibles, and only four fossil individuals, in stages 1, 2 and 5, meet this criterion. Moreover, preservation limits measurements to the anterior aspect of the mandible, no further posterior than the P₄. Humans tend to have wider mandibles (at least anteriorly) than *A. robustus* throughout ontogeny, though the difference reduces over time (Appendix I). Overall increases in overall breadth tend to be greater in *A. robustus*, but these do not reach statistical significance at p≤0.05 (Table 4.7).

The only successive growth interval in which comparisons can be made is period 1-2. During this time, all measures of overall breadth show greater relative size change in *A. robustus*, and for three of these traits humans never achieve as much size increase as *A. robustus* (Table 4.8-9). An important caveat, however, is that these results are based on relatively few comparisons (n=31-163). These increases in overall breadth likely reflect the great increase in corpus breadth seen in *A. robustus* during this period. Corroborating this, the ζ test detects great differences between species in periods 1-2 and 1-5, but differences are much smaller in period 2-5. In fact, the ζ test suggests *humans* undergo

greater change in the distance between the lingual margins of the premolars in period 2-5 (mean ζ =-0.120, P_{ζ =0}=0.959, or the probability that ζ ≥0 =0.041). *A. robustus* undergoes greater size change between the lingual margins of the canines in this time (mean ζ =-0.175, P_{ζ =0}=0.024). This difference between species is explicable by the facts that *A. robustus* premolars are very large compared to humans', and that the posterior tooth rows are nearly parallel in *A. robustus* but parabolic (i.e. premolars more divergent) in humans. Caution is warranted here, however, as each of these traits is represented by only 42-49 comparisons.

Summary of overall breadth. The few bilateral breadth comparisons that can be made seem to reflect two biological phenomena. First, the *A. robustus* mandible is not as broad as humans' early in ontogeny, and it reaches the lower end of the human range of variation by stage 5 via greater relative breadth increase across ontogeny. Second, most of the increase in *A. robustus*' overall breadth reflects the drastic thickening of the corpus that occurs in period 1-2.

Ramus height & length (traits 603 and 607)

The ramus is delicate compared with the mandibular corpus, and so it is poorly represented in the present samples. Two measurements were selected that reflect the height and anteroposterior length of the ramus, and each of them undergoes almost 40% greater size increase in *A. robustus* across all possible comparisons, although with the small sample sizes, neither of these differences are significant below $P_{\zeta \leq 0}=0.05$.

The length of the ramus is approximated by the distance from the posterior ramus margin (on the alveolar plane) to the septum between the P_{3-4} (or dm₁₋₂). Looking across ontogeny (Appendix I, Table 4.8), ramus length growth is much greater in *A. robustus* in period 1-2, and at some point between periods 3-5. As no stage 4 *A. robustus* preserves this measurement it is unclear whether the second great length increase occurs only or chiefly in either period 3-4 or 4-5, or in both. Although ζ statistics for this trait are based on a relatively small amount of comparisons (n=258 for all possible comparisons), the size differences between species suggests that increased sample sizes would not erase the growth differences detected.

The height of the ramus is approximated by the distance from the mandibular foramen to the inferior ramus margin, and is also fairly poorly preserved in the fossil sample (present only in eruption stages 1, 3 and 4). Between stages 1-3, *A. robustus* tends to undergo greater height increase than humans, though not significantly so. During period 3-4, mean ζ =0.276 (P_{ζs0}=0.083) though again this difference is not quite statistically significant. However, these two periods of greater relative size increase result in the *A. robustus* undergoing a much greater relative increase in height between eruption stages 1-4 (mean ζ =0.628), with a very low chance of sampling *A. robustus* levels of change in the human sample (P_{ζs0}=0.007).

Summary of ramus height and length. The ζ test indicates that patterns of ramus growth differ between species in a manner similar to that for other traits. The *A. robustus* ramus is positioned relatively anteriorly on the corpus, such that it tends to flank the erupted

molars posteriorly to a greater extent than in humans (Table 4.8). As a result, greater length growth in *A. robustus* probably at least partly reflects this species' large tooth sizes.

Summary of the ζ test for individual traits

Most traits tend to undergo greater relative size change in *A. robustus*, and for all but one trait, the average ζ statistic is positive. Still, in the present cross-sectional data set, $\zeta=0$ is well within the range of values for all but one trait (Table 4.7); this outlier is based on only n=14 comparisons and so must be interpreted cautiously. The distribution of ζ statistics for all traits in successive eruption stages is practically identical to the distribution of ζ statistics for the pairwise size metric, but with more variation (Fig. 4.13).

The only successive interval in which any ζ statistic is most likely to be greater than 0 is period 3-4. The mean ζ for all traits within this period is a rather high 0.301, albeit not quite significant ($P_{\zeta \le 0}=0.091$). Seventeen of the full 29 traits can be examined during period 3-4 and all but five of these traits are positive at p ≤ 0.100 (i.e. $\zeta \leq 0$ for no more than 10% of all unique resampled comparisons). Thus, the significantly positive ζ for the pairwise size metric (section 4.2) seems to reflect a global increase in size affecting nearly every part of the *A. robustus* mandible during this time. Period 1-2 also has a relatively high average ζ , with less than 20% of the unique resampled statistics less than or equal to 0. Compared with period 3-4, fewer traits in period 1-2 show such large difference in size change between species. Traits with generally high D values during this period tend to be concentrated in the breadth of the corpus, suggesting this is a time of as corpus shape, as opposed to overall size, change.



Figure 4.13. Comparison of the ζ statistic for individual traits and for the pairwise size metric across ontogeny. The red box and whiskers are statistics for all individual trait, calculated between successive eruption stages. Overlain in green are the similar ζ statistics for the overall size metric, calculated from two or more traits (cf. Fig. 4.9). Boxes include the 50% quartiles, whiskers span the entire ranges of values, and thick red/green lines are medians.

These results reflect two chief differences in patterns of mandibular size change between humans and *A. robustus*. First, during period 1-2, the *A. robustus* corpus becomes markedly thicker than humans', driving the positive ζ statistics for the pairwise size metric in section 4.2. The *A. robustus* corpus maintains larger relative increases in breadth than humans through ontogeny, resulting in a very thick jaw. The human pattern of change in corpus breadth across ontogeny is quite different from *A. robustus*'. For measures at relatively anterior tooth positions (e.g. by the premolars), corpus breadth appears either to stay constant or even decrease (Appendix I), at least in this crosssectional sample. During infancy and early childhood (i.e. stages 1-3), the erupted deciduous, and emergent permanent, molars are the posterior-most teeth, flanked laterally by the ramus. As the mandible lengthens during growth by addition of bone at the corpusramus junction (Enlow and Hans, 1996), the ramus contributes less and less to these anteriorly displaced tooth positions (Table 4.8). The corpus, independent of the ramus, is of essentially the same size across ontogeny. Consistent with this pattern, for more posterior tooth positions where breadth generally increases during earlier eruption stages in humans, average breadth actually decreases between eruption stages 4-5, as the ramus contributes less to corpus breadth.

Second, after the *A. robustus* mandible becomes much thicker (e.g. "robust") by eruption stage 2, patterns of size and shape change are similar between species until period 3-4, when *A. robustus* advances to adult values for most traits (and the pairwise size metric). Positive D values are distributed across all types of mandibular traits during this time, meaning this may not be so much a time of *shape* as global *size* change. Interestingly, ζ statistics for many traits tend to be negative – albeit weakly and not significantly – for period 4-5, suggesting a delay in global size change for humans compared with *A. robustus*.

Symphysis and anterior corpus: Qualitative description

Results of the resampling analysis agree well with qualitative (visual) observations of the fossil sample. However, not all aspects of mandibular shape and anatomy could be captured with linear metrics. The complex contours of the mandibular symphysis made it difficult to obtain homologous measures on all specimens preserving the region. The issue of homology is further raised by the disparate posterior symphysis anatomies of humans and *A. robustus*. This section therefore qualitatively describes changes that occur to the *A. robustus* anterior corpus and symphysis.

A. robustus and human mandibles in eruption stage 1, with only the deciduous dentition, appear just broad enough to house the erupted deciduous molars. That is, the bony corpus is barely broader than the deciduous molar(s) it surrounds. This holds for all

three stage 1 *A. robustus* included in this study, as well as the similarly aged SK 852 (whose preservation precluded accurate measurement, though it is very similar to SK 3978). Stage 2 *A. robustus* mandibles (SK 61 and 62), however, are noticeably larger than those in stage 1, and whereas in humans the difference is not nearly as marked. Much of this marked size increase anteriorly reflects development of the corpus lingual to the deciduous molars and symphysis. Indeed, the most drastic differences in corpus breadth growth during period 1-2 occur around the premolars and symphysis (Table 4.7).

The posterior contour of the symphysis assumes a more posterior relative position across ontogeny. The post-incisive plane, as measured from infradentale to the point where the lingual surface begins to dip inferiorly toward the genial fossa, is about 4.0 mm shorter in SK 3978 than in SK 61 and 62. Where it is preserved in eruption stages 1-3, the inferior transverse torus is poorly developed, barely projecting posteriorly beyond the genial fossa and extending to the antero-posterior level of the dm₁ when viewed superiorly. The posterior symphysis is not preserved in the youngest stage 4 fossils, SK 25 and 55b (whose M₂s are freshly in occlusion and premolars in the process of erupting), although it is probably at the same relative position as in stages 1-3. However, by the time the permanent premolars are occluded later in stage 4 (as judged by SK 6 and SKX 4446), the inferior torus has migrated posteriorly to the position of the P₄. In all older *A*. *robustus* (SKW 5 in stage 5 and adult *A. robustus*), the inferior torus is also extended to the antero-posterior level of P₄.

As the inferior torus migrates posteriorly, the anterior contour of the symphysis assumes a more vertical profile across ontogeny. In stages 1-3, the anterior surface arcs postero-inferiorly, beginning relatively high on the symphysis in lateral view. This

surface is poorly preserved in stage 4 (SK 25 and SKX 4446), but the superior part of the contour becomes more vertical anterior to the incisor roots. In SKW 5 and all adult mandibles where the anterior symphysis can be observed, its contour is vertical with the posterior arc beginning lower than the mental foramina when viewed laterally. In SKX 4446 the corpus is broken just laterally to the symphysis, revealing the anterior contour does not merely cover the incisor roots, but rather cortical bone becomes thicker inferiorly (i.e. the root appears procumbent relative to the vertical symphysis anterior to it).

The increasing verticality of the anterior symphysis coupled with a posteriorly migrating posterior symphysis is how *A*. robustus obtains a relatively thick and robust symphysis through growth. Symphyseal development observed in these fossils is in accordance with Bromage's (1989) morphogenetic interpretations of bony remodeling fields. Bromage found both the anterior and posterior symphysis was to be depository in *A. robustus* subadult mandibles, and interpreted this pattern of deposition to reflect a postero-superior direction of growth at this surface. The mesial migration of the posterior migration of the symphysis.

Summary of Results

Poor preservation and small sample sizes for *A*. *robustus* necessitated the development of a new procedure, the ζ test, to compare patterns of age-related size change with humans. The ζ test compares the amount of relative change in size between dental eruptions, by resampling all possible pairs of *A*. *robustus* and humans: $\zeta = [(\mathbf{R}_{OLDER}$

 $/\mathbf{R}_{YOUNGER}) - (\mathbf{H}_{OLDER} / \mathbf{H}_{YOUNGER})]$. This chapter presented the validation and results of this new method.

The first section demonstrated that both the pairwise size metric and the ζ test perform as expected. First, the pairwise size metric – the geometric mean of the set of traits shared by pairs of specimens – was used as a way of comparing the overall size individuals when they do not all preserve the same traits. The metric both distinguished human and A. robustus sizes across dental stages, and the human 'sizes' were shown to correspond well with humans' characteristic pattern of body size change (Table 4.1, Fig. 4.2). Next, the ζ test was used to compare the human sample with itself to assess the risk of type I statistical error. As expected, the test generally pointed to a common pattern of size change within the human sample (Table 4.2, Figs. 4.3-4). The range of ζ statistics in the test validation, while not as great as in comparisons of humans with A. robustus, nevertheless demonstrates the effects of variation inherent to a cross-sectional sample. Finally, preservation and the difference in eruption status between individuals have a negligible influence on the value of the ζ statistic (Table 4.3, Figs. 4.5-7). Variation in ζ tends to decrease as more traits were used to measure size, though this is likely due in part to the fact that more specimens are likelier to share fewer traits.

The next section presented the results of the ζ test comparing human and *A*. *robustus* mandibles for the pairwise size metric. The test detected notable differences in patterns of size change between these species (Table 4.4, Fig. 4.8). Species overlap in size in the first eruption stage, but by eruption stage 5 the *A*. *robustus* mandible has enlarged much more than humans' (Fig. 4.1). This difference in relative change was not evenly distributed across ontogeny (Fig. 4.9). *A*. *robustus* size increases more during

periods 1-2 and 3-4 and there is a very low chance (1-10%) of randomly sampling a human pair so different in size as comparably aged *A. robustus*. Conversely, humans tend to increase relatively more than *A. robustus* in periods 2-4 and 4-5, although the species differences in these latter periods are not as large or significant as the differences in periods 1-2 and 3-4. Resampled metadata (preservation, k, and difference in eruption stage, da) had significant rank correlations with the value of $\zeta \square \square \square \square \square \square \square$. However, these correlations are fairly low, explaining little variance in ζ , and appear to reflect sample preservation more than the performance of the ζ test itself (Figs. 4.10-11).

As shown in the final section, results of the ζ test for individual traits were similar to results for the pairwise size metric (Fig. 4.13). Across all age comparisons, all traits but one (corpus height at M₂) undergo greater size change on average in *A. robustus* (Fig. 4.12), though none of these differences are statistically significant at the traditional threshold (Table 4.6). Looking at these traits between successive dental eruptions revealed significant differences between species' patterns of size change (Table 4.7). Principally, period 1-2 is the time when *A. robustus*' corpus and symphysis become much broader than humans, while period 4-5 sees a more global size increase in *A. robustus*. At least some species differences in size change of individual traits likely reflects the need for *A. robustus* to accommodate much larger postcanine teeth than humans.

Chapter 5

Implications of A. robustus Mandibular Growth

The present work contributes both a new method and new results of interest to the field of anthropology. This chapter discusses the importance of these contributions, treated in three parts. First, I review the reasoning of the ζ test and its performance in comparing both a single sample with itself, and humans with the extinct *A. robustus*. Although the ζ test was able to detect differences in patterns of size change between these species, the study was hampered by certain limitations that could be reexamined in future work. Second, I discuss the implications of the results for evolutionary developmental biology. Evo-devo asks how species' distinct morphologies arise during development, and so it is important that the ζ test was able to detect ontogenetic differences between each species. Finally, I discuss the implications for the reconstruction of hominid life history. The results presented here allow inferences to be made about life history variables including growth rates and maturation landmarks, which ultimately addresses the question of the antiquity of the human life history strategy.

5.1 The *ζ* test

Motivation for the ζ *Test*

When two groups are compared, a statistical argument can be laid out as to the significance of the comparison – are the groups more similar or different than expected by chance? Because of the abysmal likelihood for any animal to fossilize after it dies,

fossils present major challenges to statistical analysis. For one, sparse samples mean that natural variation is likely to be underestimated. Moreover, the one thing of which there is a lot in a fossil sample is missing data.

Resampling and randomization statistical techniques have become common and successful means of analyzing and comparing groups in the past few decades. This is because personal computers have become faster and more available, and statistical programming languages such as **R** have become increasingly available (e.g. free) and user-friendly. As noted by Sokal and Rohlf (1995) nearly 20 years ago, resampling methods allow anyone to break the bonds of parametric statistics relying on a theoretical distribution of some test statistic, to create empirical distributions of any kind of test statistic they can imagine. As a result, a whole world of previously unavailable research questions has opened to paleoanthropologists.

While randomization has helped address the issue of fossils samples not meeting certain theoretical distributions, that fossil samples likely underestimate normal biological variation can never be avoided. Poor sampling has thus largely precluded the statistical treatment of ontogeny in fossil samples. The ζ test was therefore conceived with these challenges in mind: first, how does one compare multiple traits in multiple individuals across multiple ages, in light of copious missing data? Second, given the present *A. robustus* series, can one reject the hypothesis of indistinguishable patterns of size change with age? This raises the corollary question of whether the distinct aspects of longitudinal growth – specifically changing growth rates – can be discerned from cross-sectional samples.

One of the motivations for devising the ζ test was the desire to extract as much information as possible about mandibular growth from a sample with a large amount of missing data. In the present dataset of up to 29 measurements in 122 humans and 13 *A*. *robustus*, 47% of the cells are missing data. Missing data are a problem for paleontology and anthropology, and researchers have dealt with this in different ways. The simplest way is to limit the analysis to only complete specimens. In studies of hominid ontogeny, this means comparing size or shape changes between a single developmental stage and adulthood (e.g. Richtsmeier and Walker, 1993; McNulty et al., 2006). This has the downside of further underestimating intraspecific variation in fossil samples, and especially reduces the number of ontogenetic comparisons that can be made. However, it does avoid assumptions (and often subjectivity) required by imputation methods that include fragmentary individuals by estimating their missing values (Mundry, 1999; Strauss and Atanassov, 2006). Imputation, however, is inappropriate in the present dataset as there is too large an amount of missing data.

Rather than specifically limiting fossil to complete specimens, or to adding error to the analysis by essentially guessing missing values, the ζ test was devised to assess patterns of variation by making as few assumptions as possible. The test simply asks how probable it is to sample the same pattern of age-related variation in the fossil sample from a larger reference sample of living humans. The test assumes nothing about the underlying pattern of growth in either species, but it does make a few other assumptions that do bear on the test's outcome. These are reviewed below and discussed in terms of the test's shortcomings and successes.

Assumption 1: Cross-sectional Data Represent Realistic Patterns of Growth

The ζ test essentially boils down the question of early hominid ontogeny to one of sampling: where ontogenetic variation can be compared between humans and *A. robustus*, how likely are the two to be distinguishable? The key assumption made here is that any individual could have grown to be the size of another individual in a later eruption stage. This assumes that patterns of size change detected in the cross-sectional samples for each species reflect actual patterns of longitudinal (i.e. individual) growth. The human growth curve in height and weight is unique among animals, but individual growth patterns can be so variable that 'the' human growth curve gets lost in the noise of cross-sectional data (Tanner, 1951; Leigh, 1996). The necessarily cross-sectional nature of skeletal samples therefore raises two important and related problems for statistically comparing samples. First, the *A. robustus* sample could be small enough that it may not adequately capture the species' pattern of growth. Second, the human sample may be large and variable enough that it could encompass the fossil pattern of ontogenetic variation, potentially resulting in the acceptance of a false null hypothesis.

These two points may not be problematic for the present study. Results presented in Chapter 4 demonstrate that the ζ test is an effective means of comparing patterns of size change in skeletal samples. The validation analysis showed that the test will generally indicate no difference between groups when patterns of size change are in fact similar. Furthermore, the test indicated clear (though not always statistically significant) differences in patterns of size change between humans and *A. robustus*, consistent with visual appraisals of each species' jaws.

That results of the intraspecific (validation) and interspecific analyses differ strongly suggests that that the interspecific differences detected are real biological phenomena rather than artifacts of sampling. Figure 5.1 compares the ζ statistics from both the human-*A. robustus* and test validation distributions, highlighting the magnitude of difference in relative size change between *A. robustus* and humans. The distribution of ζ statistics from the validation (in green; cf. fig. 5.1) is centered at ζ =0, whereas the human-*A. robustus* distribution is shifted to the right of the human-only distribution, reflecting the fact that on average there is greater overall size difference between eruption stages in *A. robustus* compared with modern humans. This holds true no matter how many traits are used (Fig. 5.2). In addition, the human-only distribution is symmetrical while the human-*A. robustus* distribution has a longer right tail (i.e. a small group of relatively high ζ statistics).

This difference between randomized distributions is not an artifact of there being twice as many human-human comparisons as human-*A*. *robustus* comparisons. For instance, if ζ statistics from the validation distribution (n=58,709) are resampled 5,000 times to match the size of the human-*A*. *robustus* distribution (n=29,148), the average ζ never exceeds 0.003. This is still lower than 12.6% of the human-*A*. *robustus* ζ statistics, and nowhere near the interspecific mean ζ (0.310). Thus, the ζ test strongly points to different patterns of growth even though the human-*A*. *robustus* distribution was not quite 'significantly' different from 0.



Figure 5.1. Distribution of ζ statistics measuring difference in overall size change, from the validation and interspecific tests. The green distribution is that from the validation comparing the human sample with itself (cf. Fig. 4.3). The blue distribution contains all unique ζ statistics comparing overall size change between humans and *A. robustus* (cf. Fig 4.9).

Assumptions built into the ζ test also raise the question of whether the species differences implied by the test accurately reflect each species' growth. It was assumed that any specimen in one eruption stage could grow to the size of specimens in any older stage. However, in a cross-sectional sample with overlap in size between individuals in different eruption stages, this means that it is conceivable to sample an unrealistic shrinking between dental stages (Fig. 2.3). To address this, the 'growth ratio' of older to younger was set equal to 1 in the calculation of the ζ statistic in such cases (e.g. Fig. 3.1). This potentially introduces bias into the calculation of ζ , since ratios less than one are altered but higher ratios are not.



Number of traits (k)

Figure 5. 2. The relationship between the value of D and the number of traits used to calculate D. The results of the validation are in green while the results of the human-*A. robustus* analysis are in blue. Boxes include the 50% quartiles, whiskers include the entire ranges of values, and thick blue/green lines are medians.

However, in the development of the ζ test I found the results to be essentially the same no matter how I treated growth ratios less than one. I ran three trials of 50,000 resamplings each, in which such 'negative growth' ratios were either [1] not altered at all, [2] omitted, or [3] coerced to 1 (as in the results presented here). In trials [1] and [3], in which negative growth ratios were retained, there were about 22,000 unique comparisons, compared to just under 18,000 when negative growth ratios were omitted. The mean and range of ζ were very similar regardless, as were the probabilities of ζ being less than or equal to 0 (cf. Table 4.4). Thus, the fix used in the present study (trial [3]) does not seem introduce substantial bias in the present case. That the results of all three trials agree

showcases the flexibility of the ζ test, that it can be altered to incorporate different assumptions (e.g. whether shrinking is possible during growth).

In spite of the apparent success of the ζ test, it is still subject to the difficulties of cross-sectional data. The ζ test detected a substantial amount of intraspecific variation in age-related size change for humans (e.g. Figs. 4.3-4), creating a substantial amount of overlap between species in patterns of size change (Figs. 4.8-9). Even though the ζ test depicted regular differences between species in size change, within the cross-sectional human sample it is not impossible to randomly sample an *A. robustus*-like pattern of age-related size change, for both the pairwise size metric and individual traits.

This issue with cross-sectional data is highlighted by the ζ statistics for corpus breadth at the septum between P₄-M₁ between eruption stages 1-2 (Chapter 4.3.1). Even though SK 3978 and 64 are in the same eruption stage (stage 1) and show identical stages of tooth crown formation (Conroy and Vannier, 1991b), the former is larger than the latter for most traits they share in common. Intraspecific variation within this age group is such that the results of the test could differ substantially depending on which fossil is used to calculate ζ . In contrast, breadth at P₄-M₁ is similar between the only two fossils in stage 2 (SK 61 and 62), so that ζ statistics do not differ depending on which individual is used. It is therefore possible that an increased *A. robustus* sample size might erase any difference between humans and *A. robustus* detected by the ζ test. In this regard, it could be informative to add the *A. robustus* subadults from Drimolen to the analysis (insofar as these fossil samples can be realistically combined).

Assumption 2: Ages and Stages

The analysis assumes that the amount of time between eruption stages is the same (or at least proportional) in each species, but this may not be the case. The greater size increase of *A. robustus* mandibles during periods 1-2 and 3-4, for example, would not be as remarkable if the absolute duration of these periods were much longer in *A. robustus* than in humans. Conversely, several authors have argued on the basis of dental development that australopithecine ontogeny was a rapid, more similar to living apes than humans (e.g. Dean et al., 1993). While this could still imply a similar relative amount of time between eruption stages, it would make the greater magnitude of size increase in *A. robustus* in periods 1-2 and 3-4 even more dramatic.

However, eruption stages are more important than chronological ages from a life history standpoint, as these stages are roughly coincident with other maturational events in humans and apes (Bogin, 1999; Zihlman et al., 2007; Hochberg, 2012 b). Given the similarity between humans and *A. robustus* in molar crown development (Mann 1975) and eruption (Smith, 1986), the assumption of proportional time between eruption stages seems reasonable. In addition, dental development in wild chimpanzees of known age has been shown to be slightly later than published captive standards, slightly blurring the distinction between ape-like versus human-like developmental timing (Zihlman et al., 2004; Smith et al., 2010)

Assumption 3: A Mixed-Sex Sample

This study made no assumptions about sex, but rather pooled the entire subadult mandibular samples of *A. robustus* and the humans from Libben. For one thing, the sex of fossils can almost never be known with certainty, except in cases of especially large or

small individuals. For instance, the *A. robustus* cranium SK 48 has been interpreted as female based on canine size (Wolpoff, 1999), but as male when compared with smaller cranial remains (Lockwood et al., 2007). Sex of the human subadults from Libben is also uncertain in most cases, as they lack adequate postcrania (viz. innominates) for sexing. Even if it were possible to reliably estimate the sex of fossil mandibles, sexual size dimorphism tends not to appear until after the time of puberty in most species, including humans and apes; the present analysis assumes the same is true in *A. robustus* but this is still a hypothesis. Pooling sexes also has the benefit of keeping the fossil sample as large as possible.

The results presented in Chapter 4 reveal a large range of size variation within eruption stages (Appendix I). This is especially true for the stage 4 subsample, as this period roughly coincides with puberty and there is great individual and sexual variation in the timing of the pubertal growth spurt. As mentioned above, sex cannot usually be reliably determined for subadults before puberty, and so sex is neither known nor estimated for most individuals. The great range of sizes (individual traits and the size metric) and associated ζ statistics during eruption stage 4 (e.g. Fig. 4.2) is consistent with this pattern. The similarly large range in stage 1 is less explicable by sexual dimorphism, and it is probable that much of the size variation may reflect age variation within eruption stages.

The potential influence of sex variation on the results is most salient for eruption stages 3 and 5, which are each represented by only one fossil (SK 63 and SKW 5, respectively). SK 63 is very similar in size and anatomy to stage 2 *A. robustus* and is smaller than stage 4 *A. robustus* for most measurements; ζ statistics were slightly

negative (but not significantly different from 0) for period 2-3 but were significantly positive for period 3-4. Similarly, SKW 5 is also fairly small, within or below the range of stage 4 *A. robustus* for many mandibular dimensions (Appendix I), and similar in size to small adult mandibles such as SK 74 and SKX 5013; ζ statistics for period 4-5 were slightly negative (but again not significantly different from 0). If either of these fossils is especially small for their eruption stage (i.e. due to sexual size dimorphism; see below), this would reduce the difference between species' growth detected in this study. It may therefore be worth investigating whether teeth can be used to estimate individuals' sex, the difficulty being that not all specimens preserve the same teeth. However, an enlarged fossil sample is necessary to adequately assess specimens' sex and this potential influence on *A. robustus* growth.

Summary of the ζ Test

The ζ test is a novel means of comparing patterns of variation in the face of small samples and large amounts of missing data. The test (for the pairwise size metric) behaved as expected when comparing patterns of size change in the human sample with itself, and contrarily pointed toward differences between humans and *A. robustus*. It should be remembered that the test was run exactly the same in both the validation and interspecific analyses, yet the outcomes were quite different in each. In addition, the test comparing size change for individual traits further pointed to differences between species that are consistent with gross differences between species' mandibles. Using this new test, the present study is the first to present both statistical and skeletal evidence for different patterns of bodily size change between a human and australopithecine sample.

Even though the ζ test strongly pointed toward species differences in patterns of size change, viz. between eruption stages 1-2 and stages 3-4, many of these differences were not statistically significant at the traditional p<0.05 level. This cautions against interpreting the results as a clear indication of species differences, but can be at least partly explained by the test's three key assumptions: 1) species' true patterns of growth and development can be represented with cross-sectional data; 2) the time between dental eruptions is comparable/proportional between species; and 3) species' patterns of sexual dimorphism will not influence the comparison of growth between samples. The ability of the ζ test to detect differences between humans and *A. robustus* shows that these were safe assumptions to make given the limitations of the samples used. However, the species differences detected here may be validated or refuted by future fossil discoveries and advances in methods for sexing and aging skeletal samples.

5.2 Evo-devo

Chapter 2.2 presented the null hypothesis of conserved developmental processes shared by *A. robustus* and humans, which predicts either that each species should not differ in size change for any trait, or that each trait would differ similarly between species across all traits. The alternate hypothesis of different postnatal development predicts that species differences should be found for only a subset of traits. The ζ test for individual traits (Chapter 4.3) rejects the null in favor of the alternate hypothesis. This section discusses the implications of this result for hominid evo-devo, examining whether differences might be explained by jaw biomechanics and adaptation, or by tooth size and development. I then revisit the predictions from morphogenetic hypotheses of facial

growth that could not be directly examined in this study (Bromage, 1989; McCollum, 2008). I also suggest different ways of addressing questions arising from this study.

The ζ test was used to compare the *A. robustus* and human patterns of relative size change for individual traits across ontogeny. For most traits, the amount of relative change in size between dental eruptions in *A. robustus* can be found in a human sample, with the exception of the time between eruption stages 3-4 and less often between stages 1-2. Between stages 1-2, *A.* robustus experiences significantly greater increases in both corpus and overall mandibular breadth, and in corpus length posterior to the mental foramen. Between stages 3-4 the *A. robustus* mandible exhibits significantly greater increases in the ramus and most aspects of corpus height, corpus breadth around the premolars, and corpus length. Thus stages 1-2 appear to be the period during which the *A. robustus* mandible becomes characteristically "robust," and stages 3-4 witness a greater global size increase than is usually seen in humans.

One of the most notable findings of this study is that the *A. robustus* mandibular corpus becomes markedly more robust than humans early in postnatal ontogeny, rather than prenatally. Many other studies of comparative craniofacial growth have found that different groups (species or populations within species) can be distinguished in the earliest age groups (e.g. Fukase & Suwa, 2008; Ponce de Leon & Zollikofer, 2001). Consistent with these studies, the present study demonstrates that some key differences in the mandibles of humans and *A. robustus* are established early in postnatal growth.

During eruption stage 1, both species overlap in corpus height and breadth, though the few *A. robustus* values tend fall in the high end of the human range (Appendix I). In addition, unique features such as the human chin are present by the time of birth

and are notably absent in the youngest *A. robustus*. The youngest specimens in this eruption stage have their deciduous teeth nearly fully erupted, which means that at least two years of postnatal growth have occurred prior to this eruption stage (by human standards; Liversidge, 2003). Thus, even though species' mandibles are distinct in the earliest eruption stage, it cannot be definitively said that such differences were established prenatally.

Postnatally, between eruption stages 1 and 2, each species follows similar patterns of relative height change, but *A. robustus*' corpus breadth increases more than humans (Table 4.6). This relationship can be examined with the "robusticity" shape index of corpus breadth divided by height at a given tooth position (Daegling, 1989; Teaford & Ungar, 2000). Even though robusticity throughout most of the corpus decreases in both species across ontogeny (Appendix II), the *A. robustus* decreases are usually not as substantial as those implied by the human cross-sectional sample. The ζ test therefore points to an important role of postnatal ontogeny in creating the robust *A. robustus* jaw and humans' relatively gracile one. This raises the question of *why* the *A. robustus* corpus becomes so remarkably broad compared with humans.

Below, I review several possible reasons for this species difference. The relatively broad *A. robustus* mandibular corpus has been interpreted as being adapted to withstand large stresses generated during chewing (Wolpoff, 1975; White, 1977; Daegling, 1989; Teaford & Ungar, 2000). This anatomy could be the direct result of bone itself responding adaptively to the use of the mandible during ontogeny. More indirectly, the growth difference could be genetically determined, inherent to the mandibular development of the species, because of natural selection to create a robust corpus. Also

indirectly related to diet, the larger postcanine teeth of *A. robustus* may necessitate housing in a larger jaw. In this case, the species differences in changing mandibular breadth and robusticity may reflect size of the developing dentition rather than biomechanical adaptations. Not all these possibilities are necessarily mutually exclusive, and each is treated below.

Jaw Biomechanics and Feeding Adaptation.

As discussed in Chapter 1, the *A. robustus* mandible is able to help generate, and then withstand, large muscle forces associated with chewing. This strength is provided by a relatively broad (i.e. "robust") corpus and symphysis, as well as the distribution of cortical bone thereabout. Because human and *A. robustus* mandibular shapes reflect different biomechanical properties, it is possible that growth differences between these species reflect divergent dietary demands. Indeed, Taylor (2002) found that African ape (*Pan* and *Gorilla*) mandibles generally grew so as to increase resistance to masticatory stresses across ontogeny.

However, biomechanical inferences, especially related to ontogeny, must be made cautiously. Taylor (2002, 2006) also found that not all differences in shape (and shape change) between species were fully predictable from their dietary differences. Similarly, Daegling (1989, 1996) warned that anatomical differences between African apes do not necessarily correspond to significant biomechanical differences. Daegling (1989, 2007) has also shown that raw corpus breadth measures and the "robusticity index" are poor predictors of strength against torsion and bending, stressing that while a robust shape may create a cross-section strong against twisting and bending, the distribution of cortical

bone about the cross-section is key to determining strength. Finally, Daegling and Hylander (2000) review the case that the robust australopithecine mandible is actually overdesigned for handling quotidian chewing forces; some inferred biomechanical advantages of the *A. robustus* may instead be non-adaptive 'spandrels' (Gould and Lewontin, 1979). The following discussion treats the robust corpora of subadult *A. robustus* as having greater strength than humans, a reasonable supposition that must be hypothetical until data on cortical bone distribution are available to more accurately estimate the strength of *A. robustus* youths' jaws (Daegling, 1989; Taylor, 2006).

If the relatively more robust corpus of *A. robustus* reflects a functional adaptation to chewing stresses experienced in life, the relative increase in *A. robustus*' (inferred) corpus strength during period 1-2 could reflect early weaning and early adoption of an adult diet in this species. Indeed, *A. robustus* deciduous teeth experience a great deal of (dietary) wear prior to M_1 eruption and it has been suggested that this pattern is indicative of early weaning (Aiello et al., 1991). Aiello and colleagues likened this pattern with extant gorillas rather than early-weaning humans. That is, they interpreted early weaning to be indicative of a relatively fast overall life history and short maturation period like gorillas, rather than a truncated infancy like in humans.

However, in support of a more human-like pattern of weaning, McKinley (1971) found that the estimated survivorship curve of *A. robustus* from Swartkrans and Kromdraai could be best explained by an estimated age at sexual maturity of 11 years and interbirth interval (i.e. time from birth to weaning plus gestation) of 3-4 years (maximum). This is significant because this spacing is at the low end of the wild chimpanzee (Marsden et al., 2006) and gorilla ranges (Kennedy, 2005; Robbins et al., 2009), but the

(modeled) age at sexual maturity is comparable to chimpanzees' and slightly older than gorillas'. Short birth spacing would have been key to *A. robustus*' reproductive success and maintenance of a viable population in light of high mortality and short life spans (Wolpoff, 1979; Lovejoy 1981). Little other work on *A. robustus* paleodemography has been done, but the issue deserves further attention, especially given increased mandibular samples (Brain, 1981, 2004; Sutton et al., 2009; Keyser et al., 2000; Pickering et al., 2012) since McKinley's (1971) study.

Further evidence as to potential early weaning in A. robustus could also come from a more systematic and comprehensive examination of patterns of macro- and microwear on deciduous teeth. Aiello and colleagues (1991) compared australopiths only to extant apes. Examining a wider range of variation within and between species, including other hominids or species that wean at different ages (i.e. early), could help shed considerable light on the relationship between weaning and tooth wear (e.g., Bullington, 1991) in early hominids. An important difference between humans' and gorillas' early weaning is that in humans it is accompanied by specially prepared, high quality foods for weanlings (Sellen, 2007), whereas a gorilla weanling would be ingesting harder foods like adults do (Kennedy, 2005; Humphrey 2010). This predicts that macroscopic levels of deciduous wear should be greater in gorillas than humans, providing a baseline for understanding the likelihood and nature of early weaning in A. *robustus* and other early hominids. However, the large variability in practices of weaning, juvenile feeding and diet across human populations (Sellen, 2007), highlights the need to examine multiple human samples (viz. prehistoric populations).

The isotopic composition of teeth could provide further information about possible early weaning in hominids. Carbon, nitrogen and other elements from an animal's diet contribute to the development of the animal's various tissues, including teeth (reviewed in Ungar & Sponheimer, 2011). All plants discriminate against the ¹³C isotope in favor of ¹²C in photosynthesis, but do so differently such that leafy plans (C₃ pathway) have lower ratio of ^{13/12}C than sedges and tropical grasses (C₄ pathway). The carbon isotopes ratios (expressed as δ^{13} C) within animal teeth have been shown to be effective discriminators of different types of diet. Animals eating C₄ plants will tend to have higher δ^{13} C values than animals that eat leaves and fruits.

Analysis of the isotopic composition of *A. robustus* teeth shows a wide range of δ^{13} C values, suggesting the hominid had a rather variable, generalized diet (Lee-thorp & Sponheimer, 2006; Ungar & Sponheimer, 2011). Sponheimer and colleagues (2006) further analyzed the isotopic composition of enamel across the crown height of individual *A. robustus* teeth, thus examining dietary variation within an individual over the time the tooth was developing. Values of δ^{13} C varied in all four *A. robustus* teeth sampled, and in two of these showed a fairly stable decrease in δ^{13} C. Of these two, one was an M₃ and the other an unspecified molar: the former would not have been formed during infancy and the full transition to an adult diet, and the latter may have been but this cannot be known for sure.

A similar analysis of the isotopic composition of *A. robustus* tooth crowns could therefore shed light on the question of early weaning in this species. For instance, I₁ and P₃ crowns would have been forming during period 1-2 when *A. robustus* first experiences a drastic increase in corpus breadth (Conroy and Vannier, 1991b). A 'longitudinal'

analysis of these teeth in *A. robustus* could test the null hypothesis that δ^{13} C values will be consistent throughout the height of the crown. Alternatively, a consistent change in in δ^{13} C values within teeth could provide evidence of weaning during this time. However, carbon isotope ratios can indicate a general dietary tendency (e.g. leaves vs. grasses), but cannot unequivocally tell exactly what an animal was eating and when. Nevertheless, it would be very informative to further investigate the relationship between jaw growth and the isotopic correlates of diet.

Regardless if hypothesis of early weaning in *A. robustus* is correct, the question still remains *why* the corpus so drastically broadens (and presumably strengthens) during this time. On the one hand, the (inferred) increased strength of the *A. robustus* mandibular corpus relative to humans during period 1-2 could be genetically programmed into the species' development. Presumably this could be due to natural selection acting on subadults, favoring those who can achieve foraging independence from older individuals earlier. Unfortunately, this is a rather untestable hypothesis. On the other hand, *A. robustus* early increase in corpus breadth may reflect a mechanical response of bone to the adoption of an adult-like diet. It is difficult to distinguish which of these two is most likely, if either.

The Influence of Developing Teeth

Dietary differences between humans and *A. robustus* are reflected in the size and morphology of each species' teeth (Wolpoff, 1975). In Chapter 2 it was predicted that because humans and *A. robustus* have small canines and incisors compared with other hominids and hominoids, they may show similar patterns of size change in the anterior
corpus; this prediction was borne out, although it was based on a small number of comparisons. Conversely, *A. robustus*' massive molars and premolars might drive some of the differences in posterior corpus length and breadth between species (Wolpoff 1975; Plavcan & Daegling, 2006). The present study could not test this relationship directly, but knowledge of tooth crown formation in *A. robustus* (Conroy and Vannier, 1991) hints that some of the differences in corpus growth between *A. robustus* and humans may well result from differences in the size of the developing permanent dentition.

The ζ test comparing growth in length of the posterior corpus is consistent with the hypothesis that some aspects of jaw growth reflect the development of the dentition. The distance from the mental foramen to the distal-lingual margin of the lingual tuberosity could be compared across all successive eruption stages, but indicated significant differences in size change only during period 3-4 (Table 4.7). This measurement tracks the growth of the lingual tuberosity, which must grow posteriorly to accommodate the molar crypts and crowns as they develop (Enlow and Hans 1996). The lingual tuberosity of the only *A. robustus* in stage 3 (SK 63) encloses a partially developed M₂ crown, and there is no hint of M₃ crown formation distally (Mann, 1975; Conroy & Vannier, 1991b). By eruption stage 4, the *A. robustus* lingual tuberosity has extended posteriorly to surround a developing M₃ crown in all but one specimen (SK 25, the least mature and presumably youngest in stage 4). It is therefore reasonable to connect species differences in corpus length growth to reflect the development of differently sized M₃s.

The ζ test is more equivocal as to the influence of M₂ crown formation for species differences in posterior corpus length growth during period 1-2. While *A. robustus* does

undergo greater length increase than humans here, the difference is not significant ($P_{\zeta \le 0}=0.162$). None of the stage 1 *A. robustus* shows evidence that the M₂ has begun to form, but the only stage 2 fossil preserving the distal corpus (SK 62) has formed at least the upper portion of the M₂ crown. Results therefore hint that developing large tooth (viz. molar) crowns may be responsible for some differences in jaw growth between humans and *A. robustus*, but this deserves further testing.

Results are also equivocal as to whether *A. robustus*' large postcanine teeth drive its relatively large increases in corpus breadth. Between eruption stages 1-2, the ζ test indicates that *A. robustus* generally undergoes greater relative change in corpus breadth at P₄ (mean ζ =0.112, p=0.073). Of the three *A. robustus* in eruption stage 1, none of them has any development of an unerupted P₄ crown (Conroy and Vannier 1991b). The two *A. robustus* in stage 2 have developed at least part of this crown, however, meaning some if not all of the crown's adult breadth had been attained. It is therefore plausible that appearance and development of the P₄ crown drives some of the growth in corpus breadth between these eruption stages.

But at the same time (period 1-2), the ζ test shows a significant species differences in breadth growth at the positions of the (unerupted) P₃ and M₁, and these tooth crowns are already partially formed in stage 1 *A. robustus*. While the P₃ and M₁ suggest tooth crown formation may be less involved in driving corpus breadth growth, it is also possible that *root* formation is involved. For example, later in development, the P₄ root of SK 6 (stage 4) becomes buccolingually broader than the crown, which interestingly corresponds with *A. robustus*' significantly greater increase in corpus breadth at this position in periods 3-4 (Table 4.7). It was beyond the scope of the present study to examine the relationship between tooth formation and hominid jaw growth in more detail, but results are consistent with other studies more directly analyzing the relationship in other taxa. Boughner and Dean (2004) tested whether space in the growing jaw influences the initiation of molar crown formation, in *Pan* species and baboons (*Papio hamadryas*). They found that the length of the corpus always provided "excess" crypt space distal to the mineralizing molar crowns, and that posterior corpus length increased most when the M₃ crown began to mineralize. Moreover, the size of this space and its elongation were greater in *Papio* than in *Pan*, consistent with the longer molar teeth in the former genus. Similarly, Fukase & Suwa (2010) found that the faster-growing incisors of recent Japanese mandibles compared with archaeological samples was concomitant with faster anterior corpus height growth in the recent mandibles.

In summary, the present study did not directly examine whether the formation of tooth crowns influences size growth of the mandible, but it is very likely that at least some differences between mandibular growth between *A. robustus* and humans (i.e. in length) reflect the development of *A. robustus*' large postcanine teeth. Future research should therefore examine the relationship between jaw and tooth crown development across a wider range of taxa, especially including megadont species like australopiths. In such a study, CT data could be used to obtain mandibular metrics such as those employed here, combined with absolute measures of tooth crown width, enamel thickness, root size, crypt size, and thickness of cortical versus trabecular bone around developing teeth in subadults.

Mandibular Growth Rotations

The very tall ramus of *A. robustus* has led to the hypothesis that this species undergoes substantial postnatal growth rotation of the mandible and rest of the face (Fig. 2.2; Bromage, 1989; McCollum, 1997, 2008). McCollum (1999) has even suggested that other aspects of the robust australopithecine face, including its thick hard palate and highly hafted face onto the braincase, are merely developmental byproducts of this rotation. Facial rotation is generally measured using angles between parts of *complete* jaws and faces (e.g. Björk, 1969; Björk and Skieller, 1972; Solow and Houston, 1988; Wang et al., 2009), but such data cannot be collected on the majority of the fossil sample. Because facial rotation is essentially a phenomenon of disparate height increase between the front and back of the jaw (Björk and Skieller, 1972), rotation may instead be indirectly addressed by comparing ζ test results for anterior and posterior mandibular heights.

The most anterior and posterior corpus heights in this study are at the symphysis (X711) and ramus (X607), respectively. Across all possible comparisons (Table 4.6) the mean ζ is greater for ramus height than symphysis height (Fig. 5.3), consistent with greater rotation in *A. robustus*. Across ontogeny (i.e. periods 1-4 or 1-5), average ζ for each trait is very different. The ζ test indicates that symphysis height increases more in *A. robustus* than humans from period 1-5 (ζ =0.215, P ζ _{s0}= 0.056, n=107). This period could not be assessed for ramus height, but it can in period 1-4, when mean ζ =0.628 (P ζ _{s0}=0.007, n=285). ζ for ramus height is much greater and significantly higher than 0 compared with ζ for symphysis height, consistent with a larger growth rotation in *A. robustus*. Unfortunately, poor fossil preservation means that these two measurements

cannot be fully compared either within individuals or across ontogeny (Table 4.7), making it difficult to define when and how exactly this difference is brought about.



Figure 5.3. All ζ statistics for height at the mandibular symphysis and at the ramus. Boxes include the 50% quartiles, dots are outliers, and solid black lines are medians.

The ζ test does provide evidence to suggest that the greater rotation in *A*. robustus may occur mostly during period 3-4. Early in ontogeny, ζ statistics for both ramus and symphysis height are very similar and not significantly greater than 0 (symphysis height in period 1-2: mean ζ =0.127, P ζ _{s0}==0.171; ramus height in period 1-3: mean ζ =0.177, P ζ _{s0}=0.157). Species differences, if they truly exist, are proportional in the front and back of the mandible, implying no difference between species in the amount of rotation. The amount of relative size increase anteriorly at the symphysis in each species is essentially the same from eruption stage 2-5 (mean ζ =-0.004, P ζ _{s0}=0.527, n=58). Ramus height is not preserved in exactly the same eruption stages, but during period 3-4 ramus height increases much more in *A*. *robustus* than humans on average (mean ζ =0.276, P ζ _{s0}=0.083,

n=216). Taken together, this suggests that anterior and posterior facial growth rates are similar in each species prior to eruption stage 4, but between period 3-4 growth of the *A*. *robustus* posterior face is relatively faster than the in front compared with humans. In other words, the rate of facial rotation may be similar between species until period 3-4 when *A*. *robustus*' face rotates faster. The inferred rotation pattern of *A*. *robustus* differs from the human pattern of rotation, which is usually greatest prior to the eruption of permanent teeth (i.e. period 1-3; Karlsen, 1997; Wang et al. 2009).

The fragmentary fossil record prevents a more definitive and rigorous study of rotation. The ζ test lends support, albeit somewhat equivocal, to the hypothesis of greater facial rotation in *A. robustus* than humans during postnatal growth. Species differences in facial height growth across ontogeny probably contribute to this difference, but there is modest evidence that the bulk of the extra rotation in *A. robustus* occurs during period 3-4, along with the greater overall mandibular growth that occurs in this species.

Bone Remodeling Fields

The proximate mechanisms of mandibular size and shape change are the differential deposition and resorption of bone on its surfaces (Enlow and Harris, 1964). The distribution of these depository and resorptive fields on the faces of different species are consistent with the anatomical differences between them (O'Higgins and Jones, 1998). For instance, the development of humans' pronounced chin is reflected in bone deposition around the symphysis inferiorly but resorption superiorly around the anterior alveolar margin. In prognathic taxa such as *A. africanus*, chimpanzees and macaques, the entire surface is depository (Johnson et al., 1976; Bromage, 1989; Enlow and Hans,

1996). Fields of bony remodeling thus reflects the different vectors of growth in these different facial types: forward in prognathic species and more downward and forward in humans.

Results of the present study are consistent with Bromage's (1989) interpretations of *A. robustus* facial growth based on remodeling fields. Bromage interpreted the resorptive surface of the maxillary clivus and anterolateral mandibular corpus of *A. robustus* to indicate a facial rotation and downward component of facial growth in this species, similar to modern humans but unlike more prognathic taxa. Consistent with this interpretation, the ζ tests suggest *A. robustus*' mandible and face may have experienced a greater anterior rotation than modern humans. Similarly, humans' slender mandibular corpora have a large resorptive surface medially extending from the submandibular fossa beneath the premolars to the ramus posteriorly (Enlow and Hans 1996). This region in *A. robustus* is depository (Bromage 1989), consistent with this species' greater increases in corpus breadth compared with modern humans detected by the ζ test.

Even though Bromage's (1989) interpretations were borne out by the ζ test, the distribution of remodeling fields alone cannot account for the species differences in the magnitude of size changes detected by the ζ test (e.g. McCollum, 2008). As Bromage (1989) recognized, the *rate* of deposition and resorption within these fields is another important variable influencing bony size and shape change. The mandibular corpus, like the shafts of long bones, grows by bone deposition on the external (periosteal) surface and resorption of the internal (endosteal) surface. A reasonable hypothesis would be that *A. robustus* and humans differ in the rate at which these surfaces deposit and resorb bone (cf. the discussion of jaw biomechanical development above).

To test this and other hypotheses about the morphogenetic mechanisms underlying *A. robustus* mandibular growth, Synchrotron micro-computed tomography may eventually prove a useful way of investigating fossil histology beneath the bony surface. This is currently the highest-resolution tomographic technique, and has been used to study the histology in recent and fossil teeth (Tafforeau et al., 2012), as well as histology of recent human bone (Cooper et al., 2011; Pacureanu et al., 2012). The technology is quite expensive, and so has yet to be used extensively on bony fossil material (but see Mazurier et al. 2006; Carlson et al. 2011). This imaging technology has yet to be used to examine the pattern and process of bone growth and remodeling in fossils similar to the classic studies of Enlow and others using ground sections (Enlow, 1962; Enlow and Harris, 1964; Enlow et al., 1971; but see Houssaye et al., 2010 for application to living snakes).

Summary of Evo-Devo

Differences between humans and *A. robustus* in the growth of individual mandibular measurements seem ultimately reflect dietary differences between these species. On the one hand, corpus breadth increases more, and more rapidly, in *A. robustus*. This theoretically increases the mechanical strength of the corpus and symphysis to withstand high forces generated during chewing. This is a hypothesis that merits testing by analyzing cortical bone thickness and distribution about the corpus in developing the jaws of *A. robustus* and other taxa (cf. Daegling and Grine, 1991). On the other hand, length of the posterior corpus increases more in *A. robustus* than humans, concomitant with the former's larger (e.g. longer) teeth. It is possible that differences in

corpus breadth growth between species also reflect the development of differently sized postcanine teeth breadth, although evidence here is equivocal. Tooth size differences cannot account for species differences in breadth growth at the symphysis (Table 4.7), suggesting a biomechanical explanation is more viable for this region (White, 1977).

The test detected a greater degree of facial rotation during growth, perhaps chiefly during period 3-4, and greater rates of corpus thickening in *A. robustus*, as predicted from models of facial morphogenesis (Bromage 1989). These results should be interpreted with a degree of caution, since the inferred mandibular rotation was based on comparing ζ statistics for different corpus heights, rather than testing a true measure of 'rotatedness' in individuals. While the broadening of the *A. robustus* mandibular corpus throughout ontogeny is consistent the species' pattern of mandibular remodeling fields, the severity of this broadening is not necessarily predicted on this basis (McCollum, 2008).

The present research indicates an important role for postnatal ontogeny in creating the morphological differences between *A. robustus* and humans, and raises a number of research questions that deserve treatment in the future. Future studies should employ state of the art computed tomography (CT) imaging techniques to better quantify patterns and rates of bone growth in these hominids. CT data have been extensively used to analyze bony macrostructure for decades, but the technology is now capable of viewing microstructure as well. These new sources of data, combined with novel analyses such as the ζ test, open the door to countless new and interesting questions in anthropology and evolutionary and developmental biology.

5.3 Hominid Life History

Chapter 2.2 presented the null hypothesis that aspects of modern human life history, namely patterns of growth, were identical between humans and *A. robustus*, which would suggest that the human pattern was established by the early Pleistocene. This hypothesis predicts humans and *A. robustus* jaws would follow the same pattern of overall size change between eruption stages, but the results presented in Chapter 4.2 rejected this hypothesis. This section discusses the implications of this result for hominid life history. I begin by addressing the extrapolation of inferences from the mandible to bodily growth. I then discuss this study's findings in terms of whether *A. robustus* may have experienced the unique aspects of human life history stages, including childhood, middle childhood and adolescence. Finally, I treat the proposal that *A. robustus* males continued growth into adulthood, in terms of the outcome of the ζ test as well as data from adult *A. robustus* not presented above.

Results of the ζ test corroborate the visual assessment that the *A. robustus* mandible undergoes much greater increase in overall size prior to the occlusion of the third molar compared with modern humans. This size change is concentrated in two periods, first prior to the full occlusion of the first molar and then between the eruptions of the first and second molars. Before situating these results in the context of hominid life history, it should be noted that one must be careful extrapolating results from *mandibular* size to inferences about *body* size.

A working assumption of the study is that mandibular growth reflects the same processes as the rest of the body, and several studies have shown this to be the case (Bergersen, 1972; Lewis et al. 1985; Franchi et al., 2001, 2007; Antón and Leigh, 2003). Recently, Coquerelle and colleagues (2010b, 2011) examined development of sexual

dimorphism in a cross-sectional sample of recent human mandibles. While their study did not directly compare mandibular and bodily growth, their multivariate mandibular size and shape growth curves (2011: Fig. 3a-b) are similar to published growth curves for mass and stature in living humans (e.g. Bogin 2009, Walker et al. 2006): rates of change (for size and shape) decelerate until around five years of age, level out until puberty (around age 10-12) when there is a short spurt after which growth decelerates or ceases. Sex differences in mandibular growth also mirrored those in bodily growth.

Chapter 4.1 showed that the pairwise size metric used in this study is also consistent with patterns of human body size change (Fig. 4.2). However, it was also shown that the *A. robustus* mandible becomes much larger than humans despite starting at similar sizes in stage 1. Body size estimates for *A. robustus* adults, based on isolated postcrania, range from 30-40 kg (McHenry and Coffing, 2000; Pickering et al., 2012). Most human populations are larger than this (Eveleth and Tanner, 1988), so it cannot strictly be that mandibular growth reflects body growth the exact same way in these species (Wood and Aiello, 1998). In this regard, it is important that the ζ statistic measures *relative* change in size between dental eruptions, minimizing the effects of the absolute size differences between species at later stages of eruption.

With this caveat in mind, the following discussion interprets the results of the analysis in terms of humans' life history stages, proxied by dental eruption (Bogin 1999, Hochberg 2012). The discussion focuses chiefly on the results of the ζ test for the pairwise size metric between successive eruption stages as a proxy for body size since this is an important life history variable (Robson and Wood 2008). Although the

dichotomization of 'ape- or human-like' was decried above, these models are contrasted here to illustrate the unique aspects of humans' pattern of growth.

Infancy or Childhood?

The period between eruption stages 1-3 roughly corresponds with the infant period of chimpanzees, in which the subadult feeds mostly by nursing (Zihlman et al., 2007). Humans, on the other hand, tend to wean from nursing long before the first molar begins erupting (i.e. early in eruption stage 1). Because of this unique dependency, the time from weaning to first molar eruption in humans is considered an evolutionarily novel growth phase, childhood, and present only in humans (Bogin 1997, 1999; Hochberg 2012). Humans rapidly decelerate bodily growth rates at this time, whereas (captive) chimpanzee growth accelerates in this period (Walker et al., 2006b). Mandibular size change in dental eruption periods 1-2 and 2-3, then, could be thought of as representing earlier and later phases of childhood growth, respectively, in humans, but representing infant growth in other apes.

During period 1-3, the *A. robustus* mandible undergoes greater size change than humans' on average, and there is a fairly low probability of sampling a pair of period 1-3 humans as different in size as pairs of *A. robustus* (mean ζ =0.126, P ζ =0.158). However, one cannot reject the null hypothesis of indistinguishable growth patterns between period 1-3 overall. Looking at periods 1-2 and 2-3 specifically make it difficult to interpret this weak rejection.

In period 1-2, mean ζ for the pairwise size metric is 0.149 (P $\zeta_{\leq 0}$ =0.091), but during period 2-3, the species' patterns of size change are less distinct (Table 4.4). If

these differences in size increase reflect bodily growth rate differences between species, this could argue against *A. robustus* having slow, human-like growth rates during childhood. The high average ζ statistic in period 1-2 followed by near-zero average ζ in period 2-3 could suggest either a delayed deceleration in body growth compared to humans, or possibly an acceleration in period 1-2 similar to chimpanzees. Along these lines, Bastir and Rosas (2004: 506) using geometric morphometric methods found that chimpanzees' mandibular size increases more rapidly than humans' both prior to the eruption of M₁ and thence to M₂ emergence (cf. periods 1-3 and 3-4 in this study).

Alternatively, the differences between species' mandibular growth in period 1-3 may be less directly related to body size, as discussed earlier in section 5.1. The high average ζ statistics during period 1-2 are driven by the significantly positive D values for measures of corpus breadth (Table 4.7). As this is the time when the species' megadont M₂ and P₄ crowns begin forming, the severe mandibular size change in *A. robustus* may simply be a reflection of having large postcanine teeth. An additional possibility, relevant to life history, is that these young mandibles were capable of, and used for, processing an adult-like diet. If this is the case, this may be a marker of relatively early weaning in this species, as described above.

Early weaning is an important part of the human adaptive strategy, as it allows females to reduce the time that ovulation is suppressed due to lactation, thereby decreasing the amount of time between births (Lovejoy, 1981; Kaplan et al., 2000). Human children are not fully equipped handle an adult diet, given their small deciduous teeth and digestive tracts, and so weaned children are highly reliant on older individuals to obtain and prepare their food for them (Sellen, 2007). If the robusticity of stage 2 *A*.

robustus mandibles provides comparable bending and torsion strength to adults, as suggested in section 5.2, this could indicate *A*. *robustus* weanlings were more independent foragers than humans of comparable dental age.

This study's implications for whether *A. robustus* experienced a human-like childhood or ape-like later infancy depend on a number of things. Depending on how mandibular growth tracks that of the body, ζ statistics between period 1-2 and 2-3 could imply more rapid growth than in humans before permanent teeth begin erupting in period 2-3. Alternatively, because ζ statistics in period 1-2 reflect differences confined largely to corpus breadth growth, this could imply that overall (i.e. body) growth and relative age at weaning are similar between these hominids. These alternatives cannot be distinguished in the present study, nor are they mutually exclusive (cf. section 5.2 above).

Juvenility or Middle Childhood?

The period between eruption stages 3-4 corresponds with the juvenile period of humans and other primates (Bogin 1999). This period lasts from the time of M_1 eruption until puberty, roughly from 4-8 years of age in wild chimpanzees (Pusey 1990; Zihlman et al. 2004, 2007), and around 6-11 years in most human populations (Bogin 2009; Walker et al. 2006). During this time chimpanzees continue high bodily growth rates from the preceding infancy period (Pusey et al. 2005; Walker et al. 2006).

In humans, however, bodily growth rates usually descend to their lowest levels of all ontogeny (Walker et al. 2006a-b, Hochberg 2012). This is also a time critical period of social development in humans, known to psychologists as "middle childhood" (Mann and Monge 2010; Campbell, 2006, 2011). One developmental marker of the beginning of

middle childhood, seen only in humans and other hominids, is a 'quiescent period,' a delay of a few years between the eruption of the permanent incisors on the one hand, and the canines, premolars and M₂ on the other (Liversidge 2003; Mann and Monge 2010). Slow growth during the juvenile/childhood phase is critical to the human adaptive strategy because it reduces the energetic demands on the older individuals who must help feed multiple dependent offspring (Gurven and Walker 2006). This relative (but not total) freedom from foraging allows human youngsters additional time to learn social behaviors and cultural mores (Mann and Monge 2010; Hochberg 2012). The hallmark of human juvenility, or middle childhood, is therefore the preferential devotion of energies to social and cognitive development over body size growth compared with apes.

The high, significantly positive D distributions for both the pairwise size metric and individual traits between eruption stages 3-4 suggest that *A. robustus* likely underwent greater size increase than humans – a juvenile and not middle childhood stage. This is the only period in which the ζ test detected a significant difference in overall size between species (P $\zeta_{\leq 0}$ =0.019), and reflects size change throughout the mandible (Table 4.7). Period 3-4 stands out as the only successive eruption period in which *A. robustus* ' corpus height increases significantly more than humans'.

Mann (1975) and others (Mann et al. 1990; Mann and Monge 2010) have argued that all hominid juveniles to date show a human-like quiescent period of no dental eruptions, indicative of a middle childhood. If so, then the drastic difference in mandibular, and possibly body, size increase between humans and *A. robustus* would imply that the nature of middle childhood has changed since the early Pleistocene. Slow growth of human children during this time has been viewed as an adaptation to reduce

energetic requirements of multiple dependent offspring (Gurven and Walker 2006; Bogin 2009). The present results would therefore suggest *A. robustus* juveniles would have been more independent than human juveniles, consistent with some interpretations of mandibular growth during childhood described above.

It should be noted that this result relies on a sole stage 3 *A. robustus* (SK 63). This specimen is more similar in size and morphology to SK 61 and 62 (stage 2) than to stage 4 fossils (Appendix II). If SK 63 is on the small end of the true range of *A. robustus* variation at this stage of tooth eruption, then the difference from humans in overall size change may not be so great as indicated by this study.

In sum, results of the ζ test for pairwise size during period 3-4 are consistent with relatively rapid, ape-like growth during the juvenile period in *A. robustus*. While rapid juvenile growth for early hominids was suggested on the basis of dental development as it was understood over 20 years ago (e.g. Bromage and Dean 1985; but see Mann et al. 1990 for the problem with their inference), the present study is the first to strongly indicate this for an australopithecine skeletal element. Nevertheless, further study of the relationship between dental development (within the crypt) and mandibular growth, especially in megadont taxa like *A. robustus*, may well show this difference to reflect tooth size and not bodily growth (see section 5.2 above). Moreover, given the stronger size similarity between SK 63 and stage 2 *A. robustus*, if additional *A. robustus* mandibles in eruption stage 3 (M₁ recently occluded), this would reduce the apparent difference between humans and *A. robustus* in growth in this period.

Adolescence

The final period of subadult growth examined in the present study, early adolescence, is represented by eruption stages 4-5 (prior to full occlusion of M₃). This period lasts from around 8-10 years in wild chimpanzees (Pusey 1990; Zihlman et al. 2007), but around 11-15 in recent, well-nourished human populations. Humans are unique among primates in that a growth spurt in body size is delayed until around age 10-12 and 12-14 years in girls and boys, respectively (Bogin 1999; Leigh 2001). Moreover, while body mass spurts are known to occur in other species (viz. males; Leigh 1996), none experiences the marked increase in linear skeletal dimensions seen in human limbs (Bogin 1999; Hamada and Udono 2002) and jaws (Bergersen 1972; Anton and Leigh 2003; Franchi et al. 2007; Coquerelle et al. 2011).

Whereas period 3-4 saw the greatest increase in overall size of *A. robustus* mandibles, there is little to no size increase in most mandibular measurements in period 4-5 (Table 4.7). In contrast, corpus height at several positions undergoes a non-trivially, and occasionally significantly, greater size change in humans than in *A. robustus*. This pattern of significantly positive ζ statistics in period 3-4 and mildly negative ζ statistics in period 4-5, is very similar to the difference in mandibular size change between humans and chimpanzees detected using geometric morphometrics (Bastir and Rosas, 2004). As this mandibular pattern is similar to published difference in these species' body mass growth curves (Walker et al., 2006) the results of the ζ test are consistent with expectations of an adolescent spurt in humans but not *A. robustus*. However, this may be due at least in part to the fact that only a single fossil (SKW 5) is represented by this stage, so it is possible that an expanded fossil sample would give a different impression.

Beyond subadult growth

The only other study of growth in *A. robustus* compared age-related size variation in adult maxillae and mandibles from Swartkrans, Drimolen and Kromdraai (Lockwood et al., 2007). In that study, adults were ranked by age based on tooth wear, and by size based largely on visual appraisal. Larger maxillae tended to be older but there were a few old and small specimens, whereas mandibles tended to increase in size across all age groups. Lockwood and colleagues interpreted the pattern of maxillary variation to reflect bimaturism, in which females achieve adult size prior to adulthood while whereas males continue growth into adulthood, resulting in increasing levels of sexual dimorphism. Bimaturism is known to occur for body mass among many living primates including nonhuman great apes (gorillas and bonobos: Leigh and Shea, 1996), and has concomitantly been found in the facial skeleton of orangutans and gorillas (Randall, 1943; Shea, 1985; Masterson and Leutenegger, 1992; McCollum and Ward, 1997; Daegling, 1996; Schaefer et al., 2004; Hens, 2005; Cobb and Higgins, 2007).

The present study cannot refute or confirm the claim of bimaturism in *A. robustus*, since adults were not studied and individuals' sexes are unknown in the fossil and human samples. The sex of the oldest subadult (and therefore most pertinent to this question) analyzed here, SKW 5, is fairly ambiguous. This specimen is among the smallest of the otherwise fully adult mandibles (as judged by M₃ occlusion) analyzed by Lockwood and colleagues (2007), which might support the argument of bimaturism if the specimen is male. To this end, its corpus is one of the broadest in the Swartkrans *A. robustus* sample. However, its height is among the shortest (cf. Grine and Daegling 1993), even compared with younger (stage 4) specimens (Appendix 1). Indeed, the specimen may well be

female: it is even (visually) smaller than DNH 7 from Drimolen, the very small fossil off which Lockwood and colleagues base their claims of sex in *A. robustus*. SKW 5's dental dimensions also fall in the lower end of the entire range of *A. robustus* from the Swartkrans, Kromdraai and Drimolen sites (Moggi-Cecchi et al. 2010). In short, the claim of bimaturism can only be tested against a larger fossil sample with more secure sex identification than are available at present.

The pattern of subadult growth in *A. robustus* implied by the ζ test is not inconsistent with the hypothesis of bimaturism, whereas extended male growth would seem less likely if the *A. robustus* pattern of size change were similar/identical to humans' since humans do no show as marked bimaturism as gorillas or orangutans. Metric data from adult *A. robustus* (not presented above) also do not refute the hypothesis. The lingual tuberosity is an important locus of mandibular growth, elongating the corpus anteriorly by deposition on its posterior surface (Enlow and Harris, 1964; Enlow and Hans 1996). Linear distances from points along the alveolar margin to the distal-lingual margin of the lingual tuberosity are around 5.0 mm shorter on SKW 5 than measurable adults with M₃ erupted (SK 23, SK 34 an SK 12). Within the adults, however, these increase only modestly with age as assessed by tooth wear (<3.0 mm).

There is equivocal and modest support for the hypothesis of bimaturism in the mandible of *A. robustus*. An alternative hypothesis to bimaturism is that *A. robustus* adult facial variation is the result of natural selection for larger adults to live longer than smaller ones. Distinguishing these hypotheses is not easy given fossil samples, especially because these hypotheses are not necessarily mutually exclusive. The evidence presented here and by Lockwood and others (2007) lend modest or equivocal support to the claim

of bimaturism, and the question should be explored further by comparing *A. robustus* with known bimaturing species such as mandrills, orangutans and gorillas, as well as with other species showing different degrees of adult growth.

Summary of Life History

Investigating the antiquity of the human life history strategy was a major impetus for the analysis of mandibular growth in *Australopithecus robustus*. One of the most unique aspects of this strategy is a long period of bodily growth with non-linear rates: in most populations, bodily growth slows during childhood, rises briefly around the time of puberty and then comes to a halt; this pattern is seen in no other animal (Eveleth and Tanner 1991; Bogin 1999; Walker et al. 2006). In contrast, both wild (Pusey et al., 2005) and captive (Walker et al., 2006a) chimpanzees and other African apes (Leigh and Shea, 1996) have relatively high growth rates throughout ontogeny.

Life history inferences are cautiously made on the basis of the size and shape of the mandible, chiefly as measured by the pairwise size metric (Chapter 3). Previous analyses of human mandibular growth (Bergersen 1972; Franchi et al. 2007; Coquerelle et al. 2011) suggest the size and shape of the mandible changes in a manner similar to the body, and the pairwise size metric used in the present study also patterns by eruption stage as expected based on human bodily growth. Although the *A. robustus* mandible becomes much larger than humans', the effect of this on the analysis is minimized because the ζ statistic, as the difference between two ratios, is a measure of relative and not absolute size change.

Important for life history, *A. robustus*' mandible experiences its most dramatic size change during period 3-4, whereas the greater size change occurs in period 4-5 for the human sample. This does not necessarily preclude the presence of a human-like 'middle-childhood' as assessed by dental development (Mann and Monge 2010), but it does imply an important difference between humans and *A. robustus* at this stage of maturation. If *A. robustus* were as dependent as modern humans of comparable age, such rapid growth would entail a larger amount of energy, obtained from parents and other caretakers (Gurven and Walker, 2006). Rather, juvenile production (food acquisition) in *A. robustus* may have been high as in other animals and unlike humans (Kaplan et al. 2000). Concomitant with a relatively early, dramatic increase in size, a human-like adolescent growth spurt probably was not present in *A. robustus*. Finally, results of the present study could be consistent with early weaning (Aiello et al., 1991) and extended male growth (Lockwood et al., 2007), but they certainly do not confirm these claims.

Summary of A. robustus mandibular growth

This chapter reviewed the implications of this study for paleoanthropology. The first contribution is a novel statistical procedure, the ζ test, to assess patterns of variation in skeletal samples otherwise less amenable to study. The test makes few assumptions, and performs as expected and intended: the test did not distinguish the human pattern from itself, yet it was able to detect a number of ways that the *A. robustus* pattern differed from humans. Nevertheless, some aspects of cross-sectional sampling cannot be overcome by the test, especially when individual sex and age cannot be known.

The differences in patterns of mandibular growth detected by the ζ test further contribute to our understanding of hominid evolutionary developmental biology and life history. For evo-devo, results show how *A. robustus*' unique adult mandibular anatomy is attained through growth. The *A. robustus* mandible rapidly becomes broad early in ontogeny, prior to the eruption of the first molar. This change could be due to the development of large tooth crowns within the mandibular corpus, and/or to prepare the infant for the mechanical stresses resulting from of a normal diet following nursing. Results were also consistent with predictions of more extreme facial rotation (due to "posterior facial hypertrophy"; McCollum 1997: 380), and corpus thickening based on observed patterns of periosteal bone deposition and resorption (Bromage 1989; McCollum 1999).

Results also provide the first clear skeletal evidence suggesting that the human pattern of growth, important to our life history strategy, was not present in the early Pleistocene hominid *A. robustus*. In contrast with the human pattern of growth, *A. robustus*' mandible grew relatively more rapidly during the juvenile period (stages 3-4) rather than delaying a growth spurt until adolescence (stages 4-5). The pattern of differences in size change between humans and *A. robustus* is similar to that found comparing mandibular growth in humans and living chimpanzees (Bastir and Rosas 2004), who are known to differ from humans in body mass growth the same way (Pusey et a. 2005; Walker et al. 2006b).

The present study therefore provides important evidence about both life history and mandibular (size and shape) growth in *A. robustus*. At the same time, the test raises a number of questions about the nature of these patterns that could very feasibly be

followed up. The evidence for the possibility of early weaning in *A. robustus* (Aiello et al. 1991) can be further sought in dental microwear studies of the Swartkrans juveniles, as well as in the bone histology and biomechanics of their jaws. And to repeat the adage of paleo-anthropology: additional fossils sample will be crucial to definitively show whether the patterns of growth detected by the present study are correct.

Chapter 6

A Look Back and Ahead

This dissertation provides new insights into the biology of *Australopithecus robustus*. In light of this study's success in elucidating the patterns of size and shape change in the *A. robustus* mandible, with implications for hominid development and life history, a number of questions remain or arise from this work. The purpose of this chapter is therefore to review the rationale and results of the present investigation into patterns of growth. I conclude by suggesting future research directions stemming from this work.

Chapter 1 introduced the species under consideration, *A. robustus*. Adult jaws of this species are large and robust, and while evidence is equivocal as to this species' exact diet, it is clear that its mandibular morphology was capable of withstanding the stress and strains of powerful and prolonged chewing. This analysis thus asked how and when this anatomy was created during subadult growth. This hardy anatomy also ensured relatively good preservation, creating the best hominid sample on which to conduct an ontogenetic study. This sample is therefore uniquely poised among early Pleistocene hominid skeletal elements, to test the hypothesis that patterns of size change cannot be distinguished from humans'.

Chapter 2 discussed the motivations for statistically comparing size change in the mandibles of humans and *A. robustus*. The *A. robustus* sample is better suited than any other early hominid sample to address the evo-devo question of whether species

differences come about. This is often framed in terms of whether differences are present early (i.e. prenatally), and whether patterns of shape change are basically the same between species. Although this question has been asked of the earlier hominid *Australopithecus africanus* (e.g. Ackermann and Krovitz, 2002; McNulty et al., 2006), these studies have relied on a single subadult fossil (the Taung skull, in eruption stage 3) and 1-3 adults of unknown age. Such a strategy, focusing on the most complete specimens, underestimates static variation within age groups and necessarily ignores ontogenetic variation between age groups.

Important for life history, this study was also better suited to ask whether the unique human patterns of size change (i.e. growth rates) were present in an early hominid. Our pattern of growth, characterized by a childhood/juvenile period slow growth followed by an adolescent spurt, is essential for both providing youth enough time to learn necessary social and linguistic skills, while also minimizing energy requirements. Most studies of early hominid life history have relied on extrapolating inferences from *dental* development to the rest of the skeleton, which is problematic conceptually (i.e. ape and human body growth rates differ markedly in spite of similar tooth crown formation times). Those studies that did employ skeletal elements (viz. faces and jaws) either relied on a sole fossil subadult, and/or on a dichotomous 'ape- or human-like' framework that hindered the interpretation of results.

Utilizing a the relatively large *A. robustus* ontogenetic series of mandibles and a new statistical test, this study was designed to avoid many of the difficulties and issues faced by previous studies of hominid ontogeny. Testing the null hypothesis that *A. robustus* could not be distinguished from humans in terms of size change between dental

eruptions, this analysis was devised to have more easily interpretable results for the evodevo and life history of an early hominid species. For evo-devo, this hypothesis implied that anatomical differences between humans and *A. robustus* were established early in development, and that species followed the same pattern of size/shape change. For life history, the null hypothesis implied that the human growth pattern, and possibly other aspects of life history, would be shared with *A. robustus* as a homology.

Chapter 3 described the materials and methods used to test this hypothesis. Fossil samples have prevented rigorous ontogenetic analysis because of small sizes and copious missing data. The *A. robustus* mandibular series from the site of Swartkrans suffers least from these challenges of all early hominid ontogenetic series. The pattern of *A. robustus* ontogenetic variation was sought in a sample of human mandibles from the prehistoric American site of Libben (800-1100 CE; Lovejoy et al., 1977). This sample was chosen because it is among the largest skeletal 'populations,' and appears to have come from an egalitarian hunter-gatherer population (Meindl et al., 2008). Mandibles of humans and *A. robustus* were assigned to one of five dental eruption stages spanning from the completion of the deciduous dentition to just before the occlusion of M₃, as eruption stages are comparable between species and correlated with life history events. Data for this analysis were 29 linear measurements and a pairwise size metric (the geometric mean) computed from these variables, allowing all measurable fossils to be included in the study.

Even though the *A. robustus* mandibular series is relatively complete for a fossil sample, it cannot be analyzed by traditional analytical methods, such as geometric morphometrics and other multivariate statistical techniques. These methods require both

complete specimens, and make assumptions about sample variation that fossil assemblages do not meet. Chapter 3 therefore presented the ζ test, which compares two samples for the relative amount of change in size that occurs between dental eruption stages. The ζ test resamples *pairs* of individuals in each sample, and calculates a test statistic as the difference between the *A. robustus* and human growth ratios: $\zeta = [(\mathbf{R}_{OLDER} / \mathbf{R}_{YOUNGER}) - (\mathbf{H}_{OLDER} / \mathbf{H}_{YOUNGER})]$. This randomization procedure was run an excessive number of iterations, so that all possible comparisons were made and redundant observations could be removed (i.e. exact randomization). The null hypothesis predicts that resampled ζ statistics will equal zero (no difference between species) on average.

The ζ test was run for the pairwise size metric, based on whatever traits are shared between resampled pairs of specimens, and for every individual trait. The ζ test for the pairwise size metric is important logistically for comparing fragmentary fossils and theoretically for assessing life history, while the ζ test for individual traits is important for analyzing shape change vis-à-vis evo-devo. By describing growth in terms of pairwise comparisons, the ζ test is uniquely able to include all relevant fossils, not just the most complete ones. Rather than specifying any specific pattern or model of growth (e.g. 'apeor human-like'), the test simply asks whether the fossil pattern of variation is possible or likely to be sampled in a larger human reference.

Chapter 4 demonstrated that the ζ test is appropriate for, and capable of, comparing samples, and presented the results of the test comparing patterns of mandibular size change in humans and *A. robustus*. The first section demonstrated that both the pairwise size metric and the ζ test perform as expected. The size metric, which was calculated based on only the measurements shared between pairs of specimens,

distinguished human and *A. robustus* sizes across ontogeny, according well with visual appraisal of each species' mandibles. Importantly, the pattern of change in the human 'sizes' appeared to roughly mirror our pattern of bodily growth; stage 4 human sizes tended to be a little smaller than those in stage 3, an artifact of the cross-sectional sample, but this effect was minimized with a fix to the calculation of the ζ statistic preventing 'negative growth.'

Next, the ζ test was validated by comparing the human sample with itself, to assess whether the test would detect a difference when none existed (e.g. Type I statistical error). The test failed to distinguish the human sample from itself, as resampled $\zeta \Box$ tatistics equaled 0 on average. However, there was a wide range of ζ values in the test validation, underscoring the effects of natural variation within a species on a crosssectional sample. Preservation and the difference in eruption status between resampled individuals have a negligible influence on the value of the ζ statistic.

The second section of Chapter 4 presented the results of the ζ test comparing human and *A. robustus* mandibles for the pairwise size metric. There were two successive eruption periods in which the amount of size change was notably different between species, ζ being positive in each case (i.e. *A. robustus* grew more). First, *A. robustus* size was found to increase more during period 1-2, with only a 10% chance of sampling such size change in the human sample. Second and more notably, during period 3-4 there was only a 1% chance of randomly sampling a human pair so different in size as comparably aged *A. robustus*. In the other two successive periods (2-3 and 4-5), the average ζ was slightly negative (i.e. humans grew more) but not significantly different from 0. The two periods of greater size change in *A. robustus* resulted in *A. robustus* undergoing much

more size change, becoming much larger, than humans between the first and last eruption stages. In contrast to the validation analysis comparing humans with humans, resampled metadata (preservation, k, and difference in eruption stage, da) had significant rank correlations with the value of ζ in the interspecific comparison. However, these correlations were low and explained little variance in ζ . Rather, the correlations seem simply to reflect the fact that the *A. robustus* mandible becomes much larger than humans.

As shown in the third section of Chapter 4, results of the ζ test for individual traits were similar to results for the pairwise size metric. Most traits tend to undergo greater size change in *A. robustus*, though none of these was statistically significant. However, between successive eruption stages, *A. robustus*' anterior corpus and symphysis become significantly broader than humans', driving the positive ζ for the pairwise size metric in this period (growth changes to the anterior corpus were also described qualitatively at the end of the section). In period 3-4, it was shown that most aspects of the mandible undergo greater size change in *A. robustus*, a period of overall size change compared to the change more in shape during period 1-2.

The ζ test was therefore able to detect important growth differences between humans and *A*. robustus, and Chapter 5 reviewed the implications for paleoanthropology. First, this study contributes a novel statistical procedure, the ζ test, for analyzing fossil samples. The test made few assumptions about the nature of each cross-sectional sample, which do not appear to have strongly influenced the results. However, the single fossils in each dental stage 3 and 5 warrant caution in making two of the assumptions: first, that cross-sectional samples accurately depict characteristic longitudinal growth of individuals in each species, and second that sex variation in growth has a negligible influence on the

outcome of the ζ test. Each of the fossils in these two stages is of comparable overall size and anatomy to fossils in the eruption stages immediately previous. Future fossil discoveries are necessary to determine whether these two fossils are rather small for their age, and whether this would be due to sex.

The second section of Chapter 5 discussed the implications of the analysis for evo-devo. Some aspects of species' morphology, such as the topography of the mandibular symphysis, are distinct between species in the earliest dental stage. Although human and *A. robustus* mandibles are of comparable size when only the deciduous dentition is erupted, the *A. robustus* jaw rapidly becomes broad early in ontogeny, prior to the eruption of the first molar. Results of this study therefore indicate that neither of the idealized trajectories depicted in Figure 2.1 fully describes the growth differences between human and *A. robustus* mandibles. Rather, some species differences were probably established at younger ages than those analyzed here, while other differences came about or were amplified later in ontogeny. *A. robustus*' characteristic corpus robustness and relatively tall ramus (and 'rotated' jaw), for instance, arise at different times later in postnatal growth.

The most notable difference in shape growth between species relates to *A*. *robustus*' great corpus broadening between eruption stages 1-2, which could reflect several biological phenomena, not necessarily mutually exclusive. First, this could relate to the development of the species' megadont postcanine teeth. Species differences in corpus length growth almost certainly reflected different tooth sizes, but this is less clear for corpus breadth growth. Second, the anatomy attained by eruption stage 2 may reflect a biomechanical response to processing the species' diet. In turn, this corpus broadening

could be taken to indicate early weaning in this species. It was beyond the scope of this analysis to examine all these possibilities, but future work should investigate these further (see below).

The last section of Chapter 5 discussed the implications of this study's results for life history. Results also provide the first statistical, skeletal evidence that humans' characteristic pattern of growth was not present in an early Pleistocene hominid. In contrast to the greater size change in *A. robustus* in period 1-2 which was due entirely to corpus breadth growth, greater growth is spread across the entire *A. robustus* mandible during period 3-4, suggesting this is truly a major size and not shape change. In humans, period 3-4 is described as 'middle childhood' with slow growth. That *A. robustus* grew more during this time suggests that period 3-4 was a juvenile stage rather than a human-like middle childhood (or that its middle childhood was qualitatively very different from humans'), and that this species did not have a delayed adolescent growth spurt similar to humans. However, as noted above, this result rests on two relatively small fossils in each stages 3 and 5.

Future analyses

The present study thus provides some of the clearest evidence for differences in skeletal growth and development between humans and a closely related sister species, raising a number of questions to be addressed by future research. It was noted in Chapter 5 that the present findings may be consistent with the hypothesis of early weaning in *A*. *robustus*, as proposed by Aiello and colleagues (1991) on the basis of deciduous tooth wear. More direct evidence of such a subsistence shift could come from an ontogenetic

analysis of deciduous tooth microwear, as well as longitudinal analyses of the carbon isotopic composition of teeth forming during this period (cf. Sponheimer et al. 2006).

In addition, if *A. robustus*' great corpus breadth increase between eruption stages 1-2 reflected a biomechanical adaptation resulting from early weaning, this hypothesis could only really be tested by knowing the distribution of cortical bone about the mandibular corpus (Daegling 1991, 2007). The internal anatomy of fossil specimens can be obtained from CT scans (e.g. Conroy and Vannier 1991), and so a future study following from this dissertation would be to use CT data to examine the development of corpus strength in *A. robustus*, humans and other apes (cf. Holmes and Ruff, 2010).

Indeed the study of growth patterns more generally could benefit from the use of 'virtual' methods of data collection and analysis (Weber, 2001). In Chapters 4-5, it was demonstrated that at least some of the differences in human and *A. robustus* mandibular growth are probably due to their differently sized molar crowns and roots, but this relationship could not be directly examined in this analysis for lack of access to the internal anatomy of most specimens. CT data, however, would allow a direct examination of the covariation between corpus growth and tooth crown and root development (e.g. Boughner and Dean 2004; Coquerelle et al. 2010b, 2011).

Thus, an intriguing avenue for future research is the potential use of highresolution micro-CT (i.e. synchrotron) to analyze bone histology (Mazurier et al., 2006). The external surfaces of fossils have been examined for the telltale signs of bone deposition and resorption via microscopy (Bromage 1989; McCollum, 2008; Martinez-Maza et al. 2010, 2011; Kraniotioi et al. 2011), but the endosteal surfaces of bones are also important sites of growth. If the bony shape changes of the *A. robustus* mandibular

corpus identified in this study are adaptive responses to loads experienced during, say, weaning, virtual bone histology could help identify the mechanism by which this happens (e.g. rapid periosteal deposition and/or reduced endosteal resorption).

Although synchrotron micro-CT has been able to examine the sub-micron structure of recent bones (Cooper et al., 2011; Pacureanu et al. 2012), the mineralization of individual fossils will always influence the usefulness of radiographic imaging (Mann 1975; Smith and Tafforeau 2008; Smith et al., 2012). Many of the important histological signatures of bone modeling and remodeling identified in ground sections involve subtle differences in mineralization between older versus newer bone (e.g. Enlow, 1962). The fossilization process alters the mineral structure of bones and teeth, which has already raised difficulties in the analysis of dental microstructure. It has yet to be shown that the histological signatures of bony growth can be identified in any fossil using micro-CT.

While much can be learned with new techniques for data collection, this study also shows that novel analytical methods also promise to help extract novel information from fossil samples. The ζ test is a new addition to an already large body of randomization/resampling procedures designed to make statistical inferences about fragmentary samples (e.g. Lockwood, 1999; Braga and Francis, 2003; Lee and Wolpoff, 2003; Van Arsdale, 2006; Gordon et al., 2008; Lague et al., 2008; Royer et al., 2009; DeSilva, 2011). The analysis relied on making pairwise comparisons between individuals, and was shown to be a powerful means of analyzing a fragmentary sample such as the *A. robustus* ontogenetic series.

However, it could be argued that a more accurate randomization procedure would have calculated test statistics based on resampled humans that *exactly* matched the *entire*

fossil sample size and composition. In other words, future analyses could rather resample humans to exactly match the fossil sample: 13 specimens with the exact same number of individuals in each eruption stage as the fossils (Table 3.2). Such an analysis may be difficult to operationalize, as a single test statistic cannot adequately describe the myriad possible comparisons between different eruption stages. However, multiple test statistics, such as one for each comparison between successive eruption stages, could be extracted with each resampling iteration (e.g. as was done with for resampled metadata).

Along these lines, Martín-González and colleagues (2012) recently used resampling to estimate growth curve parameters describing postnatal growth from birth to around 5 years in modern humans and Neandertals. This team used isolated femora and tibiae of Neandertal subadults from throughout Eurasia to estimate each individual's height, and these estimates were plotted against individuals' chronological ages as estimated from human skeletal and dental maturation standards. Because of uncertainty in both the height and age estimates, they resampled all different estimated ages and heights for each individual fossil, and calculated the parameters of the growth curve based on these data. They then compared these with empirical height growth curves in recent human populations. Thus, rather than calculating a single test statistic such as ζ to describe their resampled group, Martín-González et al. estimated several statistics (growth curve parameters).

An important difference between the present study and the clever Neandertalhuman growth comparison (Martín-González et al., 2012) is that these authors used resampling to estimate the Neandertal growth curve, but not to statistically compare the resulting model parameters with human models (visually, the resampled curves were

distinct). Poor and disparate preservation of the *A. robustus* mandibular sample, however, precluded such an across-ontogeny resampling strategy. It was argued to be unnecessary and inappropriate to calculate growth curves for the present study, but it could be very informative to combine the techniques of the present study and those of Martín-González et al. (2012). That is, to develop test statistics that simultaneously describe the entire growth period (e.g. Martín-González et al., 2012) and statistically compare two samples (e.g. this study).

The ζ test as executed here relied on linear measurements to describe size and shape, but the test could be extended to other metrics and data. The present study did not make use of any 'pairwise *shape* metric' as it had for size, because such a metric would not be easily interpretable in this ontogenetic context. Questions about taxonomic or sexrelated variation in fossil samples, however, could be easily treated with the ζ test including a pairwise shape metric specialized for missing data. Such metrics exist for linear measurement data, for instance log s.e._m (Thackeray et al., 1997), STET (Wolpoff and Lee, 2001; Lee and Wolpoff, 2005; Lee 2011), and correlation coefficients (Lague et al., 2008). It could also be useful to develop such a metric for 3D landmark (i.e. geometric morphometric) shape data as well.

Conclusion

This study adopted a null hypothesis testing approach to the study of mandibular growth in *A. robustus*. In addition to demonstrating important differences in mandibular growth between humans and *A. robustus*, this dissertation further demonstrated how randomization procedures can be developed to test new and interesting hypotheses about

old fossils. The ζ test presented here was used to compare patterns of ontogenetic variation, but it can easily be extended to all kinds of other questions and datasets. These methods will only be as effective as samples allow, and some hypotheses remain testable only by the discovery of new fossils. But as demonstrated and discussed in this dissertation, even long-known fossils are waiting for their secrets to be revealed by new methods and technologies.
Appendix I: Raw data plotted against eruption stage





Figure A1.1: Corpus breadth at the position of the P_3 (X223). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.2: Corpus breadth at the position of the $P_{3.4}$ (X225). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.3: Corpus breadth at the position of the P_4 (X227). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.4: Corpus breadth at the position of the P_4 - M_1 (X229). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.5: Corpus breadth at the position of the M_1 (X231). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.6: Corpus breadth at the position of the M_{1-2} (X233). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.7: Corpus breadth at the position of the M_2 (X235). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.8: Corpus breadth at the mandibular symphysis (X709). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.9: Corpus height at the position of the mental foramen (X301). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.10: Corpus height at the position between P_{3-4} (X313). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.11: Corpus height at the position of P_4 (X315). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.12: Corpus height at the position between P_4 - M_1 (X313). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.13: Corpus height at the position at M_1 (X319). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.14: Corpus height at the position between M_{1-2} (X321). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.15: Corpus height at the position between M_2 (X323). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.16: Height from the mental foramen to the corpus base (X331). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.17: Height from the mental foramen to the alveolar margin (X333). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.18: Height of the posteromedial corpus, from the base at the ramus-corpus junction to the alveolar margin (X355). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.19: Corpus height at the mandibular symphysis (X711). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.

Mandibular corpus lengths



Figure A1.20: Corpus length from infradentale to the alveolar septum between P_4 - M_1 (X401). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters. Note that two *A. robustus* share the largest measurement in stage 1.



Figure A1.21: Mandibular length from the posterior condyle to the septum between I_2 -C₁ (X403). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.22: Corpus length from the posteromedial margin of the lingual tuberosity to the mental foramen (X404). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.23: Corpus length from the anterior margin of the mandibular foramen to the lingual alveolar septum between the P_4 - M_1 (X415). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.

Overall mandibular breadths



Figure A1.24: Mandibular breadth between the left and right mental foramina (X501). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.25: Mandibular breadth between the left and right C_1 - P_3 septa (labial; X502). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.26: Mandibular breadth between the left and right C₁-P₃ septa (lingual; X502). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.27: Mandibular breadth between the left and right P_3 - P_4 septa (labial; X502). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.

Ramus length and height



Figure A1.28: Length from the posterior ramus margin to the (X603). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.



Figure A1.29: Height from the mandibular foramen to the ramus base (X607). Green circles are modern humans and black squares are *A. robustus*. Measurements are in millimeters.

Appendix II: Mandibular corpus robusticity



Figure A2.1: Corpus robusticity (breadth / height) at the position between the P_{3-4} . Green circles are modern humans and black squares are *A. robustus*.



Figure A2.2: Corpus robusticity (breadth / height) at the position of P_4 . Green circles are modern humans and black squares are *A. robustus*.



Figure A2.3: Corpus robusticity (breadth / height) at the position between the P_4 - M_1 . Green circles are modern humans and black squares are *A. robustus*.



Figure A2.4: Corpus robusticity (breadth / height) at the position of M₁. Green circles are modern humans and black squares are *A. robustus*.



Figure A2.5: Corpus robusticity (breadth / height) at the position between the M_{1-2} . Green circles are modern humans and black squares are *A. robustus*.



Figure A2.6: Corpus robusticity (breadth / height) at the position of M_2 . Green circles are modern humans and black squares are *A. robustus*.



Figure A2.6: Corpus robusticity (breadth / height) at the mandibular symphysis. Green circles are modern humans and black squares are *A. robustus*.

Appendix III: R Code for Analyses

Instructions for using this code are given in this font (Times New Roman), whereas the actual code is in Monaco font.

```
setwd("") # Enter the name of the directory of data file
data = data.frame(read.csv(""))
```

To begin, read in the .csv format data file as a data.frame by typing the file name into the quotation marks above. Individuals should be rows, the first column should be their ages, and the remaining columns the measurements/variables. Divide this data.frame into two, one for the fossils (rob) and one for the referent (lib). The next line of code transposes these so that in each data.frame, the individuals are columns and their measurements make up the rows. These will be resampled in the following codes.

```
p.lib = data.frame(t(lib)); p.rob=data.frame(t(rob))
```

```
Function to calculate the geometric mean (pairwise size metric)
```

```
GM = function(r1) {
    prod(r1)^(1/nrow(r1))
}
```

Declare the variables to be resampled

```
D = NA  # test statistic of size difference
k = NA  # number of traits
al = NA; a2 = NA  # ages in the comparison. Later can be used to examine
differences between specific age groups
f1 = NA; f2 = NA; h1 = NA; h2 = NA  # store each of the individuals each
iteration
```

```
Validation of \zeta test for the pairwise size metric (Chapter 4.1)
```

```
for (i in 1:75000)
                  {
r1 = sample(p.rob,1); r2 = sample(p.rob[,p.rob[1,]!=r1[1,1]],1) # 2 robusti
   l1 = sample(p.lib[,p.lib[1,]==r1[1,1]],2,replace=T)
                                                        # getting 2, w
replacement
  12 = sample(p.lib[,p.lib[1,]==r2[1,1]],2,replace=T)
sub = data.frame(na.omit(cbind(r1,r2,l1,l2)))
   k[i] = nrow(sub)-1; # number of traits
# ordered subset by age
  if (sub[1,1]<sub[1,2]) {ry=(sub[3]); ro=(sub[5])} else {ry=(sub[5]);
ro=(sub[3])}
  if (sub[1,1]<sub[1,2]) {hy = (sub[4]); ho=(sub[6])} else {hy=(sub[6]);
ho=(sub[4])
    s2 = data.frame(cbind(ry,ro,hy,ho))
      a1[i] = s2[1,1]; a2[i] = s2[1,2] ; # extract ages now while you still
can!
     s2=s2[2:nrow(s2),]
# calculate ratios, and correct for 'negative growth if necessary; block out if
making no assumptions
   if (k[i]==0) {gml = NA; gmr = NA} else {
```

```
gmr=GM(s2[2])/GM(s2[1]); if (gmr<1) {gmr = 1} # {=1} or {=NA} makes
assumption about growth &/or sampling
     gml=GM(s2[4])/GM(s2[3]); if (gml<1) {gml = 1}
    }
# calculate test statistic
   if (k[i] == 0) {D[i] = NA} else {
    D[i] = gmr-gml
  }
 # extract information about subsample
    if (k[i]==0) {f1[i] = NA; f2[i] = NA; h1[i] = NA; h2[i]=NA} else {
      f1[i]=names(ry); f2[i]=names(ro); h1[i]=names(hy); h2[i]=names(ho)
} #
         #
              #
                # #
                          #
                                  FIN!
      #
```

ζ test for the pairwise size metric (Chapter 4.2)

```
for (i in 1:50000)
                   - {
# randomly grab 2 differently aged robustus and 2 humans
r1 = sample(p.rob,1); r2 = sample(p.rob[,p.rob[1,]!=r1[1,1]],1)
   l1 = sample(p.lib[,p.lib[1,]==r1[1,1]],1)
  12 = sample(p.lib[,p.lib[1,]==r2[1,1]],1)
sub = data.frame(na.omit(cbind(r1,r2,l1,l2)))
   k[i] = nrow(sub)-1;
                         # number of traits
# ordered subset by age
  if (sub[1,1]<sub[1,2]) {ry=(sub[1]); ro=(sub[2])} else {ry=(sub[2]);
ro=(sub[1])
  if (sub[1,1]<sub[1,2]) {hy =(sub[3]); ho=(sub[4])} else {hy=(sub[4]);
ho=(sub[3])}
    s2 = data.frame(cbind(ry,ro,hy,ho))
      a1[i] = s2[1,1]; a2[i] = s2[1,2] ; # extract ages now while you still
can!
      s2=s2[2:nrow(s2),]
# calculate ratios, and correct for 'negative growth if necessary; block out if
making no assumptions
if (k[i]==0) {gml = NA; gmr = NA} else {
      gmr=GM(s2[2])/GM(s2[1]); if (gmr<1) {gmr = 1}
     gml=GM(s2[4])/GM(s2[3]); if (gml<1) {gml = 1}
   }
# calculate test statistic
   if (k[i] == 0) {D[i] = NA} else {
    D[i] = qmr-qml
  }
 # extract information about subsample
   if (k[i]==0) {f1[i] = NA; f2[i] = NA; h1[i] = NA; h2[i]=NA} else {
     f1[i]=names(ry); f2[i]=names(ro); h1[i]=names(hy); h2[i]=names(ho)
} #
         #
              #
                #
                    #
                          #
                                  FIN!
      #
```

ζ test for individual traits (Chapter 4.3)

```
a1[i] = s2[1,1]; a2[i] = s2[1,2] ; # extract ages now while you still
can!
      s2=s2[2:nrow(s2),]
# pick a random variable
v = s2[sample(1:nrow(s2),1),]
    if (k[i]==0) {tr[i]=NA} else tr[i]= row.names(v[1,]) # which trait?
# calculate species' ratios, with ratios <1 set equal to 1</pre>
    if (k[i]==0) {gml = NA; gmr = NA} else {
      gmr=v[1,2]/v[1,1]; if (gmr<1) {gmr = 1}
      gml=(v[1,4])/(v[1,3]); if (gml<1) {gml = 1}
    }
# calculate test statistic
   if (k[i] == 0) \{ D[i] = NA \} else \{ D[i] = qmr-qml \}
  # extract information about subsample
   if (k[i]==0) {f1[i] = NA; f2[i] = NA; h1[i] = NA; h2[i]=NA} else {
     f1[i]=names(ry); f2[i]=names(ro); h1[i]=names(hy); h2[i]=names(ho)
      }
          #
              # # # #
} #
    #
                                  FIN!
```

Save a copy

For each of the above resampling procedures, the output can be stored as a .csv file for future analysis in Excel, R, etc.

```
da = a2-a1 # difference between pair's age groups
total = data.frame(na.omit(cbind(tr,D,k,a1,da,f1,f2,h1,h2))) # all the
resampled data in a nice matrix
write.csv(total,file="output.csv")
```

Bibliography

- Ackermann RR, and Krovitz GE. 2002. Common patterns of facial ontogeny in the hominid lineage. *The Anatomical Record* 269: 142-147.
- Ackermann RR. 2005. Variation in Neandertals: a response to Harvati (2003). *Journal of Human Evolution* 48: 643-646.
- Ahern JCM, Hawks JD, and Lee S-H. 2005. Neandertal taxonomy reconsidered, again: A response to Harvati et al. (2003). *Journal of Human Evolution*: 647-652.
- Aiello LC, Montgomery C, and Dean MC. 1991. The natural history of deciduous tooth attrition in hominoids. *Journal of Human Evolution* 21: 397-412.
- Aiello LC, and Wheeler P. 1995. The expensive-tissue hypothesis: The brain and the digestive system in human and primate evolution. *Current Anthropology* 36: 199-221.
- Alberch P, Gould SJ, Oster GF, and Wake DB. 1979. Size and shape in ontogeny and phylogeny. *Paleobiology* 5: 296-317.
- Allman J, and Hasenstaub A. 1999. Brains, maturation times, and parenting. *Neurobiology of Aging* 20: 447-454.
- Anemone RL, Mooney MP, and Siegel MI. 1996. Longitudinal study of dental development in chimpanzees of known chronological age: Implications for understanding the age at death of Plio-Pleistocene hominids. *American Journal of Physical Anthropology* 133: 119-133.
- Antoine D, Hillson S, and Dean MC. 2009. The developmental clock of dental enamel: A test for the periodicity of prism cross-striations in modern humans and an evaluation of the most likely sources of error in histological studies of this kind. *Journal of Anatomy* 214: 45-55.
- Antón SC, and Leigh SR. 2003. Growth and life history in *Homo erectus*. In: Thompson JL, Krovitz GE, Andrew J Nelson, editors. *Patterns of Growth and Development in the Genus Homo*. Cambridge: Cambridge University Press. p 219-245.
- Backwell LR, and D'Errico F. 2001. Evidence of termite foraging by Swartkrans early hominids. *Proceedings of the National Academy of Sciences* 98: 1358-63.
- Backwell L, and d' Errico F. 2008. Early hominid bone tools from Drimolen, South Africa. *Journal of Archaeological Science* 35: 2880-2894.
- Balter V, Blichert-Toft J, Braga J, Telouk P, Thackeray JF, and Albarède F. 2008. U–Pb dating of fossil enamel from the Swartkrans Pleistocene hominid site, South Africa. *Earth and Planetary Science Letters* 267: 236-246.
- Beecher RM, and Corruccini RS. 1981. Effects of dietary consistency on craniofacial and occlusal development in the rat. *The Angle Orthodontist* 51: 61-69.
- Begg PR. 1954. Stone Age man's dentition: With reference to anatomically correct occlusion, the etiology of malocclusion, and a technique for its treatment. *American Journal of Orthodontics* 40: 517-531.

- Berge C. 1998. Heterochronic processes in human evolution: an ontogenetic analysis of the hominid pelvis. *American Journal of Physical Anthropology* 105: 441-59.
- Berge C, and Gommery D. 1999. Le sacrum de Sterkfontein Sts 14 Q (*Australopithecus africanus*): Nouvelles donnees sur la croissance et sur l'age osseux du specimen (hommage A R. Broom et J. T. Robinson). *Comptes Rendus Academie Sciences Paris* 329: 227-232.
- Bergersen EO. 1972. The male adolescent facial growth spurt: Its prediction and relation to skeletal maturation. *The Angle Orthodontist* 42: 319-338.
- Beynon AD, and Dean MC. 1988. Distinct dental development patterns in early fossil hominids. *Nature* 355: 509-14.
- Björk A. 1969. Prediction of mandibular growth rotation. *American Journal of* Orthodontics and Dentofacial Orthopedics 55: 585-599.
- Björk A, and Skieller V. 1972. Facial development and tooth eruption: An implant study at the age of puberty. *American Journal of Orthodontics* 62: 339-383.
- Bogin B. 1997. Evolutionary hypotheses for human childhood. *Yearbook of Physical Anthropology* 89: 63-89.
- Bogin B. 1999. Evolutionary Perspective on Human Growth. Annual Review of Anthropology 28: 109-153.
- Bogin B. 2003. The human pattern of growth and development in paleontological perspective. In: Thompson JL, Krovitz GE, Nelson AJ, editors. *Patterns of Growth and Development in the Genus Homo*. Cambridge: Cambridge University Press. p 15-44.
- Bogin B. 2009. Childhood, adolescence, and longevity: A multilevel model of the evolution of reserve capacity in human life history. *American Journal of Human Biology* 21: 567-577.
- Bogin B, and Smith BH. 1996. Evolution of the Human Life Cycle. *American Journal of Human Biology* 8: 703-716.
- Boughner JC, and Dean MC. 2004. Does space in the jaw influence the timing of molar crown initiation? A model using baboons (*Papio anubis*) and great apes (*Pan troglodytes, Pan paniscus*). Journal of Human Evolution 46: 255-77.
- Braga J. 1998. Chimpanzee Variation Facilitates the Interpretation of the Incisive Suture Closure in South African Plio-Pleistocene Hominids. *American Journal of Physical Anthropology* 135: 121-135.
- Braga J, and Thackeray JF. 2003. Early *Homo* at Kromdraai B : probabilistic and morphological analysis of the lower dentition. *Homo* 2: 269-279.
- Brain CK. 1981. *The Hunters or the Hunted? An Introduction to African Cave Taphonomy*. Chicago: University of Chicago Press.
- Brain CK. 2004. Swartkrans: *A Cave's Chronicle of Early Man*. Second edition. Pretoria: Transvaal Museum.
- Bromage TG. 1989. Ontogeny of the early hominid face. *Journal of Human Evolution*: 751-773.
- Bromage TG, and Dean MC. 1985. Re-evaluation of the age at death of immature fossil hominids. *Nature* 317: 525-7.
- Broom R. 1938. The Pleistocene anthropoid apes of South Africa. Nature 142: 377-379.
- Broom R. 1949. Another new type of fossil ape-man. Nature 163: 57.
- Broom R, and Robinson JT. 1949. A new type of fossil man. Nature 164: 322-323.

- Broom R, and Robinson JT. 1952. Swartkrans Ape-Man: *Paranthropus crassidens*. *Memoirs of the Transvaal Museum* 6: 123.
- Brown FH, Harris J, Leakey R, and Walker A. 1985. Early *Homo erectus* skeleton from West Lake Turkana, Kenya. *Nature* 316: 788-92.
- Brown P, Sutikna T, Morwood MJ, Soejono RP, Jatmiko, Saptomo EW, and Due RA. 2004. A new small-bodied hominin from the Late Pleistocene of Flores, Indonesia. *Nature* 431:1055-1061.
- Bullington J. 1991. Deciduous dental microwear of prehistoric juveniles from the Lower Illinois River Valley. *American Journal of Physical Anthropology* 84: 59-73.
- Bulygina E, Mitteroecker P, and Aiello LC. 2006. Ontogeny of facial dimorphism and patterns of individual development within one human population. *American Journal of Physical Anthropology* 131: 432-43.
- Campbell BC. 2006. Adrenarche and the Evolution of Human Life History. *American Journal of Human Biology* 18: 569-589.
- Campbell BC. 2011. Adrenarche and middle childhood. Human Nature 22: 327-349.
- Cardoso HFV. 2007. Environmental effects on skeletal versus dental development: Using a documented subadult skeletal sample to test a basic assumption in human osteological research. *American Journal of Physical Anthropology* 132: 223-233.
- Cerling TE, Mbua E, Kirera FM, Kyalo F, Grine FE, and Leakey MG. 2011. Diet of *Paranthropus boisei* in the early Pleistocene of East Africa. *Proceedings of the National Academy of Sciences* 108: 9337-9340.
- Charnov EL, and Berrigan D. 1993. Why do female primates have such long lifespans and so few babies? or life in the slow lane. *Evolutionary Anthropology* 1: 191-194.
- Clegg M, and Aiello LC. 1999. A comparison of the Nariokotome *Homo erectus* with juveniles from a modern human population. *American Journal of Physical Anthropology* 93: 81-93.
- Cobb SN, and O'Higgins P. 2004. Hominins do not share a common postnatal facial ontogenetic shape trajectory. *Journal of Experimental Zoology. Part B, Molecular and Developmental Evolution* 321: 302-321.
- Cobb SN, and Higgins PO. 2007. The ontogeny of sexual dimorphism in the facial skeleton of the African apes. *Journal of Human Evolution* 53: 176-190.
- Cofran Z, and Thackeray JF. 2010. One or two species? A morphometric comparison between robust australopithecines from Kromdraai and Swartkrans. *South African Journal of Science* 106: 2-5.
- Conroy GC, and Vannier MW. 1991a. Dental development in South African australopithecines. Part I: Problems of pattern and chronology. *American Journal* of Physical Anthropology 86: 121-136.
- Conroy GC, and Vannier MW. 1991b. Dental development in South African australopithecines. Part II: Dental stage assessment. *American Journal of Physical Anthropology* 86: 137-156.
- Cooper DML, Erickson B, Peele a G, Hannah K, Thomas CDL, and Clement JG. 2011. Visualization of 3D osteon morphology by synchrotron radiation micro-CT. *Journal of Anatomy*: 481-489.

Coquerelle M, Bookstein FL, Braga J, Halazonetis DJ, and Weber GW. 2010a. Fetal and infant growth patterns of the mandibular symphysis in modern humans and chimpanzees (*Pan troglodytes*). *Journal of Anatomy* 217: 507-520.

Coquerelle M, Bayle P, Bookstein FL, Braga J, Halazonetis DJ, Katina S, and Weber GW. 2010b. The association between dental mineralization and mandibular form: A study combining additive conjoint measurement and geometric morphometrics. Journal of Anthropological Sciences 88:129-150.

Coquerelle M, Bookstein FL, Halazonetis DJ, Weber GW, and Mitteroecker P. 2011. Sexual dimorphism of the human mandible and its association with dental development. American Journal of Physical Anthropology 145:192-202.

Coqueugniot H, Hublin J-J, Veillon F, Houët F, and Jacob T. 2004. Early brain growth in *Homo erectus* and implications for cognitive ability. *Nature* 431: 299-302.

Coqueugniot H, and Hublin J-J. 2012. Age-related changes of digital endocranial volume during human ontogeny: Results from an osteological reference collection. *American Journal of Physical Anthropology* 147: 312-318.

Crespi BJ. 2011. The evolutionary biology of child health. *Proceedings of the Royal* Society B 278: 1441-1449.

Daegling DJ. 1989. Biomechanics of cross-sectional size and shape in the hominoid mandibular corpus. *American Journal of Physical Anthropology* 80: 91-106.

Daegling DJ. 1996. Growth in the mandibles of African apes. *Journal of Human Evolution* 30: 315-341.

- Daegling DJ. 2007. Morphometric estimation of torsional stiffness and strength in primate mandibles. *American Journal of Physical Anthropology* 132:261-266.
- Daegling DJ, and Grine FE. 1991. Compact bone distribution and biomechanics of early hominid mandibles. *American Journal of Physical Anthropology* 86: 321-339.

Daegling DJ, and Hylander WL. 2000. Experimental observation, theoretical models, and biomechanical inference in the study of mandibular form. *American Journal of Physical Anthropology* 112: 541-551.

Dahlberg AA, and Menegaz-Bock RM. 1958. Emergence of the permanent teeth in Pima Indian children. *Journal of Dental Research* 17:1123-1140.

Darroch JN, and Mosimann JE. 1985. Canonical and principal components of shape. *Biometrika* 72: 241-252.

de Ruiter DJ. 2003. Revised faunal lists for Members 1-3 of Swartkrans, South Africa. Annals of the Transvaal Museum 40: 29-41.

DeSilva JM. 2011. A shift toward birthing relatively large infants early in human evolution. *Proceedings of the National Academy of Sciences* 108: 1022-1027.

DeSilva JM, and Lesnik JJ. 2006. Chimpanzee neonatal brain size: Implications for brain growth in *Homo erectus. Journal of Human Evolution* 51: 207-212.

DeSilva JM, and Lesnik JJ. 2008. Brain size at birth throughout human evolution: a new method for estimating neonatal brain size in hominins. *Journal of Human Evolution* 55: 1064-74.

Dean MC, Leakey MG, Reid D, Schrenk F, Schwartz GT, Stringer C, and Walker A. 2001. Growth processes in teeth distinguish modern humans from *Homo erectus* and earlier hominins. *Nature* 414: 1-4.

Dean MC, Beynon AD, Thackeray JF, and Macho GA. 1993. Histological reconstruction of dental development and age at death of a juvenile *Paranthropus robustus*

specimen, SK 63, from Swartkrans, South Africa. *American Journal of Physical Anthropology* 91: 401-19.

- Dean MC, and Smith BH. 2009. Growth and development of the Nariokotome youth, KNM-WT 15000. In: *The First Humans: Origin and Early Evolution of the Genus Homo*. Springer. p 101-120.
- Dobbing J, and Sands J. 1979. Comparative aspects of the brain growth spurt. *Early Human Development* 311: 79-83.
- Efron B, and Tibshirani R. 1991. Statistical data analysis in the computer age. *Science* 253: 390-395.
- Enlow DH. 1962. A study of the post-natal growth and remodeling of bone. *American Journal of Anatomy* 110:79-101.
- Enlow DH, and Harris DB. 1964. A study of the postnatal growth of the human mandible. *American Journal of Orthodontics* 50: 25-50.
- Enlow DH, Kuroda T, and Lewis AB. 1971. The morphological and morphogenetic basis for craniofacial form and pattern. *The Angle Orthodontist* 41: 161-188.
- Enlow DH, and Hans MG. 1996. Essentials of Facial Growth. Philadelphia: Saunders.
- Eveleth PB, and Tanner JM. 1988. *Worldwide Variation in Human Growth*. Cambridge: Cambridge University Press.
- Franchi L, Baccetti T, and McNamara J. 2001. Thin-plate spline analysis of mandibular growth. *The Angle Orthodontist* 71: 83-89.
- Franchi L, Baccetti T, Stahl F, and McNamara JA. 2007. Thin-plate spline analysis of craniofacial growth in Class I and Class II subjects. *The Angle Orthodontist* 77: 595-601.
- Fukase H, and Suwa G. 2008. Growth-related changes in prehistoric Jomon and modern Japanese mandibles with emphasis on cortical bone distribution. *American Journal of Physical Anthropology* 136: 441-454.
- Fukase H, and Suwa G. 2010. Influence of size and placement of developing teeth in determining anterior corpus height in prehistoric Jomon and modern Japanese mandibles. *Anthropological Science* 118: 75-86.
- Fuller K. 1996. Analysis of the probability of multiple taxa in a combined sample of Swartkrans and Kromdraai dental material. *American Journal of Physical Anthropology* 101: 429-439.
- Garn SM, and Koski K. 1957. Tooth eruption sequence in fossil and recent man. *Nature* 190: 442-443.
- Gathogo PN, and Brown FH. 2006. Revised stratigraphy of Area 123, Koobi Fora, Kenya, and new age estimates of its fossil mammals, including hominins. Journal of Human Evolution 51: 471-479.
- German RZ. 2004. The ontogeny of sexual dimorphism: The implications of longitudinal vs. cross-sectionl data for studying heterochrony in mammals. In: Anapol F, German RZ, Jablonski NG, editors. *Shaping Primate Evolution*. Cambridge: Cambridge University Press. p 11-23.
- Gibbard PL, Head MJ, Walker MJC, and The Subcommission on Quaternary Stratigraphy. 2010. Formal ratification of the Quaternary System/Period and the Pleistocene Series/Epoch with a base at 2.58 Ma. *Journal of Quaternary Science* 25: 96-102.

- Gilbert SF, Opitz JM, and Raff RA. 1996. Resynthesizing evolutionary and developmental biology. *Developmental Biology* 173: 357-72.
- Gordon AD, Green DJ, and Richmond BG. 2008. Strong postcranial size dimorphism in *Australopithecus afarensis*: Results from two new resampling methods for multivariate data sets with missing data. *American Journal of Physical Anthropology* 328: 311-328.
- Gould SJ. 1977. Ontogeny and Phylogeny. Cambridge: Belknap Press.
- Gould SJ. 2000. Of coiled oysters and big brains: How to rescue the terminology of heterochrony, now gone astray. *Evolution and Development* 2: 241-248.
- Gould SJ, and Lewontin RC. 1979. The spandrels of San Marco and the Panglossian Paradigm: A critique of the Adaptationist programme. *Proceedings of the Royal Society B* 205: 581-598.
- Grine FE. 1982. A new juvenile hominid (Mammalia: Primates) from Member 3, Kromdraai Formation, Transvaal, South Africa. *Annals of the Transvaal Museum* 33: 165-239.
- Grine FE. 1987. On the eruption pattern of the permanent incisors and first permanent molars in *Paranthropus*. *American Journal of Physical Anthropology* 72: 353-359.
- Grine FE, and Daegling DJ. 1993. New mandible of *Paranthropus robustus* from Member 1, Swartkrans Formation, South Africa. *Journal of Human Evolution* 24: 319-333.
- Grine FE, Sponheimer M, Ungar PS, Lee-Thorp J, and Teaford MF. 2012. Dental microwear and stable isotopes inform the paleoecology of extinct hominins. *American Journal of Physical Anthropology* 148: 285-317.
- Guatelli-Steinberg D. 2009. Recent studies of dental development in Neandertals: Implications for Neandertal life histories. *Evolutionary Anthropology* 18: 9-20.
- Gurven M, and Walker RS. 2006. Energetic demand of multiple dependents and the evolution of slow human growth. *Proceedings of the Royal Society B* 273: 835-841.
- Hall BK. 2007. Homoplasy and homology: Dichotomy or continuum? *Journal of Human Evolution* 52: 473-479.
- Hamada Y, and Udono T. 2002. Longitudinal analysis of length growth in the chimpanzee (*Pan troglodytes*). *American Journal of Physical Anthropology* 118: 268-284.
- Hawkes K, O'Connel JF Blurton Jones NG, Alvarez H, and Charnov EL. 1998. Grandmothering, menopause, and the evolution of human life histories. *Proceedings of the National Academy of Sciences* 95: 1336-1339.
- Held LI. 2010. The evolutionary geometry of human anatomy: Discovering our inner fly. *Evolutionary Anthropology* 19: 227-235.
- Hens SM. 2005. Ontogeny of craniofacial sexual dimorphism in the orangutan (*Pongo pygmaeus*) I: Face and palate. *American Journal of Primatology* 65: 149-166.
- Herries AIR, Curnoe D, and Adams JW. 2009. A multi-disciplinary seriation of early *Homo* and *Paranthropus* bearing palaeocaves in southern Africa. *Quaternary International* 202: 14-28.
- Hochberg Z. 2012. Evo-Devo of Child Growth: A Treatise on Child Growth and Human Evolution. Hoboken: Wiley-Blackwell.

Holmes MA, and Ruff CB. 2011. Dietary effects on development of the human mandibular corpus. *American Journal of Physical Anthropology* 145: 615-628.

- Howell N. 1982. Village composition implied by a paleodemographic life table: The Libben site. *American Journal of Physical Anthropology* 59: 263-269.
- Humphrey LT. 2010. Weaning behaviour in human evolution. Seminars in Cell & Developmental Biology 21: 453-461.
- Huxley JS. 1932. Problems of Relative Growth. London: Methuen & Co. Ltd.
- Hägg U, and Taranger J. 1981. Dental emergence stages and the pubertal growth spurt. *Acta Odontologica Scandinavica* 39: 295-306.
- Johnson PA, Atkinson PJ, and Moore WJ. 1976. The development and structure of the chimpanzee mandible. *Journal of Anatomy* 122: 467-477.
- Jolicoeur P, Abidi H, and Pontier J. 1991. Human stature: Which growth model? *Growth, Development and Aging* 55: 129-132.
- Jungers WL, Falsetti AB, and Wall CE. 1995. Shape, relative size, and size-adjustments in morphometrics. *American Journal of Physical Anthropology* 38: 137-161.
- Kaplan H, Hill K, Lancaster J, and Hurtado AM. 2000. A theory of human life history evolution: Diet, intelligence, and longevity. *Evolutionary Anthropology* 9: 156-185.
- Karlsen AT. 1997. Association between facial height development and mandibular growth rotation in low and high MP-SN angle faces: A longitudinal study. *The Angle Orthodontist* 67: 103-110.
- Kaszycka KA. 2002. Status of Kromdraai: Cranial, Mandibular and Dental Morphology, Systematic Relationships, and Significance of the Kromdraai Hominids. Paris: CNRS Éditions.
- Kaszycka KA. 2010. Sexual dimorphism and social structure in *Australopithecus robustus*: How strong is the evidence for "hominid harems"? *American Journal of Physical Anthropology* S50: 139.
- Kennedy GE. 2005. From the ape's dilemma to the weanling's dilemma: Early weaning and its evolutionary context. *Journal of Human Evolution* 48: 123-45.
- Keyser AW. 2000. The Drimolen skull: the most complete australopithecine cranium and mandible to date. *South African Journal of Science* 96: 189-193.
- Keyser AW, Menter CG, Moggi-Cecchi J, Pickering TR, and Berger LR. 2000. Drimolen: A new hominid-bearing site in Gauteng, South Africa. South African Journal of Science 96: 193-197.
- Kimbel WH, White TD, and Johanson DC. 1988. Implications of KNM-WT 17000 for the evolution of "robust" Australopithecus. In: Grine FE, editor. *Evolutionary History of the "Robust" Australopithecines*. New York: Aldine de Gruyter. p 259-268.
- Koshy S, and Tandon S. 1998. Dental age assessment: The applicability of Demirjian's method in South Indian children. *Forensic Science International* 94: 73-85.
- Kraemer HC, Horvat JR, Doering C, and McGinnis PR. 1982. Male chimpanzee development focusing on adolescence: Integration of behavioral with physiological changes. *Primates* 23: 393-405.
- Kramer KL, and Greaves RD. 2010. Synchrony between growth and reproductive patterns in human females: early investment in growth among Pumé foragers. *American Journal of Physical Anthropology* 244: 235-244.

- Krogman WM. 1931a. Growth changes in the skull and face of the Orang-Utan. *American Journal Of Anatomy* 47: 343-365.
- Krogman WM. 1931b. Growth changes in the skull and face of the Chimpanzee. *American Journal Of Anatomy* 47: 325-342.
- Krogman WM. 1931c. Growth changes in the skull and face of the Gorilla. American *Journal of Anatomy* 47: 89-115.
- Kuykendall KL. 1996. Dental development in chimpanzees (*Pan troglodytes*): The Timing of Tooth Calcification Stages. *American Journal of Physical Anthropology* 157: 135-157.
- Lacruz RS, Rozzi FR, and Bromage TG. 2006. Variation in enamel development of South African fossil hominids. *Journal of Human Evolution* 51: 580-90.
- Lacruz RS, Dean MC, Ramirez-Rozzi F, and Bromage TG. 2008. Megadontia, striae periodicity and patterns of enamel secretion in Plio-Pleistocene fossil hominins. *Journal of Anatomy* 213: 148-158.
- Lague MR, Collard NJ, Richmond BG, and Wood BA. 2008. Hominid mandibular corpus shape variation and its utility for recognizing species diversity within fossil Homo. *Journal of Anatomy* 213: 670-685.
- Laird AK. 1967. Evolution of the human growth curve. Growth 31: 345-355.
- Leakey LSB, and Leakey MD. 1964. Recent discoveries of fossil hominids in Tanganyika: At Olduvai and near Lake Natron. *Nature* 202: 5-7.
- Leakey LSB, Tobias PV, and Napier JR. 1964. A new species of the genus *Homo* from Olduvai Gorge. *Nature* 202: 7-9.
- Leakey RE, and Walker A. 1976. *Australopithecus*, *Homo erectus* and the single species hypothesis. *Nature* 26: 572-574.
- Lee S-H. 2011. How many variables are too few? Effect of sample size in STET, a method to test conspecificity for pairs of unknown species. *PaleoAnthropology*: 260-267.
- Lee S-H, and Wolpoff MH. 2003. The pattern of evolution in Pleistocene human brain size. *Paleobiology* 29: 186-196.
- Lee S-H, and Wolpoff MH. 2005. Habiline variation: A new approach using STET. *Theory in Biosciences* 124: 25-40.
- Lee-Thorp J, and Sponheimer M. 2006. Contributions of biogeochemistry to understanding hominin dietary ecology. *Yearbook of Physical Anthropology* 148: 131-148.
- Lee-thorp JA, Thackeray JF, and van der Merwe N. 2000. The hunters and hunted revisited. *Journal of Human Evolution* 39: 565-576.
- Leigh SR. 1996. Evolution of human growth spurts. *American Journal of Physical Anthropology* 101: 455-474.
- Leigh SR. 2001. Evolution of human growth. Evolutionary Anthropology 10: 223-236.
- Leigh SR. 2006. Brain ontogeny and life history in *Homo erectus*. Journal of Human Evolution 50: 104-108.
- Leigh SR, and Shea BT. 1996. Ontogeny of body size variation in African apes. *American Journal of Physical Anthropology* 99: 43-65.
- Leigh SR, and Park PB. 1998. Evolution of Human Growth Prolongation. American Journal of Physical Anthropology 107: 331-350.

- Lele SR, and Richtsmeier JT. 2001. An Invariant Approach to Statistical Analysis of Shapes. Boca Raton: CRC Press, LLC.
- Leonard WR, and Robertson ML. 1992. Nutritional requirements and human evolution: A bioenergetics Model. *American Journal of Human Biology* 4: 179-195.
- Leonard WR, and Robertson ML. 1996. On Diet, Energy Metabolism, and Brain Size in Human Evolution. *Current Anthropology* 37: 125-129.
- Leonard WR, Robertson ML, Snodgrass JJ, and Kuzawa CW. 2003. Metabolic correlates of hominid brain evolution. *Comparative Biochemistry and Physiology Part A: Molecular & Integrative Physiology* 136: 5-15.
- Lesnik JJ. 2011a. Tools and Termites: Implications for the Foraging Behavior of the Swartkrans Hominids. PhD thesis, University of Michigan.
- Lesnik JJ. 2011b. Bone Tool Texture Analysis and the Role of Termites in the Diet of South African Hominids. *PaleoAnthropology*: 268-281.
- Lesnik J, and Thackeray JF. 2007. The efficiency of stone and bone tools for opening termite mounds: Implications for hominid tool use at Swartkrans. *South African Journal of Science* 103: 354-356.
- Lewis AB, and Garn SM. 1960. The relationship between tooth formation and other maturational factors. *The Angle Orthodontist* 30: 70-77.
- Lewis AB, Roche AF, and Wagner B. 1985. Pubertal spurts in cranial base and mandible: Comparisons within individuals. *The Angle Orthodontist* 55: 17-30.
- Liversidge HM. 2003. Variation in modern human dental development. In: Thompson JL, Krovitz GE, Nelson AJ, editors. Patterns of Growth and Development in the Genus Homo. Cambridge: Cambridge University Press. p 73-1113.
- Locke JL, and Bogin B. 2006. Language and life history: A new perspective on the development and evolution of human language. *Behavioral and Brain Sciences* 29: 259-325.
- Lockwood CA. 1999. Sexual dimorphism in the face of Australopithecus africanus. American Journal of Physical Anthropology 127: 97-127.
- Lockwood CA, Menter CG, Moggi-Cecchi J, and Keyser AW. 2007. Extended male growth in a fossil hominin species. *Science* 318: 1443-1446.
- Lordkipanidze D, Jashashvili T, Vekua A, Ponce de León MS, Zollikofer CPE, Rightmire GP, Pontzer H, Ferring R, Oms O, Tappen M, et al. 2007. Postcranial evidence from early *Homo* from Dmanisi, Georgia. *Nature* 449: 305-310.
- Lovejoy CO. 1981. The Origin of Man. Science 211: 341-350.
- Lovejoy CO. 2009. Reexamining Human Origins in Light of *Ardipithecus ramidus*. *Science* 326: 74e1-8.
- Lovejoy CO, Meindl RS, Pryzbeck TR, Barton TJ, Heiple KG, and Kotting D. 1977. Paleodemography of the Libben Site, Ottawa County, Ohio. *Science* 198: 291-293.
- Lovejoy CO, Russell KF, and Harrison ML. 1990. Long bone growth velocity in the Libben population. *American Journal of Human Biology* 2: 533-541.
- Ludwig KR, and Renne PR. 2000. Geochronology on the paleoanthropological time scale. *Evolutionary Anthropology* 9: 101.
- Macho GA. 2001. Primate molar crown formation times and life history evolution revisited. *American Journal of Primatology* 55: 189-201.

- Macho GA, and Wood BA. The Role of Time and Timing in Hominid Dental Evolution. *Evolutionary Anthropology*: 17-31.
- Manly BF. 2007. *Randomization, Bootstrap and Monte Carlo Methods in Biology*. Third edition. Chapman & Hall / CRC.
- Mann A. 1975. Some Paleodemographic Aspects of the South African Australopithecines. Philadelphia: University of Pennsylvania Publications.
- Mann A. 1988. The nature of Taung dental maturation. *Nature* 333: 123.
- Mann A, Lampl M, and Monge J. 1990. Patterns of ontogeny in human evolution: Evidence from dental development. *American Journal of Physical Anthropology* 33: 111-150.
- Marsden SB, Marsden D, and Thompson ME. 2006. Demographic and female life history parameters of free-ranging chimpanzees at the Chimpanzee Rehabilitation Project, River Gambia National Park. *International Journal of Primatology* 27: 391-410.
- Martín-González J, Mateos A, Goikoetxea I, Leonard WR, and Rodríguez J. 2012. Differences between Neandertal and modern human infant and child growth models. *Journal of Human Evolution* 2012.
- Masterson TJ, and Leutenegger W. 1992. Ontogenetic patterns of sexual dimorphism in the cranium of Bornean orang-utans (*Pongo pygmaeus pygmaeus*). Journal of Human Evolution 23: 3-26.
- Mattfeldt T. 2011. A brief introduction to computer-intensive methods, with a view towards applications in spatial statistics and stereology. *Journal of Microscopy* 242: 1-9.
- McCollum MA. 1997. Palatal thickening and facial form in *Paranthropus*: Examination of alternative developmental models. *American Journal of Physical Anthropology* 103: 375-392.
- McCollum MA. 1999. The robust australopithecine face: A morphogenetic perspective. *Science* 284: 301-305.
- McCollum MA. 2008. Nasomaxillary remodeling and facial form in robust *Australopithecus*: a reassessment. *Journal of Human Evolution* 54: 2-14.
- McCollum MA, and Sharpe PT. 2001. Developmental genetics and early hominid craniodental evolution. *BioEssays* 23: 481-493.
- McCollum MA, and Ward SC. 1997. Subnasoalveolar anatomy and hominoid phylogeny: Evidence from comparative ontogeny. *American Journal of Physical Anthropology* 102: 377-405.
- McHenry HM, and Coffing K. 2000. *Australopithecus* to *Homo*: Transformations in body and mind. *Annual Review of Anthropology* 29: 125-46.
- McNulty KP, Frost SR, and Strait DS. 2006. Examining affinities of the Taung child by developmental simulation. *Journal of Human Evolution* 51: 274-296.
- McKinley KR. 1971. Survivorship in gracile and robust australopithecines: a demographic comparison and a proposed birth model. *American Journal of Physical Anthropology* 34: 417-426.
- Meindl RS, Mensforth RP, and Lovejoy CO. 2008. The Libben site: A hunting, fishing, and gathering village from the eastern Late Woodlands of North America. Analysis and implications for paleodemography and human origins. In: Bocquet-Appel J-P, editor. *Recent Advances in Paleodemography*. Dordrecht: Springer. p 259-275.

- Mensforth RP. 1985. Relative tibia long bone growth in the Libben and Bt-5 prehistoric skeletal populations. *American Journal of Physical Anthropology* 68: 247-262.
- Menter CG, Kuykendall KL, Keyser AW, and Conroy GC. 1999. First record of hominid teeth from the Plio-Pleistocene site of Gondolin, South Africa. *Journal of Human Evolution* 37: 299-307.
- Migliano AB, Vinicius L, and Mirazo M. 2007. Life history trade-offs explain the evolution of human pygmies. *Proceedings of the National Academy of Sciences* 104: 18-21.
- Mitteroecker P, Gunz P, Bernhard M, and Schaefer K. 2004. Comparison of cranial ontogenetic trajectories among great apes and humans. *Journal of Human Evolution* 46: 679-698.
- Moggi-Cecchi J, Menter C, Boccone S, and Keyser AW. 2010. Early hominin dental remains from the Plio-Pleistocene site of Drimolen, South Africa. *Journal of Human Evolution* 58: 374-405.
- Monge J, Mann A, Stout A, and Wadenya R. 2007. Dental calcification stages of the permanent M1 and M2 in U. S. children of African-American and European-American ancestry born in the 1990s. In: Bailey SE, Hublin JJ, editors. *Dental Perspectives on Human Evolution*. Dordrecht: Springer. p 263-274.
- Monge J, and Mann A. 2010. Dental maturation, middle childhood and the pattern of growth and development in earlier hominins. In: Broadfield D, Yuan M, Schick K, Toth N, editors. *The Human Brain Evolving: Paleoneurological Studies in Honor of Ralph L. Holloway*. Bloomington: Stone Age Institute Press. p 235-246.
- Moorrees CF, Fanning EA, and Hunt EE. 1963. Age variation of formation stages for ten permanent teeth. *Journal of Dental Research* 42: 1490-1502.
- Moss ML, and Young RW. 1960. A functional approach to craniology. *American Journal of Physical Anthropology* 18: 281-92.
- Mundry R. 1999. Testing related samples with missing values: A permutation approach. *Animal Behaviour* 58: 1143-1153.
- Okazaki K. 2004. A morphological study on the growth patterns of ancient people in the northern Kyushu-Yamaguchi region, Japan. *Anthropological Science* 112: 219-234.
- O'Higgins P, and Jones N. 1998. Facial growth in *Cercocebus torquatus*: an application of three-dimensional geometric morphometric techniques to the study of morphological variation. *Journal of Anatomy* 193: 251-272.
- Pacureanu A, Langer M, Boller E, Tafforeau P, and Peyrin F. 2012. Nanoscale imaging of the bone cell network with synchrotron X-ray tomography: optimization of acquisition setup. *Medical Physics* 39: 2229-2238.
- Pickering R, Kramers JD, John P, Ruiter DJD, and Woodhead JD. 2011. Contemporary flowstone development links early hominin bearing cave deposits in South Africa. *Earth and Planetary Science Letters* 306: 23-32.
- Pickering TR. 2006. Subsistence behaviour of South African Pleistocene hominids. South African Journal of Science 102: 205-211.
- Pickering TR, Heaton JL, Clarke RJ, Sutton MB, Brain CK, and Kuman K. 2012. New hominid fossils from Member 1 of the Swartkrans formation, South Africa. *Journal of Human Evolution* 62: 618-628.

- Plavcan JM, and Daegling DJ. 2006. Interspecific and intraspecific relationships between tooth size and jaw size in primates. *Journal of Human Evolution* 51: 171-184.
- Ponce de Leon MS, and Zollikofer CPE. 2001. Neanderthal cranial ontogeny and its implications for late hominid diversity. *Nature* 412: 534-538.
- Potvin C, and Roff DA. 1993. Distribution-free and robust statistical methods: Viable alternatives to parametric statistics? *Ecology* 74: 1617-1628.
- R Development Core Team (2011). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria.
- Raff RA. 2000. Evo-devo: The evolution of a new discipline. *Nature Reviews Genetics* 1:74-79.
- Raff RA. 2007. Written in stone: Fossils, genes and evo-devo. *Nature Reviews Genetics* 8: 911-920.
- Rafiq AM, Udagawa J, Lundh T, Jahan E, Matsumoto A, Sekine J, and Otani H. 2012. Mathematical analysis of mandibular morphogenesis by micro-CT-based mouse and alizarin red S-stained-based human studies during development. *Anatomical Record* 295: 313-27.
- Rak Y. 1983. The Australopithecine Face. New York: Academic Press.
- Rak Y, and Hylander WL. 2008. What Else Is the Tall Mandibular Ramus of the Robust Australopiths Good For? In: Vinyard C, Ravosa MJ, Wall C, editors. *Primate Craniofacial Function and Biology*. Boston, MA: Springer US. p 431-442.
- Randall FE. 1943. The skeletal and dental development and variability of the gorilla. *Human Biology* 15: 236-254.
- Richtsmeier JT, and Lele SR. 1993. A coordinate-free approach to the analysis of growth patterns: Models and theoretical considerations. *Biological Reviews* 68: 381-411.
- Richtsmeier JT, and Walker A. 1993. A Morphometric Study of Facial Growth. In: *The Nariokotome Homo erectus Skeleton*. p 391-420.
- Richtsmeier JT, Deleon VB, and Lele SR. 2002. The Promise of Geometric Morphometrics. *Yearbook of Physical Anthropology* 45: 63-91.
- Robbins MM, Gray M, Kagoda E, and Robbins AM. 2009. Population dynamics of the Bwindi mountain gorillas. *Biological Conservation* 142: 2886-2895.
- Robinson JT. 1954. Prehominid dentition and hominid evolution. Evolution 8: 324-334.
- Robinson JT. 1972. *Early Hominid Posture and Locomotion*. Chicago: University of Chicago Press. Robson SL, and Wood B. 2008. Hominin life history: Reconstruction and evolution. *Journal of Anatomy* 212: 394-425.
- Rohlf FJ. 1990. Morphometrics. *Annual Review of Ecology and Systematics* 21: 299-316.
- Rook L, and O'Higgins P. 2005. A comparative study of adult facial morphology and its ontogeny in the fossil macaque *Macaca majori* from Capo Figari, Sardinia, Italy. *Folia Primatologica* 76: 151-171.
- Royer DF, Lockwood CA, Scott JE, and Grine FE. 2009. Size variation in early human mandibles and molars from Klasies River, South Africa: Comparison with other Middle and Late Pleistocene assemblages and with modern humans. *American Journal of Physical Anthropology* 140: 312-323.

- Sacher GA. 1978. Longevity, aging and death: An evolutionary perspective. *Gerontologist* 18: 112-119.
- Sardi ML, Novellino PS, and Pucciarelli M. 2006. Craniofacial morphology in the Argentine Center-West: Consequences of the transition to food production. *American Journal of Physical Anthropology* 130: 333-343.
- Schaefer K, Mitteroecker P, Gunz P, Bernhard M, and Bookstein FL. 2004. Craniofacial sexual dimorphism patterns and allometry among extant hominids. *Annals of Anatomy* 186: 471-478.
- Scheuer L, and Black S. 2000. *Developmental Juvenile Osteology*. New York: Elsevier Academic Press.
- Scott R, Ungar PS, Bergstrom T, Brown CA, Grine FE, Teaford MF, and Walker A. 2005. Dental microwear texture analysis shows within-species diet variability in fossil hominins. *Nature* 436: 693-695.
- Shea BT. 1985. The ontogeny of sexual dimorphism in the African apes. *American Journal of Primatology* 8: 183-188.
- Smith BH. 1986. Dental development in *Australopithecus* and early *Homo. Nature* 323: 327-330.
- Smith BH. 1989. Dental development as a measure of life history in Primates. *Evolution* 43: 683-688.
- Smith BH. 1991. Dental development and the evolution of life history in Hominidae. American Journal of Physical Anthropology 86: 157-174.
- Smith BH. 1993. The Physiological Age of KNM-WT 15000. In: *The Nariokotome Homo erectus Skeleton*. p 195-220.
- Smith BH. 1994. Sequence of Emergence of the Permanent Teeth in Macaca, Pan, Homo, and Australopithecus: Its Evolutionary Significance. American Journal of Human Biology 6: 61-76.
- Smith BH, and Tompkins RL. 1995. Toward a life history of the Hominidae. *Annual Review of Anthropology* 24: 257-279.
- Smith RJ, Gannon PJ, and Smith BH. 1995. Ontogeny of australopithecines and early *Homo*: evidence from cranial capacity and dental eruption. *Journal of Human Evolution* 29: 155-168.
- Smith SL. 2004. Skeletal age, dental age, and the maturation of KNM-WT 15000. American Journal of Physical Anthropology 125: 105-120.
- Smith TM, Reid DJ, Dean MC, Olejniczak AJ, and Martin LB. 2007. Molar development in common chimpanzees (*Pan troglodytes*). Journal of Human *Evolution* 52: 201-216.
- Smith TM, Smith BH, Reid DJ, Siedel H, Vigilant L, Hublin J-J, and Boesch C. 2010. Dental development of the Taï Forest chimpanzees revisited. *Journal of Human Evolution* 58: 363-373.
- Sokal RR, and Rohlf FJ. 1995. *Biometry*. Second edition. New York: W. H. Freeman and Co.
- Solow B, and Houston WJ. 1988. Mandibular rotations: Concepts and terminology. *European Journal of Orthodontics* 10: 177-179.
- Sponheimer M, Passey B, de Ruiter DJ, Guatelli-Steinberg D, Cerling TE, and Lee-Thorp JA. 2006. Isotopic evidence for dietary variability in the early hominin *Paranthropus robustus*. *Science* 314: 980-982.

- Stearns SC. 1977. The Evolution of life history traits: A critique of the theory and a review of the data. *Annual Review of Ecology and Systematics* 8: 145-171.
- Stearns SC. 2000. Life history evolution: successes, limitations, and prospects. *Naturwissenschaften* 87: 476-486.
- Steininger C, Berger LR, and Kuhn BF. 2008. A partial skull of *Paranthropus robustus* from Cooper's Cave, South Africa. South African Journal of Science 101: 143-146.
- Strait DS, and Grine FE. 2004. Inferring hominoid and early hominid phylogeny using craniodental characters: The role of fossil taxa. *Journal of Human Evolution* 47: 399-452.
- Strauss RE, and Atanassov MN. 2006. Determining best complete subsets of specimens and characters for multivariate morphometric studies in the presence of large amounts of missing data. *Biological Journal of the Linnean Society* 88: 309-328.
- Sutton MB, Rayne T, Pickering R, Brain CK, Clarke RJ, Heaton JL, and Kuman K. 2009. Newly discovered fossil- and artifact-bearing deposits, uranium-series ages, and Plio-Pleistocene hominids at Swartkrans Cave, South Africa. *Journal of Human Evolution* 57: 688-696.
- Suwa G, Asfaw B, Beyene Y, White TD, Katoh S, Nagaoka S, Nakaya H, Uzawa K, Renne P, and WoldeGabriel G. 1997. The first skull of *Australopithecus boisei*. *Nature* 389: 489-492.
- Tafforeau P, Zermeno JP, and Smith TM. 2012. Tracking cellular-level enamel growth and structure in 4D with synchrotron imaging. *Journal of Human Evolution* 62: 424-428.
- Tanner JM. 1951. Some notes on the reporting of growth data. *Human Biology* 23: 93-159.
- Tardieu C. 1998. Short adolescence in early hominids: Infantile and adolescent growth of the human femur. *American Journal of Physical Anthropology* 107: 163-178.
- Taylor AB. 2002. Masticatory form and function in the African apes. *American Journal* of *Physical Anthropology* 117: 133-156.
- Taylor AB. 2006. Diet and mandibular morphology in African apes. *International Journal of Primatology* 27:181-201.
- Teaford MF, and Ungar PS. 2000. Diet and the evolution of the earliest human ancestors. *Proceedings of the National Academy of Sciences* 97: 13506-13511.
- Thackeray JF, Bellamy CL, Bellars D, Bronner G, Bronner L, Chimimba C, Fourie H, Kemp A, Kruger M, Plug I, Prinsloo S, Toms R, Van Zyl AJ, Whiting MJ. 1997. Probabilities of conspecificity: Application of a morphometric technique to modern taxa and fossil specimens attributed to *Australopithecus* and *Homo*. South African Journal of Science 93:195-196.
- Thompson JL, and Nelson AJ. 2000. The place of Neandertals in the evolution of hominid patterns of growth and development. *Journal of Human Evolution* 38: 475-495.
- Tompkins RL. 1996. Relative dental development of Upper Pleistocene hominids compared to human population variation. *American Journal of Physical Anthropology* 99: 103-118.
- Ungar PS, Grine FE, and Teaford MF. 2008. Dental microwear and diet of the Plio-Pleistocene hominin *Paranthropus boisei*. *PLoS One* 3: e2044.
Ungar PS, and Sponheimer M. 2011. The diets of early hominins. Science 344: 190-193.

- Van Arsdale AP. 2006. *Mandibular Variation in early Homo from Dmanisi, Georgia*. PhD thesis, University of Michigan.
- Van Arsdale AP, Cofran Z, and Papakyrikos A. 2012. Size and skeletal development in early Homo erectus: A comparison of Nariokotome and Dmanisi [abstract]. *American Journal of Physical Anthropology* 147: 290.
- van der Merwe NJ, Masao FT, and Bamford MK. 2008. Isotopic evidence for contrasting diets of early hominins *Homo habilis* and *Australopithecus boisei* of Tanzania. *South African Journal of Science* 104: 253-255.

Van Valen L. 1973. Review: Festschrift. Science 180: 488.

- von Cramon-Taubadel N. 2011. Global human mandibular variation reflects differences in agricultural and hunter-gatherer subsistence strategies. *Proceedings of the National Academy of Sciences* 108: 19546-19551.
- Walker RS, Gurven M, Hill KIM, Migliano A, Chagnon N, Souza RDE, Djurovic G, Hames R, Hurtado AM, Kaplan H, et al. 2006a. Growth rates and life histories in twenty-two small-scale societies. *American Journal of Human Biology* 18: 295-311.
- Walker RS, Hill K, Burger O, and Hurtado AM. 2006b. Life in the slow lane revisited: Ontogenetic separation between chimpanzees and humans. *American Journal of Physical Anthropology* 129: 577-583.
- Wang MK, Buschang PH, and Behrents R. 2009. Mandibular rotation and remodeling changes during early childhood. *The Angle Orthodontist* 79: 271-275.
- Watts ES, and Gavan JA. 1982. Postnatal growth of nonhuman primates: The problem of the adolescent spurt. *Human Biology* 54: 53-70.
- Wealthall RJ. 2002. Surface remodelling of the facial skeleton in juvenile *Macaca mulatta*: implications for sexual dimorphism. *Folia Primatologica* 73: 49-53.
- Weber GW. 2001. Virtual Anthropology (VA): A Call for Glasnost in paleoanthropology. *The Anatomical Record* 265: 193-201.
- White TD. 1977. *The Anterior Mandibular Corpus of Early African Hominidae: Functional Significance of Shape and Size*. PhD thesis, University of Michigan.
- Wolpoff MH. 1975. Some aspects of human mandibular evolution. In: McNamara JA, editor. *Determinants of Mandibular Form and Growth*. Ann Arbor: Center for Human Growth and Development, University of Michigan. p 1-64.
- Wolpoff MH. 1979. The Krapina dental remains. *American Journal of Physical Anthropology* 50: 67-114.
- Wolpoff MH. 1999. Paleoanthropology. Second Edition. Boston: McGraw-Hill.
- Wolpoff MH and Lee S-H. 2001. The Late Pleistocene human species of Israel. *Bulletins* et Mémoires de la Société d'Anthropologie de Paris 13: 291-310.
- Wolpoff MH, Monge J, and Lampl M. 1988. Was Taung human or an ape? *Nature* 355: 501.
- Wood B, and Aiello LC. 1998. Taxonomic and functional implications of mandibular scaling in early hominins. *American Journal of Physical Anthropology* 105: 523-538.
- Wood JW, Milner GR, Harpending HC, and Weiss KM. 1992. The osteological paradox: problems of inferring prehistoric health from skeletal samples. *Current Anthropology* 33: 343-370.

- Zihlman AL, Bolter D, and Boesch C. 2004. Wild chimpanzee dentition and its implications for assessing life history in immature hominin fossils. *Proceedings* of the National Academy of Sciences 101: 10541-10543.
- Zihlman AL, Bolter DR, and Boesch C. 2007. Skeletal and dental growth and development in chimpanzees of the Taï National Park, Côte D'Ivoire. *Journal of Zoology* 273: 63-73.
- Zollikofer CP, and Ponce de León MS. 2004. Kinematics of cranial ontogeny: heterotopy, heterochrony, and geometric morphometric analysis of growth models. *Journal of Experimental Zoology. Part B, Molecular and Developmental Evolution* 302: 322-40.
- Zumpano MP, and Richtsmeier JT. 2003. Growth-Related Shape Changes in the Fetal Craniofacial Complex of Humans (*Homo sapiens*) and Pigtailed Macaques (*Macaca nemestrina*): A 3D-CT Comparative Analysis. *American Journal of Physical Anthropology* 120: 339 -351.